University of South Carolina Scholar Commons

Theses and Dissertations

2016

Implementation Processes And Impacts Of School Store Policies In South Korea

Seul Ki Choi University of South Carolina

Follow this and additional works at: https://scholarcommons.sc.edu/etd Part of the <u>Public Health Education and Promotion Commons</u>

Recommended Citation

Choi, S. K. (2016). Implementation Processes And Impacts Of School Store Policies In South Korea. (Doctoral dissertation). Retrieved from https://scholarcommons.sc.edu/etd/3841

This Open Access Dissertation is brought to you by Scholar Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Scholar Commons. For more information, please contact dillarda@mailbox.sc.edu.



IMPLEMENTATION PROCESSES AND IMPACTS OF SCHOOL STORE POLICIES IN SOUTH KOREA

by

Seul Ki Choi

Bachelor of Science Seoul National University, 2007

Master of Public Health Seoul National University, 2009

Submitted in Partial Fulfillment of the Requirements

For the Degree of Doctor of Philosophy in

Health Promotion, Education, and Behavior

The Norman J. Arnold School of Public Health

University of South Carolina

2016

Accepted by:

Edward A. Frongillo, Major Professor

Christine E. Blake, Committee Member

James F. Thrasher, Committee Member

Mark E. Tompkins, Committee Member

Paul Allen Miller, Vice Provost and Interim Dean of Graduate Studies



© Copyright by Seul Ki Choi, 2016 All Rights Reserved



ACKNOWLEDGEMENTS

There are not enough words to express my thanks to all who have believed, supported, and encouraged me over the years. I would like first thank my advisor, Dr. Edward Frongillo, for his continuous encouragement and guidance during the PhD program. I could not have completed my program without his support. I am fortunate to have him as my mentor.

I wish to express my sincere thanks to my committee members Drs. Christine Blake, Jim Thrasher, and Mark Tompkins for their understanding, kindness, valuable feedback, and insights during the whole process. I am thankful to my MPH program advisor, Dr. Hyojee Joung, for allowing me to use her data for my dissertation. I also want to thank my interviewees and all people who helped me conduct interviews.

I would like to acknowledge and thank all people in Cancer Prevention and Control Program for their supports for my graduate assistantship and for giving me opportunities to broaden my research interests.

I would also like to thank my friends in South Carolina for being with me through both happy and hard times. Special thanks to my friends in Korea for their emotional supports regardless of the physical distance. Finally, I would like to extend my gratitude to my family for their endless love and unconditional supports.



www.manaraa.com

ABSTRACT

With the concern of unhealthy school food environments, the South Korean government enacted two school nutrition policies to reduce unhealthy foods in school stores. The first policy banned on soft drinks sales in school stores enacted in 2007. The second policy restricted energy-dense nutrient-poor (EDNP) food sales in school stores enacted in 2009. This study aims to examine the changes in adolescents' food intake and foods sold in school stores due to the two policies and to understand the policy implementation processes.

The Korea National Health and Nutrition Examination Surveys were used to examine trends in adolescents' energy intake from school store foods during 1998-2012. Energy intake from instant noodles consistently decreased both at home and in school stores. Energy intake from soft drinks away from home or school rapidly increased beginning in 2008. All foods sold in school stores were observed before (2006) and after (2013) the implementation of policies. The mean number of soft drinks sold in a school store significantly decreased in 2013 (0.3 items) compared to in 2006 (1.9 items, p=0.032). However, soft drinks were available in 50% of school stores observed in 2013 and all school stores observed in 2013 sold EDNP foods. In the qualitative interviews with policy actors, despite of the policy noncompliance, all interviewees perceived that school stores complied with the policy. Perspectives and values towards the policy were different for each actor group. Poor monitoring, lack of awareness of the policy, profit-



iv

seeking, and lack of interest in school stores were identified as reasons for incomplete implementation of the policy.

The two school store policies in South Korea appear to have had a modest impact on overall diet of adolescents and food availability in school stores. Comprehensive policies that target diverse eating places are needed to improve adolescents' overall diet. This study supports the needs for consideration of perspectives of various actors, especially for those who are affected by the policy, when implementing school nutrition policies. Understanding of various actors' involvement in policy processes can inform strategies to enhance implementation and thereby reach outcomes that align with the original policy intent.



TABLE OF CONTENTS

ACKNOWLEDGEMENTS
Abstract iv
LIST OF TABLES
LIST OF FIGURES ix
LIST OF ABBREVIATIONSx
CHAPTER 1 INTRODUCTION1
CHAPTER 2 BACKGROUND AND SIGNIFICANCE
2.1 School Food Environments
2.2 SCHOOL NUTRITION POLICIES
2.3 POLICY PROCESS
2.4 IMPLEMENTATION PROCESSES OF SCHOOL NUTRITION POLICIES10
2.5 SCHOOL FOOD ENVIRONMENTS AND RELATED POLICIES IN SOUTH KOREA12
2.6 Conceptual Framework
2.7 SIGNIFICANCE
CHAPTER 3 METHODOLOGY
3.1 Methods for Manuscript 1
3.2 Methods for Manuscript 2
3.3 Methods for Manuscript 3
3.4 ETHICAL APPROVAL



CHAPTER 4 RESULTS	.37
4.1 Manuscript 1: Korean Adolescents' Energy Intake of Selected Foods by Eating Place from 1998 to 2012 during Implementation of Two	S
NATIONAL SCHOOL NUTRITION POLICIES	.37
4.2 MANUSCRIPT 2: COMPARISON OF FOOD AVAILABILITY IN SCHOOL STORES IN SEOUL, SOUTH KOREA BEFORE AND AFTER IMPLEMENTATION OF TWO NATION FOOD- AND NUTRIENT- BASED POLICIES	[AL .62
4.3 MANUSCRIPT 3: WHY DO SCHOOL STORES IN SOUTH KOREA STILL SELL Restricted Foods Four Years After the School Store Policy?	.84
CHAPTER 5 SUMMARY, IMPLICATIONS AND RECOMMENDATIONS1	12
References	20
Appendix A – Interview Guide1	32
APPENDIX B – LIST OF CHILDREN'S FAVORITE FOODS	44



LIST OF TABLES

Table 2.1 Two nutrition policies focusing on school food environments in South Korea	ι14
Table 2.2 Criteria of the energy-dense nutrient-poor (EDNP) foods	16
Table 4.1 Two nutrition policies focusing on the school food environment in South Korea	41
Table 4.2 Demographic characteristics of adolescents from 1998-2012 Korea National Health and Nutrition Surveys	45
Table 4.3 Distinct changes in trend of energy intake from school store foods during selected periods	48
Table 4.4 Criteria of the energy-dense nutrient-poor (EDNP) foods	68
Table 4.5 Characteristics of school stores observed in 2006 and 2013	70
Table 4.6 Mean number of foods and beverages sold in school stores in 2006 and 2013	71
Table 4.7 Availability of energy-dense nutrient-poor (EDNP) foods in school stores in 2013	73
Table 4.8 Selected interview questions relevant to the elements of the social process model	88



LIST OF FIGURES

Figure 2.1 The causal model of the 2009 policy on restriction energy-dense nutrient-po	or
food sales in Children's Food Safety and Protection Zones	.17
·	
Figure 2.2 Conceptual model of the processes of the Korean school store policies	.23
Figure 4.1 Secular trends of energy intake from selected school store foods	.46



LIST OF ABBREVIATIONS

CHC	Community Health Center
EDNP	Energy-Dense Nutrient-Poor
FVJ	Fruit and Vegetable Juice
KNHANES	Korea National Health and Nutrition Examination
LOWESS	LOcally WEighed Scatter plot Smoothing
MFDS	Ministry of Food and Drug Safety
MoE	Ministry of Education
SSB	



CHAPTER 1

Introduction

Adolescents spend the majority of their waking time in school. School environments influence students' health behaviors, including food choices and dietary behaviors (Briefel, Crepinsek, Cabili, Wilson, & Gleason, 2009; Mensink, Schwinghammer, & Smeets, 2012; Wechsler, Devereaux, Davis, & Collins, 2000). Thus, schools have often served as places for health policies and programs to address adolescents' health issues (Jaime & Lock, 2009). Schools play an important role in their students' diets because students have one-third of their daily energy intake from schools through lunch and/or snacks in school stores (Briefel, Wilson, & Gleason, 2009). School meals are provided to students under the supervision of dietitians, whereas students choose snacks from school stores by themselves. Hence, foods and beverages available for sale (hereafter food availability) in school stores could directly affect students' diets. Food availability was found to be a key factor contributing to both healthy and unhealthy dietary behaviors of students during school hours (Briefel, Crepinsek, et al., 2009; Kubik, Lytle, Hannan, Perry, & Story, 2003; Larson & Story, 2010). Because most foods and beverages sold in school stores are often high in energy and low in nutrients, the limited healthy food options in school stores are criticized as one of the causes of childhood obesity and children's unhealthy diet (Clarke, Fletcher, Lancashire, Pallan, & Adab,



www.manaraa.com

2013; Story, Kaphingst, & French, 2006). For these reasons, the U.S. Institute of Medicine has recommended restrictions on unhealthy food and beverages sold outside of the school meal program (Stallings & Yaktine, 2007).

Efforts to improve school food environments have been made in many countries. One such effort is the enactment of policies that limit unhealthy food sales in schools. With the concern of increasing childhood obesity and unhealthy school food environments, South Korea has also enacted two national school nutrition policies to reduce unhealthy foods in school stores. The Korea Ministry of Education (MoE) required a ban on sales of soft drinks and recommended banning sales of instant noodles, fried foods, and fast foods in schools, including school stores, at the beginning of the school year in 2007 (hereafter 2007 policy). In 2009, the Special Act on Safety Management of Children's Dietary Life (hereafter the Special Act), a comprehensive nutrition policy targeted multiple nutritional problems of Korean children, was enacted. One component of the Special Act is a nutrient-based policy that restricts energy-dense nutrient-poor (EDNP) food sales in school stores (hereafter 2009 policy).

The two school store policies that restrict soft drinks and EDNP food sales in school stores have been implemented nationwide for the last nine and six years, respectively. Every school store is supposed to follow the school store policies, but several studies and media have reported that restricted foods are still available in some school stores (Ahn, 2009; S. C. Kim et al., 2010; S. Y. Lee, Lee, Kim, & Kim, 2012; B. Song, 2010). Nevertheless, there is a lack of studies that examine to what extent the school store policies have influenced adolescents' diets, and foods and beverages in school stores, and how the policies have been enforced and implemented. Thus, this study



www.manaraa.com

aims to examine the changes in food intake of adolescents and foods available in school stores due to the school store policies and to understand implementation processes of the 2009 policy in South Korea.

This study consists of three manuscripts. The first manuscript of the dissertation examines trends of Korean adolescents' energy intake from selected school store foods by eating places during 1998-2012. This study aims to assess the extent to which adolescents' intake of the target foods of the school store policies shifted. The second manuscript assesses foods and beverages sold in school stores in South Korea in 2006 and 2013. The aim of the study is to assess how the two different national school store policies have changed school store food environments. The third manuscript reports qualitative research that aims to understand the implementation process of the 2009 policy through interviews with people who are directly or indirectly involved in the translation of the policy into specific rules and practices. The research questions of this study are: (1) How do policy actors describe policy implementation processes?, (2) Who are the policy actors and how do they act?, (3) Are the seven elements of social process (Clark, 2002) reflected differently across schools and policy actors?, and (4) What factors affect the extent of policy implementation among schools?

Chapter 2 describes a review of the scientific literature that provides current understanding of school food environments and efforts to improve school food environments both in South Korea and worldwide. Knowledge gaps are also stated. Chapter 3 presents methodology of the study. Chapter 4 presents the three manuscripts. Chapter 5 includes in overall discussion and conclusions, and implications for future research.



www.manaraa.com

CHAPTER 2

Background and Significance

2.1 School Food Environments

Individual efforts to have healthy lifestyles can succeed within a supportive environment. Schools are important settings for shaping students' health behaviors including healthy diets (Story et al., 2006; Wojcicki & Heyman, 2006). Students have up to two meals as well as snacks at school during a school day and their peers and teachers serve as role-models for dietary behaviors (Story et al., 2006; Wojcicki & Heyman, 2006).

Food environments regarding food accessibility and food availability influence people's food purchase and consumption (Kant & Graubard, 2003; Martin et al., 2012; Wansink, 2004). Concerns about school food environments have been raised because of unhealthy foods, especially soft drinks and EDNP foods, sold in schools. Availability of unhealthy foods in school stores or vending machines in schools is associated with students' poor diet (Briefel, Crepinsek, et al., 2009). For example, students of schools with stores or snack bars had more energy intake from sugar-sweetened beverage (SSB) than those attending schools without stores or snack bars (Briefel, Crepinsek, et al., 2009). For students attending public schools in the U.S., schools are one of major sources of SSB intake. They consumed more SSB in schools than at home or in other places.



Their in-school energy intake from SSB is estimated to range from 16-90 kcal per school day, which turns into 3,404-17,100 kcal per school year (Briefel, Crepinsek, et al., 2009).

Access to unhealthy foods in schools not only increases unhealthy food intake but also decreases healthy food intake. Middle-school students who gain access to school snack bars consume less fruit, milk, and regular-vegetables and more SSB and high-fat vegetables, compared to when they were in elementary school with access to only school meals (Cullen & Zakeri, 2004). In addition, students' daily intake of fruit servings decreases by 11% with each snack vending machine located in a school (Kubik et al., 2003). Unhealthy food consumption may also alter intake of other healthy foods, for example, when milk is replaced with soft drinks (Y.-J. Bae & Yeon, 2013; Taillie, Afeiche, Eldridge, & Popkin, 2015; Vatanparast, Lo, Henry, & Whiting, 2006).

To improve students' diet, several efforts have been taken worldwide. One such effort is development and implementation of school nutrition policies aiming at creating healthier school food environments.

2.2 School Nutrition Policies

School nutrition policies have historically been targeting the safety and nutritional quality of foods provided by school meal programs and outside of school meal programs such as school stores, snack bars, and vending machines. Foods and beverages available from outside of school meal programs are called competitive foods because students' consumption of those foods often compete with consumption of school meals (Fox, Meinen, Pesik, Landis, & Remington, 2005). Most competitive foods are high in energy and low in nutritional values in U.S. schools (D. M. Finkelstein, Hill, & Whitaker, 2008;



O'Toole, Anderson, Miller, & Guthrie, 2007; Story et al., 2006), as well as in schools in South Korea (M. Kim, 2013). In light of growing concerns about the contribution of unhealthy competitive foods to unhealthy school food environments, implementation of school nutrition policies that restrict unhealthy food sales and/or require healthy foods has recently increased.

School nutrition policies targeting competitive foods in the U.S. have been effective in improving school food environments and students' food intake (Chriqui, Pickel, & Story, 2014; Cradock et al., 2011; Kubik et al., 2013; Schwartz, Novak, & Fiore, 2009; Taber et al., 2011). U.S. students in states with a policy targeting SSB sales in schools purchase less SSB and consumed fewer servings of SSB than those in states without a restrictive policy (Jones, Gonzalez, & Frongillo, 2009; Taber et al., 2011). Students who attended Boston public high schools consumed less soda, other SSB, and total SSB, after a policy restricting SSB sales in Boston public high schools passed, while SSB consumption of nationwide samples in the same period did not change (Cradock et al., 2011). A recent systematic review of competitive food policies in the U.S. reported that competitive food policies are associated with positive changes in students' food consumption and food availability in schools (Chriqui et al., 2014).

Policies targeting competitive foods attempt to either limit specific food items or set a standard for nutrient content of the food. The first type of policy, the food-based policy, is mostly used to restrict sales of soft drinks or fried foods. The second type of policy, the nutrient-based policy, is used to limit foods with high energy, fat, sugar, or sodium. California's competitive food policy that has been implemented since 2007 includes food standards and beverage standards (Senate Bills 12 and 965) and showed a



good example of food- and nutrient-based policies. The food standards of California's competitive food policy specify amounts of energy, sugar, and energy from fat and saturated fat in the foods that can be sold in schools (nutrient-based policy). Beverage standards have a list of types of beverages that are allowed to be sold in schools (foodbased policy). Previous research that compared food availability in schools before and after implementation of food- and nutrient-based policies reported that both types of policies contributed to the decrease in the unhealthy food availability in schools, but the reduction in unhealthy food availability varied across studies (Chriqui, Turner, Taber, & Chaloupka, 2013; Fernandes, 2013; Peart et al., 2012; Samuels et al., 2009; Woodward-Lopez et al., 2010). In some cases, nutrient-based policies resulted in the removal of only the least healthy foods, thus access to healthy foods are not increased (Samuels et al., 2009; Woodward-Lopez et al., 2010). In general, schools sell less restricted foods under food-based policies than foods restricted under nutrient-based policies; however, none of the policies achieved 100% removal of the restricted foods (Peart et al., 2012; Samuels et al., 2009; Woodward-Lopez et al., 2010). Food-based policies are more straightforward while policy implementers often have difficulty in understanding complex nutrient criteria of nutrient-based policies (Chriqui et al., 2014; Chriqui et al., 2013; Woodward-Lopez et al., 2010). Less is known, however, about which type of policy most improves school food environments and adolescents' diets.

Knowing policy influences and the conditions under which they are most effective could help in developing and implementing more effective policies in the future. The two types of policies are known to be associated with the decrease in availability of and students' in-school consumption of the restricted foods (Chriqui et al., 2014; Cradock



www.manaraa.com

et al., 2011; Fernandes, 2013; Jones et al., 2009; Peart et al., 2012; Samuels et al., 2009; Taber et al., 2011; Woodward-Lopez et al., 2010), but these policies have limited impacts on students' overall diets (Chriqui et al., 2014; Jones et al., 2009; Taber, Chriqui, Powell, & Chaloupka, 2012). Students may compensate for restrictions in schools by consuming more of the restricted foods when they are not in school or by bringing foods from home or outside of school (Cullen, Watson, & Zakeri, 2008; E. Finkelstein, French, Variyam, & Haines, 2004; Vecchiarelli, Takayanagi, & Neumann, 2006). Some studies reported no compensatory intake at home among students while the intake of restricted foods in schools decreased (Schwartz et al., 2009). Still, less is known about where the compensatory behavior occurs and how it affects overall diets. Thus, the need to understand the influence of school nutrition policies on students' consumption in school and out of school has been brought to the fore (Chriqui et al., 2014; Cullen et al., 2008).

2.3 Policy Process

Many people and interest groups, from the government level to the individual level, are involved in the policy process. The participants of the policy process are called policy actors. Legislators and government agencies whose roles are specified in the policy are considered official actors; individual citizens, interest groups including industry, political parties, media, and research organizations are considered unofficial actors. "Unofficial" does not mean that their participation in the policy process is not expected; it means that their roles are not specified in the policy (Birkland, 2005). Both official and unofficial actors are involved in the policy process; but each actor's role and influences are not always the same during the whole policy process.



Agenda setting is a stage in which some societal problems are raised to elicit political attention from decision makers. Legislators, politicians, advocacy groups, media, and researchers have power in the agenda setting stage (Kingdon, 2002). Official actors, including legislators and politicians, are directly involved in the policy formulation and legislation processes. They may have less impact on policy implementation compared to the huge amount of power they exert in policy agenda setting and enactment (Kingdon, 2002). On the other hand, public officials may have a greater impact on policy enforcement and implementation processes although their involvement in policy agenda setting is limited (Kingdon, 2002). Public officials are usually in charge of ensuring compliance with the policies. They enforce policies aiming at preventing non-compliance of the policy and eliminate the benefits of policy non-compliance, often using monetary fines and incentives (Organization for Economic Co-operation and Development, 2009). Public officials, especially in the local level governments, are expected to be in a better position to get feedback about the policy from the front-line policy implementers and citizens, and are also expected to monitor specific indicators more regularly than policy makers (politicians), researchers involved in the policy making, and public officials in higher-level governments. Public officials receive feedback about poor implementation that does not meet the original policy intention, unmet policy goals, cost-related problems, new problems, and unanticipated consequences of the policy (Kingdon, 2002). Actors who are affected by policies, such as school store owners in the case of the school store policy, are often not included in the decision-making process, which turns into a misunderstanding of the policy intent, distrust of the government, or unsupportive responses to the policy (Roberts, Pobocik, Deek, Besgrove, & Prostine, 2009). Policies



are less likely to be implemented in expected ways when actors, whose cooperation is essential for policy implementation, do not involve themselves (either intentionally or unintentionally) in the decision making process (Cram, 1997).

Policy implementation is difficult. Many explicit and implicit factors and policy actors whose involvement differs in each stage of policy processes affect and are affected by policy implementation. Pressman and Wildavsky (1984) explained this complex situation as: "The longer the chain of causality, the more numerous the reciprocal relationships among the links and the more complex implementation becomes" (p. xxiv).

Recognizing policy actors is an important step in policy processes since they can hinder or enhance policy implementation (Clark, 2002; Clarke et al., 2013). The perspectives and attitudes of policy makers and implementers affect how the policy is being implemented (Vine & Elliott, 2014). In addition, difficulties in satisfying different perspectives among policy actors and conflicts in responsibility among policy actor groups, such as expecting other actors to support or oppose the policy implementation, may hinder implementation processes (Clarke et al., 2013; Green & Aarons, 2011).

2.4 Implementation Processes of School Nutrition Policies

Like other public policies, there are various actors in school nutrition policies including school staff (e.g., teachers, principals, school nurses, health educators, food service directors, and school board/governors), students and their families, health professionals, private sector (e.g., food industry), and non-governmental organizations (Clarke et al., 2013; Pan Canadian Joint Consortium for School Health, 2010). They affect and/or are affected by school nutrition policies, but not all of them are actively



involved in the policy process (Clarke et al., 2013; Roberts et al., 2009). Front-line actors, who are implementing school store policies such as food service personnel, are often left out of the decision-making process for development and implementation of the policy (Roberts et al., 2009).

While most actors agreed that school nutrition policies are good for students' health (Monterrosa et al., 2015; Vine & Elliott, 2014). Some actor groups, such as food industry, however, expected that school nutrition policies will not be effective due to students' unhealthy diet out of schools and wanted to have policies that focus more on family-level, nutrition education, or physical activity (Monterrosa et al., 2015; Roberts et al., 2009; Vine & Elliott, 2014). Concerns of revenue loss; students' food preference; a lack of communication, support, and coordination; complexity of policies; limited resources; and competing priorities have been identified as barriers in school nutrition policy implementation (Agron, Berends, Ellis, & Gonzalez, 2010; Clarke et al., 2013; Davee et al., 2005; Greves & Rivara, 2006; Mâsse, Naiman, & Naylor, 2013; McKenna, 2003; Roberts et al., 2009; Schwartz et al., 2012). Private food service providers or schools make profits through food sales in schools. If they expect that school nutrition policies affect their revenue, they would be less likely to comply with the policy (Mâsse et al., 2013; McKenna, 2003; Roberts et al., 2009). In addition, if students do not like the new items the policies require or items that have to be removed are students' preferred foods, food service providers would resist the policies. Limited involvement in the policy process for some policy actors, such as school food service personnel, students, and parents, causes a lack of communication which leads to unawareness of or misunderstanding the details of the policies, unsupportive attitudes toward the policies,



www.manaraa.com

and even unintentional noncompliance with the policies (Haroun, Harper, Wood, & Nelson, 2011; McKenna, 2003; Roberts et al., 2009). School staff and school food service personnel often do not have or know about resources to interpret or to implement the policies. Thus it causes difficulties in compliance with the policies, e.g., having difficulties finding foods that meet policy requirements (Samuels et al., 2009; Schwartz et al., 2012). Having resources (e.g., booklet, websites providing policy information or supports from dietitians), organized communication to increase policy awareness among policy actors, and support from school administrator, school board, parents, and politicians are identified as facilitators of school nutrition policy implementation (Agron et al., 2010; Davee et al., 2005; Dodson et al., 2009; Mâsse et al., 2013).

Previous studies reported actors' roles and thoughts on school nutrition policies from school administrators, food service personnel, public officials in governmental institutions, students, and parents. Yet there are only a few studies that investigated the views of multiple different types of actors on school nutrition policies together (McKenna, 2003; Monterrosa et al., 2015; Roberts et al., 2009; Vine & Elliott, 2014).

2.5 School Food Environments and Related Policies in South Korea

For Korean students, major food sources in schools are from school meal services (mainly for lunch) and school stores. As of 2014, 99.8% of Korean students participate in the school lunch program (Korea Ministry of Education, 2015), and a total 52% of middle and high schools in Seoul, South Korea have school stores (110 middle schools, 35%; 248 high schools, 65%) (M. Kim, 2013). School stores are small stores in schools that sell foods and beverages to students during school hours. Usually, only one store is



located in a school and they mostly offer snacks rather than meal substitutes. Chips, cookies, and bread are high-selling products in school stores (Chang et al., 2007; Y. J. Kim, 2010). Most of school stores are operated by private retailers, not schools. Thus, school stores are often not controlled by the schools while school meal service is managed by school dietitians.

The presence of school stores is related to snack intake of Korean students. Korean students attending schools with school stores are more likely to eat more snacks, to spend more money for snack purchases compared to other students attending schools without a school store (Do, 2012). More than 80% of students among those attending schools with school stores use them at least once a week (Y.-J. Kim, 2012; Yun, 2014). Given that school stores offer unhealthy snacks to students, concerns regarding unhealthy school store foods that may cause childhood obesity have been growing. To address unhealthy school store food environments, two national policies have been enacted.

In March 2006, the Korea National Youth Commission recommended banning soft drink sales in schools and youth training facilities (Korea National Youth Commission, 2006). Following this, in February 2007, the MoE announced the 2007 food-based policy that banned sales of soft drinks and recommended schools not sell instant noodles, fried foods, and fast foods. The 2007 policy began to be implemented in March 2007, the onset of the 2007 school year (Table 2.1). After the MoE's recommendation of banning instant noodle sales in schools, most regional offices of education banned instant noodle sales in school stores.

In February 2007, the Korean Ministry of Food and Drug Safety (MFDS) announced a comprehensive plan for the safety management of children's foods. This



Table 2.1 Two nutrition policies focusing on school food environments^a in South Korea

Policy		Policy contents	
	Ban on soft drink sales (Ministry of Education)		
March 2006	Korea National Youth Commission recommended banning	Require banning soft drinks sales in schools (including	
	soft drink sales in schools and youth training facilities	school stores and vending machines) and recommend	
February 2007	Announcement of the ban on soft drink sales in schools	banning on instant noodles and fried foods sales in	
March 2007	Implementation of the ban on soft drink sales in schools	schools. ^b	
	The Special Act (Ministry of Food and Drug Safety)		
March 2008	Enactment of the Special Act	Areas within 200 meters in a straight line from schools	
March 2009	Enforcement of the Special Act	are designated as Children's Food Safety and Protection	
	Begin banning energy-dense nutrient-poor (EDNP) foods	Zones. Stores that cook and/or sell children's favorite	
	sales in exemplary stores located in Children's Food Safety	foods outside of schools may voluntarily register to be	
	and Protection Zones	exemplary business places. ^c School stores and	
May 2010	First announcement of the list of EDNP products	exemplary business places may be restricted or	
		prohibited the sale of EDNP foods.	

^a Policies regarding school meal services were not considered in this study.

^b After the Ministry of Education's recommendation of ban on instant noodle sales in schools, most regional offices of education banned instant noodle sales in school stores.

^c All school stores are required to register to be exemplary business places.



plan included legislation of the Special Act (an umbrella policy of the 2009 policy) that aimed at "promoting children's health by prescribing matters necessary for supplying safe and nutritionally balanced foods in order to help children acquire proper dietary habits" (the Special Act, article 1). The Special Act was enacted in March 2008 and was implemented beginning in March 2009 (S. C. Kim et al., 2010) (Table 2.1). As a part of the Special Act, the 2009 nutrient-based policy restricted EDNP food sales in Children's Food Safety and Protection Zones and in schools. Areas within 200 meters in a straight line from schools are designated as Children's Food Safety and Protection Zones (the Special Act, articles 5 and 6). Among stores located in Children's Food Safety and Protection Zones, those that cook and/or sell children's favorite foods can voluntarily pledge to not sell EDNP foods. All stores located in schools are mandated not to sell EDNP foods. The 2009 policy set a nutrient standard for energy, protein, saturated fat, sugar, and/or sodium to define EDNP foods and beverages (Table 2.2), forbade their sale in schools and allowed voluntary compliance for stores in Children's Food Safety and Protection Zones after March 2009 (the beginning of the 2009 school year). The 2009 policy intended to improve children's health through a decrease in EDNP food sales in schools and in Children's Food Safety and Protection Zones. Figure 2.1 presents the causal model of the 2009 policy. The MFDS reported that as of 2012, among 42,765 stores that cook and/or sell children's favorite foods in Children's Food Safety and Protection Zones and in schools (including school stores), only 1,904 stores (4.2%) volunteered not to sell EDNP foods. In addition, most of the voluntary stores were school stores (Korea Ministry of Food and Drug Safety, 2013). Thus, the 2009 policy may only affect foods in school stores.



Table 2.2 Criteria of the energy-dense nutrient-poor (EDNP) foods



- (1) Foods containing over 250kcal and less than 2g of protein per serving,
- (2) Foods containing over 4g of saturated fat and less than 2g or protein per serving,
- (3) Foods containing over 17g of sugar and less than 2g of protein per serving
- (4) Foods containing over 500kcal or over 8g of saturated fat or over 34g of sugar per serving among foods that do not meet criteria of 1~3.
- * For foods which the serving size is less than 30g, the criteria is applied to 30g of the foods instead of the actual serving size.

B. Criteria for meal substitutes among children's favorite foods

- (1) Foods containing over 500kcal and less than 9g of protein per serving,
- (2) Foods containing over 500kcal and over 600mg of sodium per serving. For deep-fried noodles and noodles among noodles (only applied to noodles in containers), over 1,000mg of sodium is applied to as a criteria of EDNP.
- (3) Foods containing over 4g of saturated fat and less than 9g of protein per serving,
- (4) Foods containing over 4g of saturated fat and over 600mg of sodium per serving. For deep-fried noodles and noodles among noodles (only applied to noodles in containers), over 1,000mg of sodium is applied to as a criteria of EDNP.
- (5) Foods containing over 1,000kcal or over 8g of saturated fat per serving among foods that do not meet criteria of 1~4

^a Children's favorite foods are defined as "Foods that children prefer or eat frequently among foods under the Food Sanitation Act or the Livestock Products Sanitary Control Act, which are prescribed by Presidential Decree" (the Special Act on safety management of children's dietary life. Act no. 119882013). The list of children's favorite foods is in Appendix B.

If the policies had been implemented as intended, school store food environments

in South Korea would have improved and students would not have access to unhealthy

foods in school stores. News media and several studies have reported consistently,

however, that EDNP foods are still available in many school stores (Ahn, 2009; S. C.

Kim et al., 2010; S. Y. Lee et al., 2012; B. Song, 2010). Several studies also found



↓ Energy-dense nutrient-poor (EDNP) food sales in Children's Food Safety and Protection Zones ↓ ↓ Intake of EDNP foods among children ↓ ↑ Children who have proper dietary habits ↓ ↓ Obesity and non-communicable diseases among children

Figure 2.1 The causal model of the 2009 policy on restriction energy-dense nutrient-poor food sales in Children's Food Safety and Protection Zones

EDNP foods in school stores: 27.8% of foods sold in school stores in 2009 (S. Y. Lee et al., 2012) and 31 school stores sold EDNP foods among 51 school stores that were observed in 2010 (S. C. Kim et al., 2010). Nevertheless, two studies that evaluated the impact of the school nutrition policies on Korean adolescents' food intake reported positive changes such as decreased intake of soft drinks, confectionaries, instant noodles, and fast foods (S. G. Bae et al., 2012; K. Kim, Park, & Oh, 2013). These studies focused only on the overall frequency of food intake regardless of the eating place and did not consider the quantity of the food consumed. Thus, it is not known to what extent school nutrition policies affect Korean adolescents' food intake and whether the changes differ in each eating place (in school and out of school) before and after implementation of policies.

In addition, there is a lack of understanding of implementation processes of school store policies. In South Korea, nutrition policy processes have not been well



documented, although evaluation of the effectiveness of nutrition policies has been conducted frequently (S. Kim, 2010; S. Kim et al., 2013; S. K. Lee, 2012). If the policy process is not understood or addressed well, the sustainable effectiveness of the policy cannot be guaranteed (Pelletier et al., 2012).

2.6 Conceptual Framework

The policy process is complex (Buse, 2008; Walt et al., 2008). Although there may be political attention paid to a specific issue, it does not always turn into policy formulation and implementation (Pelletier et al., 2012). After law enactment, we cannot expect that a policy will be implemented according to the legislative intent (Kingdon, 2002). The ways that a policy is being implemented and individuals or groups who implement the policy shape implementation processes (Birkland, 2005). During the policy implementation, social values, norms, and practices are intertwined and affect its processes and results (Paudel, 2009). Implementation processes reflect interactions between implementers and other actors as well as interactions among actors. Implementation processes also explain the variation of the implementers' performance (Winter, 2003).

The policy enactment itself cannot guarantee its successful implementation (Kingdon, 2002). Understanding policy implementation processes is important because it is a critical step of the policy process (Birkland, 2005). It is problematic if policy implementation processes are treated as a "black box" when the policy evaluation is conducted (Hill & Hupe, 2002). Studies of policy implementation processes can also provide information regarding better ways to make policies whose implementation will



be consistent with their legislative intention (Birkland, 2005). This study includes an inquiry as to why school stores implement the same policy differently. Regardless of whether the policy is implemented well, policy makers and implementers can learn from both the successes and failures of the policy by tracking policy implementation processes.

Two approaches are commonly used to assess policy implementation (Ripley & Franklin, 1986). One is focusing on the compliance of implementers. Policy implementation is assessed based on whether implementers comply with a preexisting model of correct implementation in terms of procedures, timetables, and restrictions. Another approach is interested in "what is happening and why?" during policy implementation processes. This study uses both approaches to understand implementation processes of the school store policies in South Korea.

This study is guided by Lasswell (1971)'s social process model. The model emphasizes interaction between participants, who seek to maximize values, and between participants and resource environments in the social process. The model explains social process using the following seven categories:

- Participants: Individuals, groups, and organizations who are interacting in the social process
- (2) Perspectives: Participants' identity, demands, and expectations
- (3) Situations: Where social interactions take place in terms of geographic location, timing and process of events, institutionalization, and crisis
- (4) Base values: Assets or resources that participants possess: power, enlightenment, wealth, well-being, skill, affection, respect, and rectitude
- (5) Strategies: Ways to achieve values



(6) Outcomes: Changes in base values as a result of the social process

(7) Effects: Long-term outcomes

The participants include not only those participate in the process, but also those who do not currently participate in the process, but should or could participate, and are affected by the problem, situation, and the process. Perspectives are the ways participants see social problems. It is reflected by participants' perception of who they are (identity), what they want (demands), and what they assume or anticipate happening (expectations) in the social process. Situations are venues where interactions occur while people's values change. Eight values can be found in the social process include: power (making/influencing decisions), enlightenment (gathering and disseminating knowledge), wealth (production, distribution, and control of resources, wellbeing (being physically and psychologically healthy and safe), skill (having special abilities), affection (having family, friendship, intimacy, and warm relationships), respect (showing and receiving deference, freedom of choice), and rectitude (having ethical standards) (Clark, 2002; Lasswell, 1971).

Base values are values that participants possess. These can be used to obtain more values. When the values become participants' demands or outcomes that are sought, those values are scope values. Strategies include various tactics to pursue scope values. Outcomes are short-term results of social interaction which are shown as gain or loss of values. Effects are long-term outcomes of the social process (Clark, 2002; Lasswell, 1971).



Each element of the social process model affects others and shapes the process. Mapping the elements of the social process model is useful for understanding how, by whom, and why the social interaction is occurring. The social process model considers legal (policy) process as a part of the social process. Using elements of the social process model, this study attempts to understand implementation processes of the Korean school store policy (2009 policy).

Paudel (2009) described policy implementation as location- or country-specific processes which carry social values, norms, and practices and are situation-specific. This study defines implementation, using terms from the social process model (Clark, 2002), as a process that is laden with participants' values, perspectives, and interaction among participants.

In implementation research, policy implementation processes are often explained with policy outputs and outcomes together (Winter, 2003). These are, however, different in their concepts and the way they are measured. Birkland (2005) defined policy outputs as "the effort that government expends to address problems" (p. 158). Policy outcomes reflect how the target population responds to the policy implementation (Winter, 2003). It could be either intended or unintended, positive or negative (Birkland, 2005; Clark, 2002). Winter (2003) emphasized the needs to examine policy output and outcomes separately as dependent variables.

This study attempts to distinguish policy implementation processes, outputs, and outcomes and then assess all of them (Figure 2.2). Implementation processes are qualitatively measured, and outputs and outcomes of two Korean school store policies are quantitatively measured as dependent variables. Policy outputs and outcomes are



determined based on the causal model of the 2009 policy (Figure 2.1). Policy outputs are considered to be what policy implementers do. Implementers of school store policies are school store owners; thus, the policy output in this study is what school store owners do in response to the policy. School store owners select the foods they are going to sell, thus food availability in school stores reflects how well they implement school store policies. The policy output of school store policies is the food availability, specifically, availability of soft drinks (for the 2007 policy) and EDNP foods (for the 2009 policy) in school stores. The outcomes of school store policies are changes in adolescents' food intake, especially soft drinks and EDNP food intake.

2.7 Significance

An increasing amount of research has been done to evaluate school-based food and nutrition policies to improve school food environments (O'Toole et al., 2007; Schwartz et al., 2009; Vericker, 2013). There are not many studies, however, that consider both policy implementation processes and its effectiveness in terms of improving school food environments and changing diets of students simultaneously (Mâsse et al., 2013).

Implementation research can help policy makers to be aware of the root causes of problems, barriers, and facilitators of implementation. Understanding these will increase the chance of ensuring high-quality policy implementation and positive effects on public health (Panisset et al., 2012). This study examines both school store policy implementation processes and its effects to the target population.





Figure 2.2 Conceptual model of the processes of the Korean school store policies

This is the first study applying qualitative research methods to examine nutrition policy processes in South Korea. Although qualitative research methods including interviews with policy actors have been used in nutrition policy research in the U.S. and other countries, research on nutrition policy in South Korea has been conducted only to enumerate problems and to evaluate the effectiveness of the policy rather than to understand what factors affect policy implementation. This study could be a starting point to facilitate in-depth research on nutrition policies in South Korea.

Many countries have enacted policies to improve school food environments. The U.S. Department of Agriculture announced a school store policy on competitive foods in schools on February 1, 2013 (Nixon, 2013). This study will be able to provide insights for policy implementation to improve school food environments globally based on South

Korea's case.



To our knowledge, the places where Korean adolescents consume foods targeted by the school store policies have not been studied, even though their food intake away from home and the frequency of overall snack intake have increased during the last decade (Y. Lee, Shim, & Yoon, 2012). This study will add a deeper layer to our understanding of the impact of nutrition policies on changes in food intake by identifying where shifts in food intake occur after policy implementation.

Previous studies regarding school nutrition policies mostly focused on the policy implementation in public schools (Chriqui et al., 2013; Peart et al., 2012; Samuels et al., 2009; Templeton, Marlette, & Panemangalore, 2005). Public and private schools have different school management systems, including operation of school stores. In South Korea, 80% of public schools make contracts with school stores through a public competitive bidding process, whereas 63% of private schools have private contracts (H. Lee, Jang, & Kim, 2011). For schools using a public competitive bidding, the right of school store operation is sold to the highest bidder (i.e., pays the school the most for school food sales rights). The right of operation expires every year, thus the current school store owner needs to participate in the bidding annually. Store owners in public schools reported that high bidding prices are overly burdensome (Choi, Frongillo, Blake, Thrasher, & Tompkins, 2015). These differences in school store management systems may affect school store owners' selection of foods sold in school stores. School store owners who already paid high cost for the right to the store need to make more profits, thus they may want to sell high-profit products regardless of the nutritional quality of foods. Since public and private high schools in South Korea have the same tuition (excluding specialized schools; tuition for middle schools is free), attendance in private



or public schools does not reflect students' socioeconomic status. Thus, comparison of food availability between public and private schools in South Korea could provide an example of impact of school management systems on food availability in school stores.


CHAPTER 3

Methodology

The 2009 policy was a part of a larger policy (the Special Act) that aimed at improving children's diets. Several food and nutrition policies are being implemented simultaneously with the 2007 and 2009 school store policies. The two policies are implemented nationwide. Hence, the policy implementation can be considered as natural experiments. It does not have a control group since the policy is implemented nationwide. The policy implementation may occur in various ways in terms of the timing and the extent of the implementation (Petticrew et al., 2005). In addition, it is impossible to distinguish whether the changes in the target population's behaviors are due to the 2007 and 2009 policies or other similar policies and interventions. Because of these uncertainties, the causal relationship between the policies and key outcomes is difficult to assess. Thus, factors that could directly or indirectly affect policy outcomes should be acknowledged and analyzed, to the extent possible. For evaluation of natural experiments, multi-method evaluations, such as using qualitative data or routine data, are useful (Petticrew et al., 2005).

This study employed mixed methods. Changes in students' food intake (policy outcomes) and food availability in school stores (policy outputs) are quantitatively assessed and policy implementation processes are qualitatively examined.



3.1 Methods for manuscript 1

Data and participants. The Korea National Health and Nutrition Examination Survey (KNHANES) is a nationally representative, cross-sectional survey designed to assess the health and nutritional status of the population in South Korea. The survey was conducted 2-3 months in 1998 (first cycle, survey was conducted from November to December), 2001 (second cycle, survey was conducted from November to December), and 2005 (third cycle, survey was conducted from April to June). From the fourth cycle of KNAHENS (2007-2010), the period for data collection was extended to reduce seasonal bias (i.e., different foods available at different times of the year). The 2007 KHANES was conducted from July to December, and from 2008 onwards, KNHANES was conducted from January to December (http://knhanes.cdc.go.kr). Additional details of survey design and procedures were reported elsewhere (Korea Ministry of Health and Welfare & Korea Centers for Disease Control and Prevention [KCDC], 2013a; Kweon et al., 2014; http://knhanes.cdc.go.kr).

The participants of this study were adolescents who attend middle- (equivalent to 7-9 grades of the U.S. school system) or high-schools (equivalent to10-12 grades of the U.S. school system), and participated in a 24-hour dietary recall of one of the KNHANES conducted during 1998-2012. Of 74,342 participants who completed both the demographic survey and the 24-hour dietary recall, 5,025 participants were middle- or high- school students (age range: 12-19 years). After exclusion of 66 participants who reported <500kcal or \geq 5,000kcal per day, a total of 4,959 participants were included in the analysis.



Intake of school store foods. One 24-hour dietary recall was administered by trained dietitians to each participant. Although a single 24-hour dietary recall is inadequate for assessing individual intake or the distribution of usual intake, it provides suitable assessment of the mean intake of groups, which was the goal in this study (National Center for Health Statistics, 1992). In addition, data from 24-hour dietary recalls in large-scale surveys are useful to evaluate health policy impacts (Hebert et al., 2014). A further advantage of 24-hour recall for this study is its utility for identifying individual foods that the policies target, which could not be done using food frequency questionnaire data that group together food items. Participants were asked about their food intake during the previous day, as well as the eating place and the food source (e.g., school meal service, restaurant, etc.). We calculated energy intake from chips & cookies (including crackers), candy (including chocolate, gum, and jelly), milk (including flavored milk), soft drinks, fruit and vegetable juice (FVJ), and instant noodles. Energy intake from each food was examined according to eating places (home; school store; school; away from home or school). For eating episodes which occurred in a school, the eating place was considered school or school store based on the reported source for the food. Where the food source was school meal service, lunch box, or foods from home or other restaurants, we considered the eating place as school, not the school store. For other food sources, for example, purchased foods, we assumed the eating place was a school store.

Statistical analysis. Descriptive statistics were presented with weighted percentages, least squares means, and standard errors. Multiple linear regression models were used to estimate energy intake from each school store food by eating place.



Adjusted means of energy intake from each food were graphically presented to investigate the secular trend of food intake over time to take into consideration the uncertainties around policy implementation, including varied compliance across schools. Plots were drawn for each eating place using locally weighted scatter plot smoothing (LOWESS) (Cleveland, 1979). The smoothing parameter for LOWESS curves was set to a relatively high value, 0.75, to see long-term trends of food intake.

When a distinct change in energy intake from the food was seen in the graph, we estimated and compared slopes of energy intake from foods by eating place before and after the point in time for which the trend appeared to change using piecewise linear regression models. Positive values of slope indicate an increasing trend of energy intake from the food; negative values of slope indicate a decreasing trend of energy intake from the food.

All analyses were conducted using SAS survey procedures (SAS Institute Inc., Cary, NC) with sampling weights, primary sampling units, and stratification to take into account the complex sampling design (Korea Ministry of Health and Welfare & KCDC, 2013a). Potential confounders - sex, age, area of residence (metropolitan, nonmetropolitan), household income (quartiles), and total energy intake - were adjusted for all analyses. A p-value <0.05 was considered statistically significant.

3.2 Methods for manuscript 2

Participants and procedure. School stores in middle and high schools located in Seoul, South Korea were observed before (school year 2006) and after (school year 2013) implementation of school store policies. The school store observations in 2006 and 2013



had different purposes. Observations conducted in 2006 were aimed at collecting availability of school stores foods from representative sample of middle and high school in Seoul, Korea. Thus, one middle school and one high school were randomly selected from each school district among 11 school districts in Seoul. Of the 22 schools, 15 agreed to observation of their school store. Observations conducted in 2013 were a part of qualitative study (manuscript 3). We tried to select school stores with various characteristics in terms of students' sex, school level, and school type (private, public). Thus, six school stores each from one high-income school district and one low-income school district were purposefully selected. Among the total of 12 school stores observed in 2013, two school stores were also observed in 2006. The observations were conducted from December 2006 to February 2007 and from October 2013 to December 2013.

Four observers conducted school store observation in 2006. In 2013, one observer, who also participated in the observation in 2006, conducted all school store observation with the same material and method in 2006. When an observer visited school stores, all types of foods and beverages sold in the school stores were recorded. Product name, manufacturer, size, and flavor (if varied) were recorded. When school stores owners allowed, the observer took photographs of school stores and food items. For foods in school stores observed in 2013, nutrient content was recorded from the nutrition label of products, including calories, protein, saturated fat, sugar, and sodium content per one serving. In most cases, it was not possible to record nutrient content for all foods during the school store observation, because school store owners did not want the observer to stay in their school stores for a long time. If nutrient information was not



obtained by on-site observation, the information was obtained from the manufacturers' websites or by purchasing the product.

Data analysis. All foods and beverages were categorized as soft drinks, fruit and vegetable juice, tea, coffee, sports drinks, bottled water, other beverages, chips & cookies, candy, ice cream, bread, hamburger & pizza, fruit, other foods, yoghurt, milk, or flavored milk. Milk and flavored milk were further categorized based on the fat content (full- or reduced-fat). Based on the 2009 criteria for EDNP (Table 2.2), all food and beverage items available in school stores in 2013 were classified as either EDNP or not.

Descriptive statistics for school characteristics, including school level (middle school, high school), school type (private, public), students' sex (coeducational, boys only, girls only), and income level of school district (high-income, low-income), were computed using Fisher's exact test to compare characteristics of school stores observed in 2006 and 2013. Crude numbers of all types of foods and beverages sold in school stores and of foods in each category were compared between 2006 and 2013 using Student's t-tests. To compare the number of foods between 2006 and 2013 controlling for school characteristics, mixed models were fitted with school districts as random effects. Then, the total number of foods sold in school stores was included in the models in addition to school characteristics, to examine food composition in school stores in 2006 and 2013. Student's t-test was used to compare crude number of total foods and EDNP foods, and percentage of EDNP in 2013 by school level, school type, students' sex, and income level of school district. Mixed models with school districts as random effects were used to control school characteristics. The α level of 0.05 was used to determine statistical



significance. All statistical analyses were conducted using SAS 9.4 (SAS Institute, Inc., Cary, NC).

3.3 Methods for manuscript 3

This study is guided by Lasswell (1971)'s social process model. Using elements of the social process model, this study attempts to understand the implementation processes of the school store policy. We considered participants of the school store policy process as policy actors.

Participants. Participants in the school store policy implementation were considered as potential interviewees. We first chose six schools each from two school districts in Seoul, South Korea. We assumed that various school characteristics – incomelevel, students' sex composition (only for boys, only for girls, coeducational), school level (middle schools, high schools), and school type (private, public) – might be related to the processes of policy implementation. Thus, we purposefully selected two school districts based on the income level (high- versus low-income) and then selected schools to have different characteristics. When there was more than one school that had the same characteristics in the same school district, interviewed schools were randomly selected.

Conceptually-driven sequential sampling was used, wherein the initial interviewees were selected based on existing understanding, and then the rest of the interviewees were purposefully selected based on themes that emerged during the previous interviews (Miles & Huberman, 1994). We identified potential interviewees that were expected to be involved in the implementation processes of the school store policy based on reviews of research papers, government reports, and news articles that reported



on school food environments (Varvasovszky & Brugha, 2000). The pre-identified interviewees were policy actors in schools including school store owners, principals, teachers, dietitians, and administrative staff; and outside of schools including government agencies, food industry, and advocacy groups for healthy eating. During the interviews with pre-identified interviewees, if specific people or their roles related to policy implementation were mentioned, these people were sought out and recruited for interviews.

School store owners, as front-line implementers of the policy, were contacted for the interviews first. After the interviews with school store owners, potential in-school actors (e.g., school staff) in the same school were asked for the interviews. We also reached out to food industry, at least one public official in community health centers and offices of education in each school district that we selected, and public officials in the national-level governments (e.g., MFDS, MoE) for interviews. Politicians, consumer associations, and academia were frequently mentioned during the interviews, thus, they were also contacted for interviews.

Research instrument. The social process model informed the development of a semi-structured interview guide (Clark, 2002; Lasswell, 1971). It consisted of main questions, follow-up questions, and probes into emergent topics that interviewees brought up (Ulin, Robinson, & Tolley, 2004) to gain information from policy actors about the policy implementation processes focusing on policy actors, their roles, perspectives, perceptions, and relationships with other actors (Clark, 2002; Lasswell, 1971). Additional questions were included to address enforcement and monitoring of the policy. The interview guide included some different questions for each policy actor group that were



specific to their roles in the policy implementation processes. After review of the field notes and memos from the previous interviews, the interview guide was revised to include questions that emerged for subsequent interviews.

Procedure. The first author conducted all interviews. The interviewer visited school stores and asked school store owners if they were willing to participate in the interviews. The visits were made without prior notice to prevent the possibility that school store owners might change foods sold in school stores if they were aware of the interview in advance. For the same reason, contacts and interviews with school staff were always conducted after the interview with the school store owner of the same school. If a school store owner agreed to the interview, then the interview was conducted in the same day the interviewer visited. After the interview, the interviewer recorded all food and beverage sold in the school store. The details of the process of school store observations and food availability in the school stores are described in the manuscript 2. One school store owner among the 12 selected schools refused the interview, but allowed school store observation. Interviews with school staff in the same school were conducted. For other interviewees, the interviewer contacted them via phone or email to explain the purpose of the interview. Interviews were conducted either face-to-face or by phone in Korean and lasted 15-60 minutes.

A total of 33 people, including 11 school store owners; 8 school staff members; 6 public officials in community health centers, an office of education, Seoul Metropolitan government, and MFDS; 2 politicians; 3 from food company representatives; and 3 people who have been involved in school store related work (a professor, a school principal, and a person from a consumer organization), participated in the interviews.



Interviews were audio-recorded upon agreement of the interviewees. When interviewees did not want to be recorded, interview notes were taken by the interviewer. Eight school store owners, 3 school staff members, 1 public official, 1 politician, 1 from a food company representative, and 2 people who have been involved in school store related work were agreed to audio-record interviews, thus, a total of 16 interviews was audio-recorded. Among interviewees who did not agree to audio-record interviews, 1 politician and 1 food company representative provided their own memos related to interview contents to the interviewer. All data collection took place in Seoul, South Korea from October 2013 to January 2014.

Data analysis. All recorded interviews were transcribed verbatim; unrecorded interviews were reconstructed into transcripts within a day using the memos taken during the interview, field notes, and memory of the interviewer. De-identified transcripts and field notes were used for the analysis. All transcripts were reviewed by the interviewer for accuracy of the data. A subset of transcripts (n=5) was initially analyzed using open coding to capture emergent codes (Strauss & Corbin, 1990) and to label the responses of the interviewees. Using these emergent codes and elements of the social process model (Lasswell, 1971), a preliminary set of codes was developed. These codes were then applied to the first set of transcripts again and to the rest of transcripts. One researcher, who conducted all the interviews, coded all transcripts and worked with team members to develop the code list and identify themes. As the code list was revised or new themes were identified, all transcripts were re-coded using the new codes or themes. Themes based on the elements of the social process model – policy actors, base values (e.g., wealth, wellbeing, power, and relationship [affection]); perspectives (i.e., demands,



expectation, and identity), and policy outcomes – and additional themes emergent from interviews – perception of the policy, and compliance with the policy were identified. Upon completion of coding, selected themes were put in matrices to see the commonalities shared by interviewees from each policy actor group as well as different points expressed by them for the same themes (Miles & Huberman, 1994; Ulin et al., 2004). Peer review with one of co-author was conducted to reach consensus on the coding, grouping of the codes into themes, and identified themes. All transcripts were managed using NVivo 10 (QSR International, Melbourne, Australia) and analyzed by the first author.

3.4 Ethical approval

The study was reviewed and deemed exempt by the Institutional Review Board at the University of South Carolina. The protocol of the school store observation in 2006 for manuscript 2 was approved by the Institutional Review Board at the Graduate School of Public Health, Seoul National University.



CHAPTER 4

Results

4.1 Manuscript 1

KOREAN ADOLESCENTS' ENERGY INTAKE OF SELECTED FOODS BY EATING PLACE FROM 1998 TO 2012 DURING IMPLEMENTATION OF TWO NATIONAL SCHOOL NUTRITION POLICIES¹

¹ Choi, S.K., Frongillo, E.A., Blake, C.E., and Thrasher, J.F. Submitted to *Journal of Hunger and Environmental Nutrition*, 9/18/2015.



Abstract

This study examines changes in Korean adolescents' energy intake from selected foods that were offered in school stores over 15 years, during which the two policies prohibiting unhealthy food sales in school stores were implemented in 2007 and in 2009. Using 24-hour dietary recall from the Korea National Health and Nutrition Examination Surveys 1998-2012, adolescents' energy intake in each eating place was calculated. Energy intake from instant noodles consistently decreased both at home and in school stores. Energy intake from soft drinks away from home or school rapidly increased beginning in 2008. Policies regulating unhealthy food sales in school stores appear to have had a modest impact on adolescents' overall consumption of unhealthy foods. Comprehensive policies, targeting diverse eating places, are needed to improve adolescents' overall diet.

Key words: school food environments; school nutrition policy; adolescents; school store; eating place

Introduction

The school food environment influences students' food choices and dietary behaviors (Briefel, Crepinsek, et al., 2009; Chriqui et al., 2014; Larson & Story, 2010). Given that foods and beverages available in school stores are often high in energy and low in nutritional value (Story et al., 2006), policies to limit unhealthy foods in schools have increasingly been adopted around the world (U.S. Department of Agriculture & Food and Nutrition Service, 2013). A recent systematic review of research in the United States reported that policies restricting unhealthy foods in schools have reduced



unhealthy food availability and students' in-school consumption of those foods, but these policies did not influence students' overall diet and body mass index (Chriqui et al., 2014). Students may compensate for restrictions in school by consuming more of the restricted foods when they are not in school (E. Finkelstein et al., 2004; Vecchiarelli et al., 2006). Less is known, however, about where the compensation behavior occurs and how it affects overall diet. Thus, the need to understand school nutrition policies' influence on students' consumption in school and out of school has been brought to the fore (Chriqui et al., 2014; Cullen et al., 2008).

South Korea has made several efforts to improve the school food environment. For Korean students, major food sources in schools are from school meal services (mainly for lunch) and school stores. All middle and high schools in South Korea provide school lunch to 99% of their students (Korea Ministry of Education, 2015). School lunch is managed by registered dietitians in each school and is provided based on a nutrition standard stated in the School Meals Act (Yoon et al., 2012). School stores in South Korea are not allowed to sell cooked foods, thus mostly offer snack foods and beverages to students during school hours. Since most of school stores are operated by private retailers, not schools, foods offered by school stores that Korean students eat mostly between meals were less managed while nutritional quality of school lunch has been improved. Thus, the Korean government has enacted policies to reduce unhealthy foods in school stores that might contribute to excessive intake of energy, fat, or sugar. The Korean Ministry of Education required a ban on soft drinks sales in schools and recommended banning instant noodles, fried foods, and fast foods sales in schools at the beginning of the school year in 2007 (herein after 2007 policy). In addition, the Special



Act on Safety Management of Children's Dietary life that was enforced in 2009 includes a restriction of energy-dense nutrient-poor (EDNP) foods sales in school stores (herein after 2009 policy) (Table 4.1). Most school stores, however, still sell EDNP foods, thus, it is unclear to what extent the school store policies have changed food availability in school stores (Choi, Frongillo, Blake, & Thrasher, 2016). Two studies have evaluated the impact of the school nutrition policies on adolescents' intake of soft drinks, confectionaries, instant noodles, and fast foods; these studies focused only on overall frequency of food intake regardless of the eating place (S. G. Bae et al., 2012; K. Kim et al., 2013). It is not known how school nutrition policies affect Korean adolescents' food intake and whether the changes differ in each eating place (school and out of school) over a long period of time before and after implementation of policies.

We aimed to examine how Korean adolescents' energy intake from selected foods changed over 15 years. Due to the inconsistent timing of school store policy implementation in each individual school store, the changes in foods sold in school stores varied across schools and made it difficult to identify specific points to compare adolescents' food intake. Thus, we visualized food intake trends first over time and then examined how the national implementation dates of school store policies corresponded to those changes. We focused on the intake of some selected foods that adolescents could buy from their school stores (herein after school store foods) and the introduction of national policies regulating access to these foods, during this 15 year time period.



Table 4.1 Two nutrition policies focusing on the school food environment^a in South Korea

Policy		Policy contents
	Ban on soft drink sales (Ministry of Education)	
March 2006	Korea National Youth Commission recommended banning	Require banning soft drinks sales in schools (including
	soft drink sales in schools and youth training facilities	school stores and vending machines) and recommend
February 2007	Announcement of the ban on soft drink sales in schools	banning on instant noodles and fried foods sales in
March 2007	Implementation of the ban on soft drink sales in schools	schools. ^b
	The Special Act (Ministry of Food and Drug Safety)	
March 2008	Enactment of the Special Act	Areas within 200 meters in a straight line from schools
		are designated as Children's Food Safety and Protection
March 2009	Enforcement of the Special Act	Zones. Stores that cook and/or sell children's preferred
	Begin banning energy-dense nutrient-poor (EDNP) foods	foods outside of schools may voluntarily register to be
	sales in exemplary stores located in Children's Food Safety	exemplary stores. ^c School stores and exemplary stores
	and Protection Zones	should not sell EDNP foods.
May 2010	First announcement of the list of EDNP products	

^a Policies regarding school meal services were not considered in this study. ^b After the Ministry of Education's recommendation of ban on instant noodle sales in school, most regional offices of education banned instant noodle sales in school stores.

^c All school stores are considered as exemplary stores



Methods

Data and participants

The Korea National Health and Nutrition Examination Survey (KNHANES) is a nationally representative, cross-sectional survey designed to assess the health and nutritional status of the population in South Korea. The survey was conducted during different periods each year (http://knhanes.cdc.go.kr). Additional details of survey design and procedures were reported elsewhere (Korea Ministry of Health and Welfare & KCDC, 2013a; Kweon et al., 2014; http://knhanes.cdc.go.kr).

The participants of the study were adolescents who attend middle- or highschools, and participated in a 24-hour dietary recall of one of the KNHANES conducted during 1998-2012. Of 74,342 participants who completed both the demographic survey and the 24-hour dietary recall, 5,025 participants were middle- or high- school students (age range: 12-19 years). After exclusion of 66 participants who reported <500kcal or \geq 5,000kcal per day, a total of 4,959 participants were included in the analysis. *Intake of school store foods*

One 24-hour dietary recall was administered by trained dietitians to each participant. Although a single 24-hour dietary recall is inadequate for assessing individual intake or the distribution of usual intake, it provides suitable assessment of the mean intake of groups, which was the goal in this study (National Center for Health Statistics, 1992). In addition, data from 24-hour dietary recalls in large-scale surveys are useful to evaluate health policy impacts (Hebert et al., 2014). A further advantage of 24-hour recall for this study is its utility for identifying individual foods that the policies target, which could not be done using food frequency questionnaire data that group together food



items. Participants were asked about their food intake during the previous day, as well as the eating place and the food source (e.g., school meal service, restaurant, etc.). We calculated energy intake from chips & cookies (including crackers), candy (including chocolate, gum, and jelly), milk (including flavored milk), soft drinks, fruit and vegetable juice (FVJ), and instant noodles. Energy intake from each food was examined according to eating places (home; school store; school; away from home or school). For eating episodes which occurred in a school, the eating place was considered school or school store based on the reported source for the food. Where the food source was school meal service, lunch box, or foods from home or other restaurants, we considered the eating place as school, not the school store. For other food sources, for example, purchased foods, we assumed the eating place was a school store.

Statistical analysis

Descriptive statistics were presented with weighted percentages, least squares means, and standard errors. Multiple linear regression models were used to estimate energy intake from each school store food by eating place. Adjusted means of energy intake from each food were graphically presented to investigate the secular trend of food intake over time to take into consideration the uncertainties around policy implementation, including varied compliance across schools. Plots were drawn for each eating place using locally weighted scatter plot smoothing (LOWESS) (Cleveland, 1979). The smoothing parameter for LOWESS curves was set to a relatively high value, 0.75, to see long-term trends of food intake.

When a distinct change in energy intake from the food was seen in the graph, we estimated and compared slopes of energy intake from foods by eating place before and



after the point in time for which the trend appeared to change using piecewise linear regression models. Positive values of slope indicate an increasing trend of energy intake from the food; negative values of slope indicate a decreasing trend of energy intake from the food.

All analyses were conducted using SAS survey procedures (SAS Institute Inc., Cary, NC) with sampling weights, primary sampling units, and stratification to take into account the complex sampling design (Korea Ministry of Health and Welfare & KCDC, 2013a). Potential confounders - sex, age, area of residence (metropolitan, nonmetropolitan), household income (quartiles), and total energy intake - were adjusted for all analyses. A p-value <0.05 was considered statistically significant.

Results

As shown in Table 4.2, 52.4% were middle school students, 52.9% were boys, and 81.8% resided in metropolitan areas. More than 60% of their household income level was in the third or fourth quartiles.

Energy intake from school store foods away from home or school were lower than intake at home, except energy intake from soft drinks after 2009, and were higher than those intake in school or in school stores, except energy intake from milk (Figure 4.1). Energy intake from school store foods in school and in school stores were relatively small.

Energy intake from soft drinks away from home or school rapidly increased beginning in 2008. In-school store consumption of soft drinks slightly decreased after 2005, then increased again (Figure 4.1a). Although energy intake from milk at home



Yea	r 1998	2001	2005	2007	2008	2009	2010	2011	2012
	(n=917)	(n=619)	(n=515)	(n=250)	(n=558)	(n=691)	(n=514)	(n=465)	(n=430)
School level									
Middle school	43.3	55.4	58.9	51.3	57.9	53.4	54.6	50.6	47.5
High school	56.7	44.6	41.1	48.7	42.1	46.6	45.4	49.4	52.5
Sex									
Boy	51.4	50.5	53.2	56.2	49.1	53.3	52.8	50.1	56.9
Girl	48.6	49.5	46.8	43.8	50.9	46.7	47.2	49.9	43.1
Residential area									
Urban	76.8	77.9	80.0	81.5	83.8	84.2	79.1	83.5	81.5
Rural	23.2	22.1	20.0	18.5	16.2	15.8	20.9	16.5	18.5
Household income (quartile)									
Q1	16.4	15.9	15.1	8.1	11.8	14.9	15.8	15.6	11.5
Q2	22.2	25.7	28.5	27.0	20.4	22.6	31.7	27.8	23.8
Q3	34.1	28.8	30.7	33.8	29.2	33.5	26.0	29.5	31.3
Q4	27.3	29.6	25.7	31.1	38.5	29.0	26.4	27.1	33.5

Table 4.2 Demographic characteristics of adolescents from 1998-2012 Korea National Health and Nutrition Surveys^a

Weighted %, Least Square Mean ± Standard Error

^a Tabulations are weighted for the sample design. Sample sizes are unweighted.



Figure 4.1 Secular trends of energy intake from selected school store foods



fluctuated over time, milk consumption in other places changed little during the study period (Table 4.3, Figure 4.1b), while energy intake from soft drinks and FVJ increased (Figure 4.1a, 4.1c). Energy intake from instant noodles at home and in school stores decreased continuously, but the intake away from home or school remained constant over time (Figure 4.1d). Energy intake from chips & cookies at home dropped in 2008 (38.1 kcal decrease from 2007, p=0.002) then increased again. Intake of chips & cookies in school stores dropped in 2007 (13.8 kcal decrease from 2005, p=0.045), then after it remained constant (Figure 4.1e). Energy intake from candy significantly increased since 2005 in all places except school (Table 4.3, Figure 4.1f).

Energy intake from school store foods away from home or school were lower than intake at home, except energy intake from soft drinks after 2009, and were higher than those intake in school or in school stores, except energy intake from milk (Figure 4.1). Energy intake from school store foods in school and in school stores were relatively small.

Energy intake from soft drinks away from home or school rapidly increased beginning in 2008. In-school store consumption of soft drinks slightly decreased after 2005, then increased again (Figure 4.1a). Although energy intake from milk at home fluctuated over time, milk consumption in other places changed little during the study period (Table 4.3, Figure 4.1b), while energy intake from soft drinks and FVJ increased (Figure 4.1a, 4.1c). Energy intake from instant noodles at home and in school stores decreased continuously, but the intake away from home or school remained constant over time (Figure 4.1d). Energy intake from chips & cookies at home dropped in 2008 (38.1 kcal decrease from 2007, p=0.002) then increased again. Intake of chips & cookies in



Food	Eating place	Time period	Estimate ^a ±Standard Error	Р
Milk	Home	2005-2009	6.24±1.66	0.002
		2009-2012	-4.24±2.19	
Candy	Home	1998-2005	-1.51±0.44	< 0.001
		2005-2012	1.68±0.47	
	School stores	1998-2005	-0.31±0.17	0.013
		2005-2012	0.54±0.20	
	Other places	1998-2005	-0.33 ± 0.20	0.015
		2005-2012	0.91±0.35	

Table 4.3 Distinct changes in trend of energy intake from school store foods during selected periods

^a Slope, adjusted for age, sex, residential area, and household income (quartiles)

school stores dropped in 2007 (13.8 kcal decrease from 2005, p=0.045), then after it remained constant (Figure 4.1e). Energy intake from candy significantly increased since 2005 in all places except school (Table 4.3, Figure 4.1f).

In-school store consumption of soft drinks and instant noodles, which are target foods of the 2007 policy, decreased beginning at or before the 2007 policy implementation; there were no notable changes, however, in energy intake from school store foods after the 2009 policy implementation. School store food consumption was mostly unchanged after 2009, with energy intake from soft drinks in school stores higher after 2009 compared to 2007-2009.



Discussion

We examined trends in Korean adolescents' energy intake from selected school store foods during 1998-2012. At-home intake accounted for the highest proportion of energy intake from school store foods over time, with the exception of energy intake from soft drinks away from home or school after 2009. Over the entire period, energy intake both from soft drinks away from home or school and from FVJ at home and away from home or school consistently increased, while energy intake from instant noodles at home and in school stores decreased.

Given that home contributes the highest proportion of both healthy and unhealthy food consumption among adolescents (Briefel, Wilson, et al., 2009), improvement of athome consumption might benefit adolescents' diets more than improvement in other places. The change in energy intake from instant noodles at home is a positive example. Energy intake from instant noodles at home dramatically decreased, while the intake away from home or school remained constant. Instant noodles have high sodium and fat contents and are major sources of sodium and fat intake for Koreans (Korea Ministry of Health and Welfare & KCDC, 2013b; D. Y. Song, Park, Shim, & Lee, 2013). Due to consistent efforts to reduce sodium intake by the Korean government (Korea Ministry of Food and Drug Safety, 2014b) and negative images of instant noodles generated by media (Korea Food Industry Association & Chung-Ang University, 2011), a widespread perception that instant noodles are unhealthy has been created. This change in public perception may have influenced parents' provision of instant noodles to their children at home (Korea Food Industry Association & Chung-Ang University, 2011), thus, adolescents' instant noodles intake has decreased. Efforts of parents as well as



adolescents are required in order to improve adolescents' at-home diet by creating healthier home food environments, providing healthy foods, and becoming role models for healthy eating (Hanson, Neumark-Sztainer, Eisenberg, Story, & Wall, 2005; Story et al., 2006).

Korean adolescents consumed more unhealthy foods away from home or school where less restriction on food choices exists, as shown by a rapidly increasing trend of energy intake from soft drinks and a lack of reduction in energy intake from instant noodles away from home or school. Adolescents can access various food outlets with more food choices away from home or school under less supervision from their parents or teachers. Thus, food environments away from home or school need to be improved to steer adolescents to make healthy food choices. Restricting advertisements and sales of unhealthy foods and encouraging healthy food production and sales could be possible ways to improve food environments outside of home or school (Gittelsohn, 2012; World Health Organization, 2010).

Our study results suggest that the 2007 policy aimed at decreasing availability of specific foods in schools led to decreased consumption of the target foods, as seen in the decreased energy intake from instant noodles and soft drink intake in school stores in contrast to trends in intake in other eating places. Over 90% of middle and high schools sold soft drinks through vending machines or school stores before the 2007 policy (Korea National Youth Commission, 2006). By October 2007, soft drinks were available in only 7% of middle and high schools (Korea Ministry of Education & Human Resources Development, 2007). Energy intake from soft drinks in school stores decreased after the 2007 policy was implemented while intake away from home or school increased during



the same period. Energy intake from instant noodles in school stores also decreased while the intake away from home or school remained flat.

It is unclear to what extent the 2009 policy influenced adolescents' intake of specific school store foods. It raises questions about why the intake of target foods of the 2009 policy (EDNP foods) was changed less than the intake of target foods of the 2007 policy (soft drinks). Several Korean nutrition policies are simultaneously addressed by multiple government agencies. These policies often overlap and lack consistency (H. R. Kim, 2008). The 2007 and 2009 policies also overlap in terms of the targeted setting and populations, but the target foods and the way to define restricted foods differ. The 2007 policy was a food-based policy that specifically indicated which foods should not be sold in schools. It might have been easy for school store owners to take the restricted foods out of their stores. On the other hand, the 2009 policy was a nutrient-based policy. To determine restricted foods, the contents of energy, saturated fat, sugar, protein, and/or sodium per serving had to be considered (Table 2.2). Based on the 2009 policy, soft drinks and instant noodles can be sold in school if those are not EDNP. We assume one of reasons why energy intake from soft drinks in school stores increased after 2011 is that school stores resumed sales of soft drinks that were not classified as EDNP, for example a soft drink with less than 250 kcal per serving, after the 2009 policy implementation. An investigation of food availability in school stores in 2013 found that 50% of school stores in Seoul, South Korea sold soft drinks although the number of soft drinks in school stores was lower than in 2007 (Choi et al., 2016).

Not only does the nutrient-based policy allow some unhealthy food sales, but the complexity of the policy makes it hard to achieve its goals. School store owners and



school staff may be easily confused by the nutrient criteria of restricted foods and have difficulty discerning which foods they can or cannot sell (Choi et al., 2015; Woodward-Lopez et al., 2010). In the case of the 2009 policy, the nutrient criteria are complicated, and differ between types of snacks (e.g., chips & cookies) and meal substitutes (e.g., instant noodles). In addition, nutrient-based policies may have loopholes. For example, food manufacturers can avoid making EDNP products by changing the content of only one or two nutrients in their products (Choi et al., 2015; H. Kim, 2014). Previous studies reported that food-based policies make it easier to achieve compliance to a policy, and nutrient-based policies allow unhealthy foods that technically meet the nutrient criteria of the policy to be available in schools (Samuels et al., 2009; Woodward-Lopez et al., 2010). Increased energy intake from candy in school stores may be due in part to the increased in-school store availability of candy meeting the nutrient criteria of the 2009 policy (e.g. a candy with less than 17g of sugar per serving). Based on our study results, in South Korea, it is hard to make improvements to the nutritional quality of the foods available in school and to influence adolescents' intake of school store foods with a nutrient-based policy. Further studies are needed to evaluate which kind of policy is more effective to improve adolescents' food intake in which circumstances (Shroff, Jones, Frongillo, & Howlett, 2012).

In-school consumption of selected school store foods is lower than consumption in other places (Briefel, Wilson, et al., 2009); nevertheless, the school food environment is still important to adolescents' diets. Schools can help students to eat healthier foods that they would not eat away from home or school. Higher energy intake from milk in school or school store than away from home or school shown in this study could be an



example of the schools' positive influence on students' diet choices. Milk availability in school as a part of the school meal service and in school stores helps adolescents to drink more milk in school. Schools could be venues to provide healthy foods to their students to promote healthy diets (Briefel, Crepinsek, et al., 2009). Strategies aiming to improve students' diet should consider both the increase of healthy foods and the decrease of unhealthy foods in school. Unhealthy food consumption may alter intake of other healthy foods, for example, when milk is replaced by soft drinks and FVJ (Y.-J. Bae & Yeon, 2013; Korea Ministry of Agrigulture, Food and Rural Affairs, & Korea Food Industry Association, 2013b; Vatanparast et al., 2006). Our results also showed that energy intake from soft drinks and FVJ increased while energy intake from milk decrease of unhealthy foods in school. With the restriction of unhealthy foods in school, requiring healthy foods would be effective approach to improve students' diets.

Previous studies reported that Korean adolescents' consumption of unhealthy foods, including soft drinks, instant noodles, confectionary, and fast foods, had decreased after implementation of the 2007 and 2009 policies (S. G. Bae et al., 2012; K. Kim et al., 2013). To our knowledge, the places where Korean adolescents consume foods targeted by the policies has not been studied, even though their food intake away from home and their frequency of overall snack intake have increased during the last decade (Y. Lee et al., 2012). Our study adds a deeper layer to our understanding of the impact of nutrition policies on changes in food intake by identifying where shifts in food intake occur after policy implementation.



This study has several limitations. One of the potential limitations is that we could not discern EDNP foods, which are the target of the 2009 policy, from non-EDNP foods because the KNHANES food database does not provide sugar content of foods. Thus, we were focused on specific types of foods instead of EDNP foods. In addition, the KNHANES food database lacks information about processed foods, hence respondents' intake of particular foods could be misclassified with other similar food when the database does not have information of the specific food. Also, no distinction was made between zero-calorie drinks and regular drinks or full-fat milk and reduced-fat milk, therefore we grouped these in our analyses. We found that no one reported consuming zero-calorie soft drinks prior to 2009, which is unlikely to be accurate. We used the amount of food consumed in grams for sensitivity analysis; the results remained unchanged from the results that are presented in this paper. Thus, we considered that energy intake represented in this study can reflect the amount of food consumed despite the limitations of the food database. Another limitation is seasonal differences of dietary intake. Since KNAHNES was conducted in different periods by each cycle, seasonal differences in food intake may impede the estimation of real changes in food intake. In general, candy sales increase at the beginning and end of the year in South Korea because candy is often used for gifts (Korea Ministry of Agrigulture, Food and Rural Affairs, & Korea Food Industry Association, 2013a). Thus energy intake from candy in 1998 and 2001, when KNHANES was conducted during November and December, may be overestimated and the decreasing trend of energy intake from candy from 1998 to 2005 shown in the study would be smaller once the seasonal variability is taken into consideration. Among Korean adults, seasonal variation on food intake is generally small



(D. W. Kim et al., 2013). We found some consistent decreasing or increasing trends of food intake over time regardless of the different survey periods, thus we believe that seasonal variability of food intake does not significantly affect the study results. Social desirability bias may affect adolescents' response regarding intake of unhealthy foods, which are target foods of the policies. For example, adolescents may be more aware of the nutrition policies and report lower intake of the targeted foods. These do not necessarily affect the study results, however, because our purpose was to observe population-level dietary change trends rather than to examine individual-level food intake (National Center for Health Statistics, 1992). The statistical analysis we did was not hypothesis-driven because of the uncertainties of policy implementation across individual school stores.

We considered only two policies in South Korea related to the school food environment. Other policies like nutrition labeling and reductions in sodium in processed foods have been implemented during the study period. The 2009 policy originally intended to improve out-of-school food environments by prohibiting EDNP foods sales in "volunteer" stores located nearby schools, and restricting advertisement of EDNP foods on television during 5-7 pm beginning September 2010. Given that only 4.2% of stores near schools voluntarily participated in stopping the sale of EDNP foods as of 2012 (Korea Ministry of Food and Drug Safety, 2013), and we did not find a notable change in intake of specific foods after restriction of EDNP foods advertisement on television, we considered that other components of the 2009 policy mentioned above did not affect adolescents' diets during the study period. It is possible that the food industry increased



other forms of marketing to adolescents such as advertisements through the Internet, video-games, or point-of-sale (Dietz, 2013; Henriksen, 2012).

Conclusions

Our analysis highlights that attention to the diverse eating places, including school, at-home, and away from home, are important to consider when developing and evaluating policies that aim to improve adolescents' overall dietary intake. This can provide an implication not only in South Korea, but also other countries that develop and implement nutrition policies to improve adolescents' diet and health. Existing policies and previous efforts to improve adolescents' diets in South Korea mostly focused on inschool (including school store) consumption, with no policy to improve at-home diet (S. G. Bae et al., 2012). Policies regulating the school food environment may be able to change in-school consumption, but may not be enough to improve overall diet (Briefel, Crepinsek, et al., 2009; Briefel, Wilson, et al., 2009; Taber et al., 2012; Taber, Chriqui, Vuillaume, & Chaloupka, 2014). Policies that restrict one setting (e.g., school) are not enough to improve adolescents' diet (Taber et al., 2014; Vecchiarelli et al., 2006) possibly because adolescents can easily access unhealthy foods outside of school or at home (Briefel, Crepinsek, et al., 2009). Comprehensive policies and interventions targeting multiple settings and both parents' and adolescents' dietary behaviors are needed (Briefel, Wilson, et al., 2009). Policies are more likely to be effective when they are consistent with previous efforts and effectively implemented.



References

- Bae, S. G., Kim, J. Y., Kim, K. Y., Park, S. W., Bae, J., & Lee, W. K. (2012). Changes in dietary behavior among adolescents and their association with government nutrition policies in Korea, 2005-2009. *Journal of Preventive Medicine & Public Health*, 45(1), 47-59. doi:10.3961/jpmph.2012.45.1.47
- Bae, Y.-J., & Yeon, J.-Y. (2013). Evaluation of nutrient intake and diet quality according to beverage consumption status of elementary school, middle school, and high school students: from the Korean National Health and Nutrition Examination Surveys, 2007-2008. *Korean Journal of Nutrition*, 46(1), 34-49.
- Briefel, R. R., Crepinsek, M. K., Cabili, C., Wilson, A., & Gleason, P. M. (2009). School food environments and practices affect dietary behaviors of US public school children. *Journal of the American Dietetic Association*, 109(2), S91-S107. doi:10.1016/j.jada.2008.10.059
- Briefel, R. R., Wilson, A., & Gleason, P. M. (2009). Consumption of low-nutrient, energy-dense foods and beverages at school, home, and other locations among school lunch participants and nonparticipants. *Journal of the American Dietetic Association*, 109(2), S79-S90. doi:10.1016/j.jada.2008.10.064
- Choi, S. K., Frongillo, E. A., Blake, C. E., & Thrasher, J. F. (2016). Comparison of food availability in school stores in Seoul, South Korea before and after implementation of two national food- and nutrient-based policies. *Faseb Journal*, 30(1), 129.127.
- Choi, S. K., Frongillo, E. A., Blake, C. E., Thrasher, J. F., & Tompkins, M. E. (2015). Why do school stores in South Korea still sell restricted foods four years after the school store policy? *Faseb Journal*, 29(1), 741.749.
- Chriqui, J. F., Pickel, M., & Story, M. (2014). Influence of school competitive food and beverage policies on obesity, consumption, and availability: a systematic review. *JAMA Pediatrics*, 168(3), 279-286. doi:10.1001/jamapediatrics.2013.4457
- Cleveland, W. S. (1979). Robust locally weighted regression and smoothing scatterplots. *Journal of the American Statistical Association*, 74(368), 829-836.
- Cullen, K. W., Watson, K., & Zakeri, I. (2008). Improvements in middle school student dietary intake after implementation of the Texas public school nutrition policy. *American Journal of Public Health*, *98*(1), 111-117.
- Dietz, W. H. (2013). New strategies to improve food marketing to children. *Health Affairs*, *32*(9), 1652-1658.
- Finkelstein, E., French, S., Variyam, J. N., & Haines, P. S. (2004). Pros and cons of proposed interventions to promote healthy eating. *American Journal of Preventive Medicine*, 27(3), 163-171.



www.manaraa.com

- Gittelsohn, J. (2012). Interventions in small food stores to change the food environment, improve diet, and reduce risk of chronic disease. *Preventing Chronic Disease*, 9, E59.
- Hanson, N. I., Neumark-Sztainer, D., Eisenberg, M. E., Story, M., & Wall, M. (2005). Associations between parental report of the home food environment and adolescent intakes of fruits, vegetables and dairy foods. *Public Health Nutrition*, 8(1), 77-85.
- Hebert, J. R., Hurley, T. G., Steck, S. E., Miller, D. R., Tabung, F. K., Peterson, K. E., . . . Frongillo, E. A. (2014). Considering the Value of Dietary Assessment Data in Informing Nutrition-Related Health Policy. *Advanced Nutrition*, 5(4), 447-455. doi:10.3945/an.114.006189
- Henriksen, L. (2012). Comprehensive tobacco marketing restrictions: promotion, packaging, price and place. *Tobacco Control, 21*(2), 147-153.
- Kim, D. W., Park, K., Kim, J., Oh, K., Joung, H., Lee, J. E., & Paik, H. Y. (2013). Sources of variation in nutrient intake and the number of days to assess usual intake among men and women in the Seoul metropolitan area, Korea. *The British Journal of Nutrition*, 110(10), 2098-2107.
- Kim, H. (2014). Changes in the nutrient contents of children's favorite foods after policy implementation on Energy-Dense and Nutrient-Poor foods in Korea (Masters's thesis), Retreived from http://www.riss.kr/link?id=T13438993.
- Kim, H. R. (2008). A strategy for child and adolescent health promotion: nutrition, overweight and obesity. *Health and Welfare Policy Forum*, 141(4), 20-31.
- Kim, K., Park, S. M., & Oh, K. W. (2013). The impact of nutritional policy on socioeconomic disparity in the unhealthy food intake among Korean adolescents. *Appetite*, 71, 388-395. doi:10.1016/j.appet.2013.09.010
- Korea Centers for Disease Control and Prevention. Korea National Health and Nutrition Examination Survey website. Available at: http://knhanes.cdc.go.kr. Accessed January 4, 2014.
- Korea Food Industry Association, & Chung-Ang University. (2011). Processed food market report: Soft drinks (Korea Agro-Fisheries & Food Trade Corp. Publication). Naju: Korea Agro-Fisheries & Food Trade Corp.
- Korea Ministry of Agrigulture, Food and Rural Affairs, & Korea Food Industry Association. (2013a). *Processed food market report: Chocolate*. (Korea Agro-Fisheries & Food Trade Corp. Publication). Naju: Korea Agro-Fisheries & Food Trade Corp.
- Korea Ministry of Agrigulture, Food and Rural Affairs, & Korea Food Industry Association. (2013b). *Processed food market report: Milk* (Korea Agro-Fisheries



& Food Trade Corp. Publication). Naju: Korea Agro-Fisheries & Food Trade Corp.

Korea Ministry of Education. 2015. The status of school meal service in 2014. Seoul.

- Korea Ministry of Education & Human Resources Development. (2007, December 26). Status of banning soft drink sales in schools. Retrived from http://www.moe.go.kr/web/110501/ko/board/view.do?bbsId=348&boardSeq=609 67
- Korea Ministry of Food and Drug Safety. (2013, March 21). 2013 work report for the president: Become a power of the world in food safety by food safety management. Retrieved from http://www.mfds.go.kr/index.do?mid=664&pageNo=24&seq=19971&cmd=v
- Korea Ministry of Food and Drug Safety. (2014, July 18). Ongoing efforts to reduce sodium contents of foods. Retrieved from http://www.mfds.go.kr/index.do?mid=675&seq=24560&cmd=v
- Korea Ministry of Health and Welfare, & Korea Centers for Disease Control and Prevention. (2013a). Guideline for Korea National Health and Nutrition Examination Survey (KNAHENS V, 2010-2012) data use. Osong: Korea Centers for Disease Control and Prevention.
- Korea Ministry of Health and Welfare, & Korea Centers for Disease Control and Prevention. (2013b). Korean Health Statistics 2012: Korea National Health and Nutrition Examination Survey (KNHANES V-3). Osong: Korea Centers for Disease Control and Prevention.
- Korea National Youth Commission. (2006, March 28). 255 youth training centers nationwide will not sell soft drinks (Press Release).
- Kweon, S., Kim, Y., Jang, M. J., Kim, K., Choi, S., Chun, C., . . . Oh, K. (2014). Data resource profile: the Korea National Health and Nutrition Examination Survey (KNHANES). *International Journal of Epidemiology*, 43(1), 69-77. doi:10.1093/ije/dyt228
- Larson, N., & Story, M. (2010). Are 'competitive foods' sold at school making our children fat? *Health Affairs*, 29(3), 430-435.
- Lee, Y., Shim, J. E., & Yoon, J. (2012). Change of children's meal structure in terms of temporal and spatial dimensions: analysis of the data from the Korea National Health and Nutrition Examination Surveys of 1998 and 2009. *Korean Journal of Community Nutrition, 17*(1), 109-118.
- National Center for Health Statistics. (1992). *Dietary methodology workshop for the third National Health and Nutrition Examination Survey* (Vol. 4). Hyattsville, MD: National Center for Health Statistics.



- Samuels, S. E., Lawrence Bullock, S., Woodward-Lopez, G., Clark, S. E., Kao, J., Craypo, L., . . . Crawford, P. B. (2009). To what extent have high schools in California been able to implement state-mandated nutrition standards? *Journal of Adolescent Health*, 45(3), S38-S44.
- Shroff, M. R., Jones, S. J., Frongillo, E. A., & Howlett, M. (2012). Policy instruments used by states seeking to improve school food environments. *American Journal of Public Health*, 102(2), 222-229.
- Song, D. Y., Park, J. E., Shim, J. E., & Lee, J. E. (2013). Trends in the major dish groups and food groups contributing to sodium intake in the Korea National Health and Nutrition Examination Survey 1998-2010. *Korean Journal of Nutrition, 46*(1), 72-85.
- Story, M., Kaphingst, K. M., & French, S. (2006). The role of schools in obesity prevention. *Future Child*, 16(1), 109-142.
- Taber, D. R., Chriqui, J. F., Powell, L. M., & Chaloupka, F. J. (2012). Banning all sugarsweetened beverages in middle schools: reduction of in-school access and purchasing but not overall consumption. *Archives of Pediatrics and Adolescent Medicine*, 166(3), 256-262.
- Taber, D. R., Chriqui, J. F., Vuillaume, R., & Chaloupka, F. J. (2014). How state taxes and policies targeting soda consumption modify the association between school vending machines and student dietary behaviors: a cross-sectional analysis. *PloS One*, 9(8), e98249. doi:10.1371/journal.pone.0098249
- U.S. Department of Agriculture, Food and Nutrition Service. Interim final rule. 7 CFR Parts 210 and 220. National school lunch program and school breakfast program: Nutrition standards for all foods sold in school as required by the Healthy, Hunger-Free Kids Act of 2010. Federal Register. 2013; 78 (125).
- Vatanparast, H., Lo, E., Henry, C. J., & Whiting, S. J. (2006). A negative trend in calcium intake was accompanied by a substitution of milk by noncarbonated soft drinks in Canadian female students. *Nutrition Research*, *26*(7), 325-329.
- Vecchiarelli, S., Takayanagi, S., & Neumann, C. (2006). Students' perceptions of the impact of nutrition policies on dietary behaviors. *Journal of School Health*, 76(10), 525-531. doi:10.1111/j.1746-1561.2006.00153.x
- Woodward-Lopez, G., Gosliner, W., Samuels, S. E., Craypo, L., Kao, J., & Crawford, P.
 B. (2010). Lessons learned from evaluations of California's statewide school nutrition standards. *American Journal of Public Health*, 100(11), 2137-2145.
- World Health Organization. (2010). Set of recommendations on the marketing of foods and non-alcoholic beverages to children. Geneva: World Health Organization.



www.manaraa.com

Yoon, J., Kwon, S., & Shim, J. E. (2012). Present status and issues of school nutrition programs in Korea. *Asia Pacific Journal of Clinical Nutrition*, 21(1), 128-133.


4.2 Manuscript 2

COMPARISON OF FOOD AVAILABILITY IN SCHOOL STORES IN SEOUL, SOUTH KOREA BEFORE AND AFTER IMPLEMENTATION OF TWO NATIONAL FOOD- AND NUTRIENT-BASED POLICIES²

² Choi, S.K., Frongillo, E.A., Blake, C.E., and Thrasher, J.F. Submitted to *Journal of School Health*, 4/11/2016



ABSTRACT

BACKGROUND

To improve school store food environments, the South Korean government implemented two policies restricting unhealthy food sales in school stores. A food-based policy enacted in 2007 restricts specific food sales (soft drinks); and a nutrient-based policy enacted in 2009 restricts energy-dense, nutrient-poor (EDNP) food sales. The purpose of the study was to assess how the two policies have changed the school store food environment.

METHODS

Foods sold in school stores in Seoul, South Korea were observed before (2006, 15 stores) and after (2013, 12 stores) implementation of the school store policies. Food availability in school stores in 2006 and 2013 was compared and EDNP food availability in 2013 was examined.

RESULTS

When controlling the total number of foods sold in school stores and school characteristics, the mean number of soft drinks sold in a school store in 2013 (0.3 items) was significantly lower than in 2006 (1.9 items, p=0.032). Soft drinks were still available in 50% of school stores observed in 2013, with all school stores selling EDNP foods in 2013.

CONCLUSIONS

South Korean policies have had a modest influence on availability of unhealthy school store foods. Alternative strategies to improve school store food environments are needed. **Keywords:** Health Policy, Nutrition & Diet, School Food Services



School food environments affect students' dietary behaviors since students are in school most of the time they are awake, having at least one meal or snack while there (Briefel, Wilson, et al., 2009; Story et al., 2006). Efforts to improve school food environments have been mostly focused on the school meal service and other foods sold in school stores or vending machines. Increasingly, countries have attempted to regulate foods outside of the school meal program, in particular, because those foods are often energy-dense and nutrient-poor (EDNP). Consumption of unhealthy foods outside of the school meal program affects the diets of students during school hours (Cullen & Zakeri, 2004; Templeton et al., 2005; Vericker, 2013).

Policies targeting foods outside of the school meal program attempt to either limit specific food items or set a standard for nutrient content of the food. The first type of policy, the food-based policy, is most often used to restrict sales of soft drinks or fried foods. The second type of policy, the nutrient-based policy, is more often used to limit foods with high energy, fat, sugar, or sodium. Both types of policy can decrease the availability of the unhealthy foods in schools (Chriqui et al., 2013; Samuels et al., 2009) although the reduction in unhealthy food availability varies across studies (Peart et al., 2012; Samuels et al., 2009; Woodward-Lopez et al., 2010). Less is known about which type of policy most improves school food environments and whether policy impacts differ by school characteristics. Knowing policy impacts and the conditions under which they are most effective could help in developing and implementing more effective policies in the future. Two South Korean policies enacted to improve the nutritional quality of foods in school stores provide an opportunity to compare the two types of policy.



In South Korea, school children can obtain foods from school meal services, school stores, and outside of school. All elementary, middle, and high schools in South Korea hire registered dietitians, thus school meals are provided under the supervision of dietitians. In contrast, there was a lack of supervision over school store foods, which students themselves choose, until the early 2000s. Since public discomfort with unhealthy foods in school stores and students' unhealthy food choices had been growing, the Korean government implemented two national policies to improve the quality of foods sold in school stores. The Korean Ministry of Education announced a food-based policy in February 2007 (hereafter 2007 policy). The 2007 food-based policy forbade soft drink sales and recommended stop selling instant noodles and fried food in schools including vending machines and school stores by the onset of the 2007 school year (hereafter 2007 policy). A second comprehensive nutrition policy, the Special Act on Safety Management of Children's Dietary Life, targeted multiple nutritional problems of Korean children and was implemented beginning in March 2009 (hereafter 2009 policy). One component of the 2009 policy is a nutrient-based policy that restricts EDNP foods sales in school stores. The 2009 policy set a nutrient standard for energy, protein, saturated fat, sugar, and/or sodium to define EDNP foods and beverages, forbidding their sale in school stores after March 2009 (the beginning of the 2009 school year).

Previous studies regarding school nutrition policies mostly focused on the implementation of policy in public schools (Chriqui et al., 2013; Peart et al., 2012; Samuels et al., 2009; Templeton et al., 2005). Public and private schools often have different school management systems, including operation of school stores. In South Korea, 80% of public schools contract with school stores through a public competitive



bidding process, whereas 63% of private schools have private contracts (H. Lee et al., 2011). For schools using public competitive bidding, the right of operation is sold to the highest bidder (i.e., pays the school the most for school food sales rights). The right of operation expires every year, thus the current school store owner needs to participate in the bidding annually. Store owners in public schools report that high bidding prices are overly burdensome (Choi et al., 2015).

These differences in school store management systems may affect school store owners' selection of foods sold in school stores. School store owners who already paid a high cost for the right to the store need to make more profits; they may want to sell highprofit products regardless of the nutritional quality of the food. Thus, we hypothesized that school stores in public and private schools sell different types of foods and different numbers of foods that policies banned to sell. Since public and private high schools in South Korea have the same tuition (excluding specialized schools; tuition for middle schools is free), attendance in private or public schools does not reflect students' wealth. Thus, comparison of food availability between public and private schools in South Korea could provide an example of impact of school management systems on food availability in school stores.

The purpose of the study was to assess how the two different national school store policies change the school store food environments. We compared foods and beverages sold in school stores in 2006 and 2013 to assess if the 2007 food-based policy affected availability of foods and beverages in school stores. EDNP food availability in 2013 was examined to assess impacts of the 2009 nutrient-based policy. We also explored differences in food availability in school stores by school characteristics that might affect



foods sold in school stores. We hypothesized that students' age, sex, and their household income, that are related to eating behaviors and snack purchase (Briefel, Crepinsek, et al., 2009; Choi et al., 2008), and the school management type (public/private) could be related to foods and beverages sold in school stores.

METHODS

Participants and Procedure

School stores in middle and high schools located in Seoul, South Korea were observed before (school year 2006) and after (school year 2013) implementation of school store policies. In 2006, one middle school and one high school were randomly selected from each school district among 11 school districts in Seoul, South Korea. Of the 22 schools, 15 agreed to observation of their school store. In 2013, six school stores each from one high-income school district and one low-income school district were purposefully selected to make selected schools have various characteristics in terms of students' sex, school level, and school type (private, public). Among the total of 12 school stores observed in 2013, two school stores were also observed in 2006. The observations were conducted from December 2006 to February 2007 and from October 2013 to December 2013.

Four observers conducted school store observation in 2006. In 2013, one observer, who also participated in the observation in 2006, conducted all school store observation with the same material and method in 2006. When an observer visited school stores, all types of foods and beverages sold in the school stores were recorded. Product name, manufacturer, size, and flavor (if varied) were recorded. When school stores



owners allowed, the observer took photographs of school stores and food items. For foods in school stores observed in 2013, nutrient content was recorded from the nutrition label of products, including calories, protein, saturated fat, sugar, and sodium content per one serving. In most cases, it was not possible to record nutrient content for all foods during the school store observation, because school store owners did not want the observer to stay in their school stores for a long time. If nutrient information was not obtained by on-site observation, the information was obtained from the manufacturers' websites or by purchasing the product.

Data Analysis

All foods and beverages were categorized into soft drinks, fruit and vegetable juice, tea, coffee, sports drinks, bottled water, other beverages, chips & cookies, candy, ice cream, bread, hamburger & pizza, fruit, other foods, yoghurt, milk, and flavored milk. Milk and flavored milk were further categorized based on the fat content (full- or reduced-fat). Based on the 2009 criteria for EDNP (Table 4.4), all food and beverage items available in school stores in 2013 were determined to be either EDNP or not.

Descriptive statistics for school characteristics, including school level (middle school, high school), school type (private, public), students' sex (coeducational, boys only, girls only), and income level of school district (high-income, low-income), were computed using Fisher's exact test to compare characteristics of school stores observed in 2006 and 2013. Crude numbers of all types of foods and beverages sold in school stores and of foods in each category were compared between 2006 and 2013 using Student's t-tests. To compare the number of foods between 2006 and 2013 controlling for school characteristics, mixed models were fitted with school districts as random effects. Then,



Table 4.4 Criteria of the energy-dense nutrient-poor (EDNP) foods

A. Criteria for snacks among children's favorite foods^a

- (5) Foods containing over 250kcal and less than 2g of protein per serving.
- (6) Foods containing over 4g of saturated fat and less than 2g or protein per serving.
- (7) Foods containing over 17g of sugar and less than 2g of protein per serving.
- (8) Foods containing over 500kcal or over 8g of saturated fat or over 34g of sugar per serving among foods that do not meet criteria of 1~3.
- * For foods which the serving size is less than 30g, the criteria is applied to 30g of the foods instead of the actual serving size.

B. Criteria for meal substitutes among children's favorite foods

- (1) Foods containing over 500kcal and less than 9g of protein per serving.
- (2) Foods containing over 500kcal and over 600mg of sodium per serving. For deep-fried noodles and noodles among noodles (only applied to noodles in containers), over 1,000mg of sodium is applied to as a criteria of EDNP.
- (3) Foods containing over 4g of saturated fat and less than 9g of protein per serving.
- (4) Foods containing over 4g of saturated fat and over 600mg of sodium per serving. For deep-fried noodles and noodles among noodles (only applied to noodles in containers), over 1,000mg of sodium is applied to as a criteria of EDNP.
- (5) Foods containing over 1,000kcal or over 8g of saturated fat per serving among foods that do not meet criteria of 1~4.

^a Children's favorite foods are defined as "Foods that children prefer or eat frequently among foods under the Food Sanitation Act or the Livestock Products Sanitary Control Act, which are prescribed by Presidential Decree" (the Special Act on safety management of children's dietary life. Act no. 119882013). The list of children's favorite foods is in Appendix B.

the total number of foods sold in school stores was included in the models in addition to

school characteristics, to examine food composition in school stores in 2006 and 2013.

Student's t-test was used to compare crude number of total foods and EDNP foods, and

percentage of EDNP in 2013 by school level, school type, students' sex, and income level

of school district. Mixed models with school districts as random effects were used to



control school characteristics. The α level of 0.05 was used to determine statistical significance. All statistical analyses were conducted using SAS 9.4 (SAS Institute, Inc., Cary, NC).

RESULTS

On average, 40.7% of school stores were located in middle schools and 33.3% were in private schools (Table 4.5). Most school stores observed in 2006 were located in public schools (86.7%) while 41.7% of school stores observed in 2013 were located in public schools (p=0.037). More than half, 55.6% of the school stores were located in coeducational schools, 33.3% in boys' schools, and 11.1% in girls' schools. About half of the school stores were located in low-income school districts.

The mean number of food and beverage types sold in school stores was significantly higher in 2013 (102.3 items) than in 2006 (41.1 items; p<0.001), even after adjusting for school characteristics (p=0.022, Table 4.6). Within most food categories, mean numbers of food types sold in school stores observed in 2013 were significantly higher than those sold in school stores observed in 2006 (p<0.05). After adjusting for school characteristics, the mean number of some foods that can be considered healthy or low-calorie foods, such as tea, bottled water, other beverages, fruit, yoghurt, and reduced- fat non-flavored milk, and high-calorie foods including chips & cookies and hamburger & pizza, sold in 2013 was significant higher than in 2006 (p<0.05). When controlling for school characteristics and total number of food and beverage items, only the mean number of other beverages and hamburger & pizza sold was significantly higher in 2013 (p=0.037 and p=0.005, respectively), and the number of soft drinks, the target food of the



							N (%)
	School Year						
	Total		2	006	2013		na
	(n	i=27)	(n=15)		(n=12)		р
School level							
Middle school	11	(40.7)	6	(40.0)	5	(41.7)	>0.99
High school	16	(59.3)	9	(60.0)	7	(58.3)	
School type							
Private	9	(33.3)	2	(13.3)	7	(58.3)	0.037
Public	18	(66.7)	13	(86.7)	5	(41.7)	
Students' sex							
Coeducational	15	(55.6)	10	(66.7)	5	(41.7)	0.110
Boys only	9	(33.3)	5	(33.3)	4	(33.3)	
Girls only	3	(11.1)	0	(0.0)	3	(25.0)	
School district							
Low income district	14	(51.9)	8	(53.3)	6	(50.0)	0.863
High income district	13	(48.2)	7	(46.7)	6	(50.0)	

Table 4.5 Characteristics of school stores observed in 2006 and 2013

^a Fisher's exact test

was significantly lower in 2013 (p=0.032).

The 2009 nutrient-based policy seemed ineffective in prompting removal of restricted foods in school stores. All school stores observed in 2013 sold EDNP foods that were restricted by the 2009 policy. On average, 7.0 items out of 102.3 items (7.6%) sold in a school store were EDNP products (Table 4.7). School stores located in high schools and in private schools sold significantly higher number of food items (130.9 items for high schools an130.0 items for private schools) than those located in middle schools and in public schools (62.4 items for middle schools, 63.6 items for public schools; p=0.012 and p=0.016, respectively). The number of EDNP foods was slightly higher for school stores in private schools than for those in public schools (p=0.070). School stores in high-income school districts sold more food items than those in low-



	Crude					Adjusted for school characteristics ^a				Adjusted for school characteristics and total number of food and drink items ^b		
School Year	20	006	2	2013		2006	2013		2006	2013		
	Number of schools ^c	Mean±SD	Number of schools ^c	Mean±SD	р	Mean±SE	Mean±SE	р	Mean±SE	Mean±SE	р	
Total	15	41.1±27.1	12	102.3±50.7	0.001	59.7±12.5	98.5±10.7	0.022				
Soft drinks	9	1.8±1.9	6	0.8±0.9	0.070	1.5±0.5	0.7±0.9	0.433	1.9±0.5	0.3±0.5	0.032	
Fruit and vegetable juice	14	3.2±2.3	12	7.1±5.6	0.041	5.9±1.2	6.7±1.0	0.590	6.5±1.0	4.8±0.9	0.249	
Tea	6	0.7±1.1	11	4.3±3.3	0.004	$1.4{\pm}0.8$	4.1±0.6	0.007	1.7±0.7	3.0±0.6	0.143	
Coffee	4	0.4 ± 0.8	2	0.2 ± 0.4	0.345	0.3±0.3	0.2±0.3	0.728	0.4 ± 0.3	0.0 ± 0.4	0.356	
Sports drinks	14	1.9±1.1	11	3.5 ± 2.9	0.087	1.7±0.6	2.9±0.7	0.196	1.9±0.5	2.2±0.6	0.783	
Bottled water	2	0.1±0.4	8	0.8 ± 0.6	0.003	0.2 ± 0.2	0.7±0.1	0.031	0.2 ± 0.2	0.5±0.2	0.211	
Other beverages	12	1.9±1.8	12	8.3±4.9	0.001	2.5±1.2	7.9±1.1	0.002	3.2 ± 0.8	5.5±0.7	0.037	
Chips & cookies	15	12.4±8.7	12	34.3±19.6	0.003	18.0 ± 5.2	33.5±3.9	0.020	21.2±2.0	21.9±1.9	0.808	
Candy	12	6.5±6.3	12	16.4 ± 10.0	0.004	8.9±3.1	15.7±3.3	0.133	11.0 ± 1.2	8.8±1.1	0.198	
Ice cream	10	4.2±4.5	11	11.8±7.6	0.004	8.2±1.9	11.2±1.5	0.204	8.9±1.7	8.9±1.6	0.982	
Bread	13	2.6±1.4	11	3.0 ± 2.2	0.573	3.2±0.7	3.0±0.5	0.793	3.3±0.7	2.8±0.7	0.622	
Hamburger & pizza	11	1.4±1.4	12	4.7±2.9	0.003	1.3±0.7	4.5±0.5	0.001	1.3±0.7	4.5±0.7	0.005	
Fruit	0	0.0 ± 0.0	2	0.2 ± 0.4	0.108	-0.1±0.1	0.2±0.1	0.042	-0.1±0.1	0.0±0.1	0.468	
Other foods	6	1.7±2.7	8	1.3±1.3	0.549	2.1±1.0	1.2 ± 1.8	0.653	2.4 ± 0.9	0.4±1.6	0.288	
Yoghurt	1	0.1±0.3	7	1.1±1.2	0.003	0.4±0.3	1.1±0.2	0.041	0.5±0.2	0.7±0.2	0.529	
Milk	5	0.3±0.5	8	0.8 ± 0.6	0.062	0.6±0.2	0.7±0.1	0.690	0.6±0.2	0.6±0.2	0.827	
Flavored milk	13	1.8±1.1	12	4.3±2.4	0.002	3.0±0.6	4.1±0.5	0.130	3.2±0.5	3.6±0.5	0.547	

Table 4.6 Mean number of foods and beverages sold in school stores in 2006 and 2013

	Crude					Adjusted for characteristic	r school ics ^a	Adjusted for school characteristics and total number of food and drink items ^b			
School Year	2	006	2	013		2006	2013		2006	2013	
	Number of schools ^c	Mean±SD	Number of schools ^c	Mean±SD	р	Mean±SE	Mean±SE	р	Mean±SE	Mean±SE	р
Milk categorization											
Non-flavored milk, full-fat	5	0.3±0.5	6	0.6±0.7	0.272	0.7±0.2	0.5±0.2	0.507	0.7±0.2	0.5±0.2	0.528
Non-flavored milk, reduced fat	0	0.0±0.0	2	0.2±0.4	0.108	-0.1±0.1	0.2±0.1	0.042	2.4±0.5	2.9±0.6	0.563
Flavored milk, full-fat	11	1.2±1.0	11	3.0±1.9	0.004	2.4±0.5	2.9±0.6	0.485	-0.1±0.1	0.0±0.1	0.468
Flavored milk, reduced fat	8	0.6±0.6	9	1.3±1.1	0.058	0.7±0.4	1.2±0.3	0.224	0.8±0.3	0.9±0.3	0.870

Table 4.6 Mean number of foods and beverages sold in school stores in 2006 and 2013 (continued)

SD: standard deviation; SE: standard error

^a Adjusted for school level (middle, high school), school type (public, private), students' sex (coeducational, boys only, girls only), and income level of the school district. School districts were considered as random effects.

^b Same as above but with additional adjustment for total number of food items in the school store.

^C Number of schools sold the food item



	Crude					Adjusted for school characteristics ^a						
	Total number of foods		Number of I foods	EDNP	JP % EDNP foods ^b		Number of foods		Number of EDNP foods		% EDNP foods	
	Mean±SD	р	Mean±SD	Р	Mean±S D	р	Mean±SE	р	Mean±SE	р	Mean±SE	р
All schools	102.3 ± 50.7		7.0±3.6		7.6±2.8		102.7±23.9		7.8±1.9		7.8±1.8	
School level												
Middle school	62.4±21.8	0.012	5.2±1.6	0.150	9.1±2.6	0.114	77.4 ± 20.8	0.147	5.7±1.6	0.402	8.8±1.6	0.354
High school	130.9±46.1		8.3±4.2		6.5 ± 2.6		122.2±14.6		7.6±1.1		6.7±1.1	
School type												
Private	130.0±47.2	0.016	8.6±3.7	0.070	7.1±2.5	0.478	$121.0{\pm}14.5$	0.042	8.5±1.1	0.060	7.8 ± 1.1	0.986
Public	63.6±22.7		4.8 ± 2.2		8.3±3.4		70.1±16.5		4.8±1.3		7.8 ± 1.2	
Students' sex												
Coeducational	72.6±33.4	0.197	6.2±3.1	0.176	9.3±2.6	0.109	91.5±17.4	0.819	7.2±1.3	0.127	8.6±1.3	0.287
Boys only	113.3±62.2		5.5 ± 2.9		5.4±2.7		89.3±18.8		4.1±1.5		5.9±1.4	
Girls only	137.3±42.7		10.3±4.0		7.8±1.5		105.8 ± 23.6		8.7±1.9		8.8 ± 1.8	
School district												
Low income district	76.7±40.5	0.076	5.8±4.3	0.281	8.0±3.4	0.598	85.0±14.1	0.316	6.4±1.2	0.737	7.7±1.2	0.947
High income district	128.2±49.2		8.2±2.6		7.1±2.4		109.8±17.0		7.1±1.5		7.9±1.4	

Table 4.7 Availability of energy-dense nutrient-poor (EDNP) foods in school stores in 2013

SD: standard deviation; SE: standard error

^a Among school level (middle, high school), school type (public, private), students' sex (coeducational, boys only, girls only), and income level of the school district, school characteristics except the analytic characteristic were adjusted. School districts were considered as random effects.

^b Percentage of EDNP foods may differ the result of calculation that number of EDNP foods divided by total number of foods since only specific food groups are eligible to be determined EDNP or non-EDNP foods.



-income school districts (p=0.076), but there was no difference in number of EDNP foods. After controlling for school characteristics, the differences in the total number of foods, total number of EDNP foods, and percentage of EDNP foods by school characteristics became smaller. The total number of food items and total number of EDNP foods sold in school stores was higher in those located in private schools than in public schools after adjusting for other school characteristics (p=0.042 and p=0.060, respectively). Although the number of EDNP foods was slightly higher in school stores located in private schools (p=0.060), when school characteristics were adjusted, the percentages of EDNP foods were not differ by school level, school type, students' sex, or income level of the school district.

DISCUSSION

We examined changes in the foods and beverages available in school stores before and after implementation of the school store policies. The school stores observed in 2013, after the implementation of the 2007 policy that restricts soft drink sales, had significantly less soft drinks than those in 2006. Yet, half of the school stores observed in 2013 still sold soft drinks. Although the 2009 policy forbids EDNP foods sales in school stores, EDNP foods were found in all school stores observed in 2013, representing about 7.6% of foods sold.

While the quantity of food types sold in school stores has increased, the quality of school store foods has not changed much between 2006 and 2013. Despite the implementation of policies intending to improve school store food environments, there was no significant difference in the foods offered in school stores between 2006 and



www.manaraa.com

2013, when the total number of food in school stores was adjusted, with one exception: the decrease of soft drinks. This change was likely the result of the 2007 policy that bans sales of soft drinks. The increased number of other types of beverages may be due in part to substitution for soft drinks. On the other hand, the number of hamburger & pizza options sold in school stores was significantly higher in 2013 than in 2006, even after adjusting for the total number of food items sold in school stores. Hamburger & pizza products that were observed in 2013 in this study had the highest energy and sodium content per serving among all food categories (data not shown). All hamburger & pizza foods, however, had higher protein and lower energy, saturated fat, and sugar per serving than the EDNP criteria of the 2009 policy, and thus are allowed to be sold. Previous studies have reported that schools tend to remove only the least healthy foods that school nutrition policies require, but not replace those unhealthy items with healthy items (Samuels et al., 2009; Taber et al., 2015; Woodward-Lopez et al., 2010). Thus policies regulating unhealthy food sales may not contribute to increase of healthy food availability in schools (Fernandes, 2013; Peart et al., 2012). The increased availability of hamburger & pizza products which are not restricted by the policies, but are still unhealthy, and insignificant increase of availability of healthy foods such as bottled water, fruit, and reduced-fat milk, shown in this study indicates that South Korean policies also have not eliminated all unhealthy items while healthy food availability is not increased.

EDNP foods were still available after the 2009 policy implementation. There are two possible explanations. First, school store owners may not know which foods are EDNP due to a lack of information about the policy or to the complexity of applying EDNP criteria. Indeed, school store owners often do not know specific EDNP criteria and



www.manaraa.com

have difficulties discerning EDNP foods (Choi et al., 2015; Samuels et al., 2009). The Korea Ministry of Food and Drug Safety (MFDS) has posted the list of EDNP products on their website monthly and developed a smart phone application that identifies foods as EDNP using the barcode on product packaging (Korea Food and Drug Administration, 2012; Korea Ministry of Food and Drug Safety, 2014a). Most school store owners did not know about the MFDS's materials, however, and the materials did not include information of all foods sold in school stores or new food products (Choi et al., 2015).

Another possible explanation for the lack of compliance that this study found is the different contract system for the school store operation between public and private schools. School stores in private schools sold more EDNP foods than those in public schools. In contrast to public schools that have an annual public bidding for school store operation, most private schools make private contracts with school store owners and the duration of the contract is usually multiple years. Even for some private schools, school store owners are relatives of the people from the school foundation. Because of their close relationship and a long history of operation, school stores of private schools are expected to experience less monitoring compared to those in public schools (Choi et al., 2015). Yet, school stores in public schools also sold EDNP foods. With consideration of the total number of food types sold in school stores, the percentage of EDNP foods among all foods in school stores was not different between public and private schools. If EDNP foods are high-selling and high-profit products, school store owners would not want to remove the items from their stores. Since the winning bid through the public competitive bidding is increased every year, school store owners of public schools need to earn more money to apply for the next year's right of operation. Under these



circumstances, it would be hard to convince school store owners to halt specific food sales.

Given that all school stores observed in 2013 sold EDNP foods, it is questionable whether monitoring on school store foods is conducted well. During the school store observation in 2013, school store owners stated that irregular monitoring of school store foods is conducted by public officials or school staff; but they have not been caught due to violation of the policy. A qualitative study reported that one of the possible reasons of lack of compliance with the 2009 policy is inadequate monitoring of school store foods, such as no monitoring by school staff and decreased frequency of monitoring by public officials due to budget limitation (Choi et al., 2015).

This study shows that the two different types of policies in South Korea had different influences in food availability in school stores. There was a significant decrease in soft drinks which the 2007 food-based policy required, but no difference in availability of other foods in school stores between 2006 and 2009 was found despite of implementation of the 2009 nutrient-based policy. The 2007 food-based policy, however, did not completely eliminate the restricted items from the observed school stores. California's competitive food policy that included both food-based (beverage standards) and nutrient-based (energy, energy from fat and saturated fat, and sugar) policies also showed that compliance was easier to achieve with food-based than nutrient-based food policies (Peart et al., 2012; Samuels et al., 2009; Woodward-Lopez et al., 2010). The greater compliance with food-based policies may be because they are easier to interpret than nutrient-based policies (Samuels et al., 2009; Woodward-Lopez et al., 2010). No prior policy evaluation has reported complete elimination of restricted foods, regardless



www.manaraa.com

of whether the restriction was made by nutrient- or food-based policies. Other factors, however, such as revenue loss, inadequate monitoring system, and students' preferences, may hinder 100% compliance of the policies (Greves & Rivara, 2006). Observation of school stores before and after the implementation of the school store policies allowed us understanding of the contexts in which these policies were implemented and helps to identify possible reasons for non-compliance with the policies. We were also able to identify where more efforts are needed for better achievement of policy intention.

Limitations

The findings of this study should be interpreted with caution due to different school stores observed over time, small sample sizes, and different characteristics of selected school stores between 2006 and 2013. Although only two school stores were observed both in 2006 and 2013, changes in the two school stores were similar with changes in overall samples; for example, the two school stores sold increased number of foods (1.9 times and 2.7 times each), less soft drinks, and more hamburger & pizza and other beverages in 2013 than in 2006. Significantly more stores in private schools were observed in 2013 than in 2006. The different contract systems may affect school store owners' food selection, but we did not find significant difference in type of foods sold in school stores between private and public schools. We did not obtain contract information, although there is a possibility that some of stores in private schools made a contract through the public competitive bidding. Future research needs to follow-up with changes in food availability in same school stores using a larger sample size in order to track the impacts of policies.



www.manaraa.com

Because of the 7-year gap between two observations, we had different data in 2006 and 2013. In 2006, nutrition labeling policy in South Korea did not require labeling of saturated fat and sugar content of food products. Thus, we were unable to compare EDNP food availability in school stores between 2006 and 2013, and it was impossible to know whether EDNP food availability changed after the 2009 policy. The same observer conducted the school store observations both in 2006 and 2013, using the same materials to record foods in school stores and photos of foods in school stores were taken to ensure accuracy of the records. Thus, we believe that reliability of observation is not a concern in this study. In addition, validity of the data obtained by the research staff's direct observation may be higher than indirect observation by school staff or school store owners. School store observations were conducted only in one city; thus foods in school stores in other cities may differ. This study may not represent availability of foods in school stores between observer, it can still provide policy implications by comparison of two different kinds of school store policy.

Conclusions

In conclusion, the two school store policies in South Korea have had a modest influence on the improvement of school store food environments. Both policies are restrictive in terms of forbidding sales of specific items or EDNP food. Restrictive policies may not change school store food environments much if unhealthy foods that meet the policy criteria are sold in school stores. A policy that requires sales of healthy foods with the restriction of unhealthy foods could be an alternative. An increase of healthy food availability and a decrease of unhealthy food availability at the same time



could improve school store food environments as well as students' diet. Future studies need to investigate how to increase healthy foods and reduce unhealthy foods in school stores and how to encourage students to buy healthy foods instead of unhealthy foods.

In addition, increased benefits for school stores may help to reduce EDNP food availability in school stores. Current policies have only restrictions that may reduce revenue of school stores, but there is no incentive for compliance to the policy. If school store owners perceive more benefits from the compliance of the policy than noncompliance, then they are more likely to comply. Financial incentives or advantages in renewal of the school store contract could increase school store owners' perceived benefits to comply with the policy. With the direct incentives, strategies to promote school store owners' understanding of why they should not sell specific foods and how to avoid revenue loss when they comply with the policy would be helpful to improve school store food environments.

Implications for school health

These results have important implications for informing school administrators for improving school store food environments. School store food environments may not be improved by only governmental policies if school administrators are not involved. School teachers, staff, and dietitians work in schools, thus they are in a better position to monitor foods in school stores than other officials. School administrators should monitor what foods are sold in their school stores and ask for school store owners to sell healthier foods. Schools have leverage to require school stores sell healthy foods when schools make a contract with school stores. In addition, schools can utilize their dietitians or



school nurses to inform students to make a healthier food choice in school stores. Furthermore, schools can seek for help from outside to improve the school store food environments. Many governments or research teams in academia run public health intervention program such as Healthy School Store Program of the Seoul Metropolitan government. Schools should play a proactive role to make healthier food environments for students in their schools.

References

- Briefel, R. R., Crepinsek, M. K., Cabili, C., Wilson, A., & Gleason, P. M. (2009). School food environments and practices affect dietary behaviors of US public school children. *Journal of the American Dietetic Association*, 109(2), S91-S107. doi:10.1016/j.jada.2008.10.059
- Briefel, R. R., Wilson, A., & Gleason, P. M. (2009). Consumption of low-nutrient, energy-dense foods and beverages at school, home, and other locations among school lunch participants and nonparticipants. *Journal of the American Dietetic Association*, 109(2), S79-S90. doi:10.1016/j.jada.2008.10.064
- Choi, S. K., Frongillo, E. A., Blake, C. E., Thrasher, J. F., & Tompkins, M. E. (2015). Why do school stores in South Korea still sell restricted foods four years after the school store policy? *Faseb Journal*, 29(1), 741.749.
- Choi, S. K., Choi, H. J., Chang, N. S., Cho, S. H., Choi, Y. S., Park, H. K., & Joung, H. (2008). Snacking behaviors of middle and high school students in Seoul. *Korean Journal of Community Nutrition*, 13(2), 199-206.
- Chriqui, J. F., Turner, L., Taber, D. R., & Chaloupka, F. J. (2013). Association between district and state policies and US public elementary school competitive food and beverage environments. *JAMA Pediatrics*, 167(8), 714-722.
- Cullen, K. W., & Zakeri, I. (2004). Fruits, vegetables, milk, and sweetened beverages consumption and access to à la carte/snack bar meals at school. *American Journal of Public Health*, *94*(3), 463-467.
- Fernandes, M. M. (2013). A national evaluation of the impact of state policies on competitive foods in schools. *Journal of School Health*, 83(4), 249-255. doi:10.1111/josh.12024



- Greves, H. M., & Rivara, F. P. (2006). Report card on school snack food policies among the United States' largest school districts in 2004–2005: Room for improvement. *International Journal of Behavioral Nutrition and Physical Activity*, 3, 1. doi:10.1186/1479-5868-3-1
- Korea Food and Drug Administration. (2012, January 11). Check your smartphone when you worry about obesity -Development and dissemination of a smartphone application to find energy-dense nutrient-poor foods. Retrieved from http://www.mfds.go.kr/index.do?mid=664&pageNo=106&seq=16829&cmd=v
- Korea Ministry of Food and Drug Safety. (2014). A list of energy-dense nutrient-poor foods that are restricted sales in schools and exemplary stores ('14.7). Retrieved from http://www.mfds.go.kr/index.do?mid=1043&seq=8228&cmd=v
- Lee, H., Jang, S., & Kim, J. (2011, September 21). Leasing school stores needs to go through open bidding. *Naeil Newspaper*. Retrieved from http://www.naeil.com/News/economy/ViewNews.asp?nnum=625384&sid=E&tid =4
- Peart, T., Kao, J., Crawford, P. B., Samuels, S. E., Craypo, L., & Woodward-Lopez, G. (2012). Does competitive food and beverage legislation hurt meal participation or revenues in high schools? *Childhood Obesity*, 8(4), 339-346.
- Samuels, S. E., Lawrence Bullock, S., Woodward-Lopez, G., Clark, S. E., Kao, J., Craypo, L., . . . Crawford, P. B. (2009). To what extent have high schools in California been able to implement state-mandated nutrition standards? *Journal of Adolescent Health*, 45(3), S38-S44.
- Story, M., Kaphingst, K. M., & French, S. (2006). The role of schools in obesity prevention. *Future Child*, 16(1), 109-142.
- Taber, D. R., Chriqui, J. F., Powell, L. M., Perna, F. M., Robinson, W. R., & Chaloupka, F. J. (2015). Socioeconomic differences in the association between competitive food laws and the school food environment. *Journal of School Health*, 85(9), 578-586.
- Templeton, S. B., Marlette, M. A., & Panemangalore, M. (2005). Competitive foods increase the intake of energy and decrease the intake of certain nutrients by adolescents consuming school lunch. *Journal of the American Dietetic Association*, 105(2), 215-220.
- Vericker, T. C. (2013). Limited evidence that competitive food and beverage practices affect adolescent consumption behaviors. *Health Education & Behavior, 40*(1), 19-23.
- Woodward-Lopez, G., Gosliner, W., Samuels, S. E., Craypo, L., Kao, J., & Crawford, P.
 B. (2010). Lessons learned from evaluations of California's statewide school nutrition standards. *American Journal of Public Health*, 100(11), 2137-2145.



4.3 Manuscript 3

WHY DO SCHOOL STORES IN SOUTH KOREA STILL SELL RESTRICTIED FOODS FOUR YEARS AFTER THE SCHOOL STORE POLICY?³

³ Choi, S.K., Frongillo, E.A., Blake, C.E., and Thrasher, J.F. To be submitted to *Journal* of the Academy of Nutrition and Dietetics



Abstract

The South Korean government has banned sales of unhealthy foods in school stores since 2009; however, most school stores still sell restricted foods. This study aimed to understand the implementation processes and the reasons for incomplete implementation of the school nutrition policy in South Korea. Semi-structured interviews were conducted with 33 policy actors including school store owners, school staff, public officials, politicians, employees of food companies, and experts. Observations of food sold in school stores were conducted. Interview transcripts were analyzed using both emergent codes and an a priori code list (participants, base values, perspectives, and effects) based on Lasswell's social process model using NVivo 10. All interviewees perceived that school stores complied with the policy well; however, restricted foods were found in all interviewed school stores during the observations. Although all policy actors understood the intent of the policy; some of them were not interested in the policy or were unsatisfied with the policy and its outcomes such as profit loss. Perspectives and values towards the policy were different for each policy actor group. Poor monitoring of school store foods, lack of awareness of the policy, profit-seeking, and lack of interest in school stores were identified as reasons for incomplete implementation of the policy. The findings support the needs for consideration of various actors' perspectives, especially for those who are affected by the policy, when implementing school nutrition policies. A stronger understanding of various actors' involvement in policy processes can inform strategies to enhance implementation and thereby reach outcomes that align with the original policy intent.

Key Words: Policy implementation, food policy, nutrition



Introduction

Many school policies aimed at improving school food environments are being enacted worldwide, but these policies often involve challenges to intended implementation (Jaime & Lock, 2009; Kingdon, 2002). Poor implementation contributes to the limited success of the intended policy objectives (Kocken, van Kesteren, Buijs, Snel, & Dusseldorp, 2015). Thus, the study of policy implementation processes is critical to understand policy processes (Birkland, 2005). Understanding implementation processes will allow policy makers and implementers to know why some policies work and others do not and to be more readily achieved policy objectives (Birkland, 2005; Hill & Hupe, 2002).

Various people and groups participate in, affect, or are affected by policy processes. Recognizing these "policy actors" is critical to understanding the policy's effects (Clark, 2002; Clarke et al., 2013). They shape implementation processes; for example, their perspectives affect how the policy is being implemented (Vine & Elliott, 2014). In addition, difficulties in satisfying different perspectives and responsibility conflicts among actors, such as expecting other actors to support the policy, may hinder implementation (Clarke et al., 2013; Green & Aarons, 2011). Various actors, from governments to individuals, are involved in implementation of school nutrition policies (Clarke et al., 2013). Yet, what policy actors do, how they interact, and what they think about the policy are often omitted from formal documents (Tansey, 2007). Thus, as direct witnesses of and participants in the policy processes, policy actors can be important resources in understanding policy implementation.



The social process model emphasizes interaction between participants and between participants and resource environments in the social process (Lasswell, 1971). The social process can be mapped using participants, perspectives, situations, base values, strategies, outcomes, and effects (Table 4.8). Elements of the social process model affect each other and shape the process. Mapping these elements is useful for understanding how, by whom, and why the social interaction occurs. As a social process, policy implementation processes can be understood using the elements of the social process model.

The South Korean government enacted the Special Act on Safety Management of Children's Dietary Life in 2009 to restrict sales of energy-dense, nutrient-poor (EDNP) foods in school stores. EDNP foods are determined based on the nutrient content of the foods per serving size. School stores are prohibited from selling EDNP foods, but despite policy implementation, EDNP foods are still available in school stores (Choi et al., 2016; S. C. Kim et al., 2010).

While several studies have been conducted to examine the effectiveness of the Korean school store policy (K. Kim et al., 2013; S. C. Kim et al., 2010), the implementation processes are relatively unknown. In this study, we traced implementation processes of the school store policy using qualitative interviews with policy actors. Interviewing policy actors enabled us to reconstruct implementation processes and to identify various factors that affect policy implementation (Tansey, 2007).

The overall aim of this study was to understand implementation processes of the Korean school store policy. Specific aims were to understand: (1) the policy



Elements of	Definition	Selected questions asked
the social process model		
Participants (policy actors)	Individuals, groups, or institutions who are interacting in the social process	Is there anyone who influences on the selection of foods sold in school stores?
Perspectives	Identity, demands, and expectations of participants	What do you do regarding school stores? What did you expect regarding the influence of the change of foods and beverages sold in (your) school store(s)?
Base values	Assets or resources that participants possess Power: making/influencing decisions Wealth: production, distribution, and control of resources Well-being: Being physically and psychologically healthy and safe Affection (relationship): having family, friendship, intimacy, warm relationships	What do you find satisfying or fulfilling about your work (or foods sold in school stores)?
Outcomes	Changes in base values as a result of the social process	How the change of foods and beverages sold in (your) school store(s) affect school store operation?
Source: Lasswe	ell (1971)	

Table 4.8 Selected interview questions relevant to the elements of the social process model

implementation processes from diverse policy actor perspectives, and (2) why the policy has not achieved complete implementation in terms of elimination of EDNP foods in school stores. This study answered the following research questions: (1) How is the policy being implemented? (2) What do different policy actors value and how are their



values reflected on their perspectives and thoughts regarding the policy and its outcomes? and (3) Why do school stores still sell EDNP foods?

Materials and Methods

This study is guided by the social process model (Lasswell, 1971). Using elements of the social process model, we attempted to understand implementation processes of the school store policy. We considered participants of the school store policy process as policy actors.

Participants

Participants in the school store policy implementation were considered as potential interviewees. We purposefully selected two school districts in Seoul, South Korea first based on the income-level (high- versus low-income). Then we selected 6 schools each to have various characteristics in students' sex composition, school level (middle-, high-schools), and school type (private, public).

Conceptually-driven sequential sampling was used, wherein initial interviewees were selected based on existing understanding, and then the rest of the interviewees were purposefully selected based on themes that emerged during the previous interviews (Miles & Huberman, 1994). We identified potential interviewees that were expected to be involved in the policy based on reviews of research papers, government reports, and news articles. The pre-identified interviewees were policy actors in schools including school store owners, principals, and school staff; and outside of schools including governments and food industry. During the interviews with pre-identified interviewees, if specific



people's roles related to policy implementation were mentioned, these people were recruited for interviews.

School store owners, as front-line implementers of the policy, were contacted for the interviews first. After the interviews with school store owners, in-school actors in the same school were asked for the interviews. We also reached out to food industry, at least one public official in community health centers (CHCs) and offices of education in each school district that we selected, and public officials in the national-level governments (e.g., Ministry of Food and Drug Safety [MFDS], Ministry of Education [MoE]) for interviews. Politicians, consumer associations, and academia were frequently mentioned during the interviews, thus, they were also contacted for interviews.

Research instrument

The social process model informed the development of a semi-structured interview guide (Table 4.7) (Clark, 2002; Lasswell, 1971). The interview guide consisted of main questions, follow-up questions, and probes into emergent topics that interviewees brought up (Ulin et al., 2004) to gain information about implementation processes focusing on policy actors, their roles, perspectives, and relationships with other actors (Clark, 2002; Lasswell, 1971). The interview guide included different questions for each actor group that were specific to their roles in implementation processes. After review of field notes and memos from the previous interviews, the interview guide was revised to include questions that emerged for subsequent interviews.

Procedure

The first author conducted all interviews. The interviewer visited school stores and asked school store owners for the interviews. The visits were made without prior



notice to prevent the possibility that school store owners might change foods sold in school stores. When a school store owner agreed to the interview, the interview was conducted in the same day of the visit. After the interview, the interviewer recorded all items sold in the school store. The details of school store observations and food availability in school stores are described elsewhere (Choi et al., 2016). For other interviewees, the interviewer contacted them via phone or email to explain the purpose of the interview. Interviews were conducted either face-to-face or by phone and lasted 15-60 minutes.

We interviewed a total of 33 people, including 11 school store owners; 8 school staff; 6 public officials in CHCs, an office of education, Seoul Metropolitan government, and MFDS; 2 politicians; 3 food company representatives; and 3 people who have been involved in school store related work (a professor, a school principal, and a consumer organization representative). Interviews were audio-recorded upon agreement of the interviewees. When interviewees disagreed on recording, the interviewer took interview notes. Eight school store owners, 3 school staff members, 1 public official, 1 politician, 1 from a food company representative, and 2 people who have been involved in school store related work were agreed to audio-record interviews, thus, a total of 16 interviews was audio-recorded. Among interviewees who did not agree to audio-record interviews, 1 politician and 1 food company representative provided their own memos related to interview contents to the interviewer. All data collection took place in Seoul, South Korea from October 2013 to January 2014. The study protocol was reviewed and exempted by the University of South Carolina Institutional Review Board. Data analysis



www.manaraa.com

All recorded interviews were transcribed verbatim; unrecorded interviews were reconstructed into transcripts within a day using the memos taken during the interview, field notes, and memory of the interviewer. De-identified transcripts and field notes were used for the analysis. A subset of transcripts (n=5) was initially analyzed using opencoding to capture emergent codes (Strauss & Corbin, 1990) and to label the interviewees' responses. Using these emergent codes and elements of the social process model (Lasswell, 1971), a preliminary set of codes was developed. These codes were then applied to the first set of transcripts again and to the rest of transcripts. The first author coded all transcripts and worked with team members to develop the code list and identify themes. As the code list was revised or new themes were identified, all transcripts were re-coded. Themes based on the elements of the social process model – policy actors, base values, perspectives, and policy outcomes – and additional themes emergent from interviews were identified. Upon completion of coding, selected themes were put in matrices to interpret new themes and to split or to combine existing codes. The commonalities shared by interviewees from each policy actor group as well as different points expressed by them for the same theme were mapped in matrices (Miles & Huberman, 1994; Ulin et al., 2004). Peer review with one of co-authors was conducted to reach consensus on the coding, grouping of the codes into themes, and identified themes. All transcripts were managed and analyzed using NVivo 10 (QSR International).

Results

The results were organized into themes based on the social process model (Lasswell, 1971). These themes included policy actors and their base values,



perspectives, and thoughts regarding the policy and its outcomes. Compliance to the policy and its reasons were emerged during the analysis, thus these were also described.

Policy actors

Interviewees described implementation processes of the school store policy and who play as policy actors. Public officials in local (CHCs, Seoul Metropolitan government, and offices of education) and national (MFDS and MoE) governments, food industry, school staff, and school store owners were frequently mentioned as policy actors. The roles of academia and consumer associations were also mentioned.

Public officials, school store owners, and school staff mentioned what public officials do in the policy implement processes. Public officials informed policy information and monitored school food environments. CHCs hired citizens to monitor foods sold in school stores and other stores near to the schools. The monitors provided policy information to school store owners during the monitoring. In case of the city of Seoul, both the Seoul Metropolitan government and each CHC shared the budget for hiring monitors. Compared to 2012, the budget decreased in 2013, resulting in a decreased number of monitors and a lower frequency of monitoring. CHCs regularly reported the monitoring results to the Seoul Metropolitan government. Afterwards, the Seoul Metropolitan government reported it to the MFDS. The office of education in each school district also monitored school store foods. Public officials in CHCs and offices of education were specialized in food sanitation rather than school nutrition. They were in charge of school store monitoring, but did not always monitor school stores by themselves. They mentioned that direct monitoring on school stores was conducted once a year for some selected school stores.



In-school actors' roles were mentioned by public officials, school store owners, and school staff. School staff, especially those in the administrative office, was mainly in charge of the supervision of school store management. They made contracts with school store owners and observed school store foods when monitoring was required. The steering committee in each school, which consists of teachers, parents, students, and outside members, decided which foods can be sold in their school store, but the committees in some of the interviewed schools did not work for school stores. School store owners chose what to sell in school stores based on students' preference, profit, and relevant governmental and school's policies. Most teachers, students, and parents did not participate in the policy process and even were less interested in school stores.

Politicians described their role as policy making. Food industry changed some of their products to comply with the policy. The MFDS asked academia's and consumer associations' professional opinions and worked together to promote the policy.

Base values, perspectives, and thoughts regarding the policy and its outcomes

Interviewees expressed their values, perspectives, and thoughts regarding the school store policy and its outcomes. These were described below by policy actor groups. *School store owners*

School store owners stated wealth, wellbeing, relationship, and power as their important values. Wealth for school store owners was represented as revenue and income. School stores were their way "to make a living", thus they were concerned about their revenue decreasing due to the policy. Nevertheless, one of their important values was wellbeing of children. Although they did not like the negative impacts of the policy, they supported the policy from their perspectives "as parents." One of the school store owners



described her contradictory situations, struggling with decreasing revenue and considering children's health, as "double-sidedness."

"Taking into consideration children's health, I shouldn't sell soft drinks. I feel that it's difficult to do my business with many regulations. I don't want to feed bad food to children because I also have children. But everyone has double-sidedness. For children, I should not sell those (EDNP) foods to them, but for my business, it's running a deficit due to many regulations."

School store owners also valued relationships with students, school staff, and public officials. Having a good relationship with students is important for sales, which directly relate with school store owners' income, thus they tried to sell foods that students like. School store owners identified themselves as "weaker ones" and consequently they have to comply with the policy to not make any trouble with public officials and school staff. In addition, they reported having not been asked about their opinions regarding the policy because they are "weaker ones."

Although school store owners agreed that improving students' health are important, most of them (9 out of 11) claimed that the policy is unfair to them. They perceived that the policy violates freedom of business and were dissatisfied with the policy because it only blames school stores. Some of them pointed out that this "unrealistic policy" is made because public officials do not know food environments out of schools. They were unsure if the policy is effective, because of the unhealthy foods available outside of schools.

"Governments regulate only school stores although children don't become obese here [school store]; children become obese outside (of school)... The government



works without any clear plan. When they visit schools for inspection, they shouldn't look at only schools. They should go out and see what children eat first. Children just come by schools. They spend more time outside. They eat more outside. But governments blame just school stores. It seems like governments want school store owners to die."

School store owners' dissatisfaction towards being blamed and decreased revenue turned to emphasizing their weak power and distrust of governments.

Food companies

Wealth, represented by profit, was a main value of food companies. They stated that the policy is an excessive and unilateral regulation, and were concerned that the policy may result in downsizing of the food industry. They insisted that policies should focus more on nutrition education rather than regulation of specific food sales.

Food companies changed nutrient content of certain products, but did not think that the policy affects their revenue since school stores are a small portion of the market. Food company representatives described that it was not hard to comply with the policy because of ease of dodging the EDNP classification. They could change only one nutrient content or reduce portion size of their products to avoid being classified as EDNP. One interviewee expected that the policy will not change children's diet due to the loophole. He wanted to make the policy stricter for children's health as a parent.

"It's a blind spot of the policy, or a loophole. If a product passes one of the EDNP criteria among energy, sugar, protein, and saturated fat, then it's not EDNP, even though the product does not meet the other three criteria... The current policy that allows meeting only one criterion to be non-EDNP is burying



our heads in the sand. Honestly, we [food companies] can take advantage of the policy loophole... As a parent, I think if the policy really wants to do it right, if it wants to improve children's health, then it should require meeting all the four criteria... We [food companies] can avoid to being classified EDNP now, so I think the policy means nothing."

Public officials

Public officials claimed to value children's wellbeing the most. They expected that their work could help children having healthier diets. Thus, enlightenment was also an important value for public officials. The MFDS has developed educational materials, including a guidebook, a website, and a smartphone application, to increase public awareness of the policy. Public officials in local governments informed school store owners about the policy. This was accomplished through on-site monitoring or by sending documents to school administrative offices. Wealth (budget) was also reported as important to public officials since it allowed for monitoring. They were aware of the importance of profit for school stores. Some of them guessed that revenue of school stores might decrease, but others did not think the policy affects profits of school stores and food companies.

They thought that there is no difficulty in implementing the policy in general, but those in local-level governments (e.g., CHCs) expressed high burden of workload. Some public officials had doubts about the policy effectiveness in improving children's diets. They suggested more efforts to provide healthy foods to children and management of other kinds of unhealthy foods (e.g., high-energy and high-protein foods: pizza, hamburger).


School staff

School staff was not interested in school stores because they did not consider school store monitoring to be a part of their job. They described work related to school stores as "annoying" because it was "additional work" and increased their workload.

"I have not thought that school store related work is my work, although I monitor what foods the school store sells... I have not thought of the school store seriously."

All of the school staff interviewed, except for one, reported that they did not have any problem in school store management because their school stores do not sell restricted foods. They considered their relationship with school store owners important. They did not want to interrupt private business as long as school store owners comply with the schools' requests.

Politicians

Both politicians interviewed supported the policy. For the regions they represent, they had proposed ordinances to enhance policy implementation. They expressed that the policy could contribute to healthier school food environments. They were aware of the importance of profit for school store owners; but they did not think the policy affects profits of school stores and food companies. They required more efforts from other actors including school store owners, children, and food industry, to improve school food environments.

Perceived policy compliance

All interviewees thought that school stores do not sell EDNP foods; however, each actor group described different reasons for compliance. Public officials perceived



that the policy has been settled and school store owners are aware of the policy.

"School store owners' awareness has improved. When I worked for the Healthy School Store Program first (at another CHC), school store owners said 'whatever' although I explained EDNP foods. After I came here [current CHC she works], I visited school stores to explain the policy and then noticed that they already knew EDNP foods... Their awareness has been changed... They cooperated favorably." (A public official in a CHC)

School store owners and food companies expressed that there is no way to not comply with the policy. Both expressed that they comply with the policy because "it's the law." Concerns for the relationships between school store owners and school staff, as well as school store owners and public officials were reported as one of the reasons of policy compliance for school store owners. In addition, school store owners were worried that they may lose their stores if they violate the policy.

Why do school stores still sell EDNP foods?

In contrast to interviewees' perceived compliance to the policy, we found EDNP foods in all school stores during school store observations (3-11% of items sold in a school store) (Choi et al., 2016). Four reasons for the policy incompliance were found during the interviews.

Poor monitoring

Although all school stores sold EDNP foods, no one reported that they had been caught by monitoring. School store owners mentioned that monitoring is done without prior notice. Each of them reported a different frequency of monitoring ranging from three times a month to once a year. They described the fine or punishment, which they



www.manaraa.com

may get if they sell EDNP foods, as hearsay, but no one claimed to know exactly. They expected that they would get a warning if restricted foods are caught during monitoring, and they would be fined only if they sold restricted foods after the warning. Their description was accurate in case of monitoring by CHCs. The monitors hired by CHCs do not have the authority to impose fines or legal sanctions. If they find a school store sells restricted foods, they only can report the violation to public officials in CHCs. After that, public officials visit the store to see if the store really sells restricted foods. It is possible that school stores change their food products between visits from monitors and public officials. In addition, monitoring by CHCs has been decreased due to budget constraints. Although monitoring by offices of education is done through public officials, an interviewee from an office of education seem to focus more on sanitation than EDNP foods when they monitor school stores. For both of them, monitoring EDNP foods was competing with other tasks, thus was ended up having a low priority.

Though two government agencies concurrently monitor school stores, they did it separately and did not know what each other monitor. A public official in an office of education thought that CHCs do not monitor school stores, because schools are under the jurisdiction of the MoE and offices of education. Moreover, two different divisions in CHCs involved in school store monitoring – division of sanitation is in charge of sanitation, including EDNP food monitoring; the division of health management monitors foods in school stores that participate in the Healthy School Store Program.

While multiple government agencies monitor school stores, schools do not monitor their stores. School staff monitored school stores only when the offices of



www.manaraa.com

education required it; some school staff asked school store owners to record the foods they sold by themselves.

Lack of awareness of the policy

Although public officials considered that policy awareness had increased, one school store owner, who has run the store for 8 months at the time of the interview, and half of the school staff interviewed (4 out of 8) were not aware of the policy. In addition, most school store owners had difficulties discerning EDNP and non-EDNP foods. They expressed that the nutrient criteria of the policy are complex. School store owners who ran the store prior to the policy implementation remembered that all school store owners were required to attend an education session regarding the policy at the beginning of its implementation. After then, there was no education, although ownership of many school stores had changed. School store owners who run stores for a relatively short time reported that they have not attended an education session regarding the policy. They were aware of the policy because they "heard" it from people who visit their stores for monitoring, from school staff, or from the previous school store owner. Despite the MFDS's efforts in developing materials regarding the policy, most school store owners did not know where to find the information; they relied on a policy guidebook that was developed by the MFDS in 2009, which means the list of EDNP foods in the guidebook are outdated. In addition, some school store owners had a lack of nutrition knowledge. It would be hard to understand the policy that requires considering serving size, energy, saturated fat, protein, sugar, and sodium to determine EDNP foods for school store owners without sufficient nutrition knowledge.



For profit

Since wealth was an important value for school store owners, they tried to make a profit. Some school store owners complained about the skyrocketing cost of winning the bid to have the right to operate the school store. The competitive bidding system requires school stores owners of all public schools and some private schools to participate in the bidding every year. School store owners cannot continue to run the store if another person submits a higher bid. Thus, school store owners need to make more profit to submit a higher bid. Given that students prefer EDNP foods, it would be difficult for school store owners to not sell EDNP foods in consideration of profit.

"Children like EDNP foods, soft drinks, etc., that all policies and guidelines recommend eating less. They can't eat those foods at home because their parents don't allow. The only haven to eat those is the school store... Because children want EDNP foods, school stores sell those, which schools don't allow, secretly. I asked students to report [EDNP food sales], but they never report. Because those are what they want. In this situation, customers' and providers' interests are coincidence. Although it [EDNP food sales] is illegal, socially unacceptable, and bad for health, school store owners sell foods that make profits." (A high school principal)

The combination of interests between customers (students prefer EDNP foods) and providers (school store owners need profit and a good relationship with students) might hinder elimination of EDNP foods in school stores.

Lack of interest in school stores

Teachers and parents in general are not interested in school stores. Even students,



the customer of school stores, and school staff who are in charge of supervision do not pay attention to school stores. Most schools did not have their own school food policy; in cases in which schools have their own policy, they are almost the same as the government policy.

Public officials in local governments are supposed to monitor school stores, but they are less interested in school stores than in other work. They "don't know much" about school stores because they are too busy to visit school stores. Work related to school stores is a small part of their work.

Discussion

Four years after the policy restricting EDNP food sales was implemented, the school food environments had not changed much in South Korea (Choi et al., 2016; S. C. Kim et al., 2010). This study aimed to find the reasons for why school stores still sell the restricted EDNP foods. Public officials made efforts to implement the policy, but there was a lack of interest in school stores among policy actors. All actors understood the policy intent, but not every actor was satisfied with the policy and its outcomes. All the interviewees, including school store owners, stated that school stores do not sell EDNP foods, but this was contradicted by the school store observations. From the interviews, we found that school stores sell EDNP foods because of poor monitoring, lack of awareness or understanding of the policy and EDNP foods, profit-seeking, and lack of interest in school stores.

The school store policy is implemented by public officials, school staff, and school store owners. While public officials' and school staff's roles in policy implementation were conveyed well by all of the interviewees, school store owners'



participation was not clearly expressed neither by public officials nor by school store owners. This may be linked to the store owners' self-identity as "weaker ones." School store owners are front-line implementers, and they choose what to sell to students. They felt, however, that they were being forced to follow the policy rather than working with the government for students' health. Although public officials and politicians interviewed said that everyone could express their opinion during a certain period of the policy development, school store owners may not have known about the opportunity. In addition, if some school store owners were aware of the opportunity, their voices might not have had enough power to influence the policy. Actors who are affected by policies are often excluded in the decision-making process, which turns into a misunderstanding of the policy intent, distrust of the government, or unsupportive responses to the policy (Roberts et al., 2009). Studies suggest that policy makers need to include front-line implementers in the decision-making process and communicate with all actors to inform them about the policy (McDonnell, Probart, Weirich, Hartman, & Bailey-Davis, 2006; Roberts et al., 2009; Vine & Elliott, 2014). This would increase policy awareness among actors, improve implementation processes, and promise long-term success of the policy (McDonnell et al., 2006; Roberts et al., 2009; Vine & Elliott, 2014).

Understanding actors' values and expectations can guide development of strategies to improve policy implementation (Pan Canadian Joint Consortium for School Health, 2010). School store owners expressed double-sidedness regarding the policy. Although they were happy about the government's efforts to improve children's health, they were unhappy about their decreased income. Yet, public officials and politicians did not expect the decreased income for school store owners. These unmatched expectations



among actors may be a reason for the double-sidedness that was felt towards the policy. When participants are aware of discrepancies between their expectations and actual situations as a result of a policy, problems that may hinder policy implementation arise (Lasswell, 1971). As school store owners perceive that their two values (wealth and wellbeing) cannot be compatible under the policy, they may find their own way, such as selling EDNP foods, to try to satisfy both values. One of the possible strategies to reduce EDNP food sales could be filling the gaps between school store owners' two values, wealth and wellbeing, and between different expectations among actors (Green & Aarons, 2011). Actors' values hint towards incentives to which they are likely to respond. An important value of school store owners is wealth, which public officials and politicians did not consider in policy design and implementation. If the policy guarantees school store owners' wealth, they may be more willing to comply with the policy. Securing the school store ownership could be an incentive for policy compliance. As school store owners pointed out, the high-bidding price is one of the difficulties in running school stores; therefore, schools may provide advantages during the bidding process to the school store owners that do not sell EDNP foods.

We found a discrepancy between perceived policy compliance by policy actors, even school store owners, and the actual compliance. Interviewees believed that the policy is being implemented well, which is not true. One of the reasons for the discrepancy in perceived and actual policy compliance by school store owners and school staff may be a lack of policy awareness. Additionally, those who know the policy also had difficulties in discerning between EDNP and non-EDNP foods. Interviewees pointed out that the current EDNP criteria are too complex to understand. In consideration of the



frequent changes in school store ownership, regular education or training for school store owners and school staff, especially for those new to the job, is needed to ensure the policy implementation (Durlak & DuPre, 2008). The MFDS has developed various materials to improve awareness of the policy; but the materials do not reach to front-line implementers. Thus, strategies to promote the existing materials are needed.

The monitoring system adds another layer of complexity, creating challenges for school store owners and public officials. Both nationally and locally, the responsibilities regarding school health in South Korea overlap among multiple government agencies. For example, the MoE is responsible for all school-related issues including school store management. The MFDS works in food and nutrition issues, thus school foods are also their interest. From the beginning of the policy, the risk for duplicated policy enforcement has been inevitable. Currently, school store monitoring is split among multiple governmental agencies at the national (MFDS and MoE), municipal (e.g., Seoul Metropolitan government), and local (CHCs and offices of education) levels. This duplicated policy enforcement neither contributed effective monitoring nor reduced EDNP foods in school stores. Rather these uncoordinated efforts added burdens to public officials who felt that they had a heavy workload, as well as to school store owners who felt that their businesses were controlled by the government. Poor coordination among multi-agencies often leads to poor implementation (Pelletier et al., 2012). The current policy does not specify how to coordinate monitoring among multi-agencies. A formal mechanism will be needed to enhance coordination of multi-agencies' monitoring. It will enable government agencies to manage school store foods in a more systematic way and to set clear responsibilities of each agency for adequate allocation of human and financial



resources (Holmes, 1992). This will also ensure regular monitoring which was impossible due to a limited budget and the high workload of public officials. Additionally, the current monitoring conducted by monitors without the authority to impose a fine needs to be reconsidered. Because of the absence of regulatory authority, on-site monitoring has not been effective to capture policy violations. People in charge of monitoring should be given appropriate authority that allows their monitoring to be effective.

Some policy actors have not paid attention to school stores. Public officials in local governments and school staff who are supposed to be in charge of school store monitoring expressed little interest in monitoring school stores. Further, most parents have no interest in school stores. A lack of support from key stakeholders is a barrier for policy implementation (Agron et al., 2010; Clarke et al., 2013). Previous studies consistently emphasized the need for marketing or education strategies to inform people of the potential benefits of the policy (Agron et al., 2010; Thrasher et al., 2011; Vine & Elliott, 2014). It could garner actors' supports for the policy implementation and enhance the implementation processes (Agron et al., 2010; Vine & Elliott, 2014). Especially, if students, the customers of school stores, are aware of the policy and its importance and buy non-EDNP foods, the school store owners' concern regarding profit loss might be reduced.

The school staff's lack of interest in school stores resulted in no monitoring conducted by schools. Previous research reported schools with a formal group or designated personnel who are responsible for school nutrition and high support from staff and administrators are more likely to implement the policy well (Lucarelli et al., 2014; Thacher, 2005). Therefore, school administrators, teachers, and staff need to pay attention



to their school food environment to improve students' health. Each school also should extend the job description of an individual to include the school store management as a part of their duties.

This study has several limitations. First, for some actor groups, the number of interviewees was too small to show their perspectives as representing their groups. We were only able to use their interview content to support or verify interviews with other groups. Second, we might have missed perspectives of some potential key actors. We were unable to conduct interviews with public officials in the MoE, parents, and teachers. Lastly, the interviewees might have responded in a socially desirable manner, for example, insisting they comply with the policy.

This study also has strengths. Our study conducted interviews with diverse policy actors. Previous research on implementation of school wellness policy mostly focused on the perspectives of a limited range of actor groups, e.g., in-school actors, actors in school districts. Thus, our study could reflect a greater variety of perspectives of policy actors. While the number of interviewees is relatively small, our findings are transferable to school nutrition policy issues in other countries. Previous studies conducted in the United States, Canada, and Mexico also showed different perspectives on school nutrition policy actors and dissonant perspectives that are related to incomplete policy implementation (French, Story, Fulkerson, & Gerlach, 2003; McDonnell et al., 2006; Monterrosa et al., 2015; Vine & Elliott, 2014).

Conclusion

Policy enactment does not guarantee successful implementation (Kingdon, 2002;



McDonnell et al., 2006). This study supports the need for attention on various actors' perspectives when implementing school nutrition policies (Monterrosa et al., 2015). Policy decisions are often made by those who have higher power with minimal input from powerless actors who are affected by the policy. For future development and implementation of school nutrition policies, all policy actors should be given opportunities to discuss their thoughts on the policy. These discussions would increase actors' awareness of the policy and generate shared understanding of the policy among actors.

References

- Agron, P., Berends, V., Ellis, K., & Gonzalez, M. (2010). School wellness policies: perceptions, barriers, and needs among school leaders and wellness advocates. *Journal of School Health*, 80(11), 527-535.
- Birkland, T. A. (2005). An introduction to the policy process: Theories, concepts, and models of public policy making (2nd ed.). Armonk, NY: ME Sharpe Inc.
- Choi, S. K., Frongillo, E. A., Blake, C. E., & Thrasher, J. F. (2016). Comparison of food availability in school stores in Seoul, South Korea before and after implementation of two national food- and nutrient-based policies. *Faseb Journal*, 30(1), 129.127.
- Clark, S. G. (2002). *The policy process: a practical guide for natural resources professionals*. New Heaven: Yale University Press.
- Clarke, J., Fletcher, B., Lancashire, E., Pallan, M., & Adab, P. (2013). The views of stakeholders on the role of the primary school in preventing childhood obesity: a qualitative systematic review. *Obesity Reviews*, 14(12), 975-988. doi:10.1111/obr.12058
- Durlak, J. A., & DuPre, E. P. (2008). Implementation matters: A review of research on the influence of implementation on program outcomes and the factors affecting implementation. *American Journal of Community Psychology*, *41*(3-4), 327-350. French, S. A., Story, M., Fulkerson, J. A., & Gerlach, A. F. (2003). Food environment in secondary schools: a la carte, vending machines, and food policies and practices. *American Journal of Public Health*, *93*(7), 1161-1168.



- Green, A. E., & Aarons, G. A. (2011). A comparison of policy and direct practice stakeholder perceptions of factors affecting evidence-based practice implementation using concept mapping. *Implementation Science*, *6*, 104.
- Hill, M. J., & Hupe, P. L. (2002). *Implementing public policy: governance in theory and practice* (1st ed.). London: SAGE Publications Ltd.
- Holmes, M. (1992). Public sector management reform: convergence or divergence? *Governance*, *5*(4), 472-483.
- Jaime, P. C., & Lock, K. (2009). Do school based food and nutrition policies improve diet and reduce obesity? *Preventive Medicine*, *48*, 45-53.
- Kim, K., Park, S. M., & Oh, K. W. (2013). The impact of nutritional policy on socioeconomic disparity in the unhealthy food intake among Korean adolescents. *Appetite*, 71, 388-395. doi:10.1016/j.appet.2013.09.010
- Kim, S. C., Lee, D. H., Choi, Y. J., Seo, I. W., Yoo, S., Jang, K. A., . . . Kim., H. (2010). A study on effectiveness improvement for the children's dietary life safety management (National Institute of Food and Drug Safety Evaluation publication No. 10061Eorini232). Seoul: National Institute of Food and Drug Safety Evaluation.
- Kingdon, J. W. (2002). Agendas, alternatives, and public policies (longman classics edition) (2 ed.). New York: Longman Publishing Group.
- Kocken, P. L., van Kesteren, N., Buijs, G., Snel, J., & Dusseldorp, E. (2015). Students' beliefs and behaviour regarding low-calorie beverages, sweets or snacks: are they affected by lessons on healthy food and by changes to school vending machines? *Public Health Nutrition*, 18(9), 1545-1553.
- Lasswell, H. D. (1971). A pre-view of policy sciences. New York: American Elsevier.
- Lucarelli, J. F., Alaimo, K., Mang, E., Martin, C., Miles, R., Bailey, D., . . . Liu, H. (2014). Facilitators to promoting health in schools: is school health climate the key? *Journal of School Health*, 84(2), 133-140.
- McDonnell, E., Probart, C., Weirich, J. E., Hartman, T., & Bailey-Davis, L. (2006). School competitive food policies: perceptions of Pennsylvania public high school foodservice directors and principals. *Journal of the American Dietetic Association*, 106(2), 271-276.
- Miles, M. B., & Huberman, A. M. (1994). *Qualitative data analysis: An expanded sourcebook.* Beverly Hilles, CA: Sage Publications.
- Monterrosa, E. C., Campirano, F., Mayo, L. T., Frongillo, E. A., Cordero, S. H., Kaufer-Horwitz, M., & Rivera, J. A. (2015). Stakeholder perspectives on national policy



www.manaraa.com

for regulating the school food environment in Mexico. *Health Policy and Planning*, *30*(1), 28-38. doi:10.1093/heapol/czt094

- Pan Canadian Joint Consortium for School Health. (2010). Stakeholder engagement for improved school policy: development and implementation. *Canadian Journal of Public Health*, 101(8), S20-S23.
- Pelletier, D. L., Frongillo, E. A., Gervais, S., Hoey, L., Menon, P., Ngo, T., . . . Ahmed, T. (2012). Nutrition agenda setting, policy formulation and implementation: lessons from the Mainstreaming Nutrition Initiative. *Health Policy and Planning*, 27, 19-31.
- Roberts, S. M., Pobocik, R. S., Deek, R., Besgrove, A., & Prostine, B. A. (2009). A qualitative study of junior high school principals' and school food service directors' experiences with the Texas school nutrition policy. *Journal of Nutrition Education and Behavior*, 41(4), 293-299. doi:10.1016/j.jneb.2008.05.010
- Strauss, A., & Corbin, J. (1990). *Basics of qualitative research: Grounded theory* procedures and techniques. Newbury Park, CA: Sage Publications, Inc.
- Tansey, O. (2007). Process tracing and elite interviewing: a case for non-probability sampling. *PS: Political Science & Politics, 40*(4), 765-772.
- Thacher, A. K. (2005). From concept to practice: using the School Health Index to create healthy school environments in Rhode Island elementary schools. *Preventing Chronic Disease, 2*, A09.
- Thrasher, J. F., Huang, L., Pérez-Hernández, R., Nuederdeooe, J., Arillo-Santillán, E., & Alday, J. (2011). Evaluation of a social marketing campaign to support Mexico City's comprehensive smoke-free law. *American Journal of Public Health*, 101(2), 328-335.
- Ulin, P. R., Robinson, E. T., & Tolley, E. E. (2004). *Qualitative methods in public health: a field guide for applied research*. San Francisco, CA: Jossey-Bass.
- Vine, M. M., & Elliott, S. J. (2014). Examining local-level factors shaping school nutrition policy implementation in Ontario, Canada. *Public Health Nutrition*, 17(6), 1290-1298. doi:10.1017/S136898001300251



CHAPTER 5

Summary, Implications, and Recommendations

The two Korean school nutrition policies have been implemented to improve school store food environments and diets of Korean adolescents. This study showed that the two policies have had a modest impact on the availability of foods offered in school stores and on diets of adolescents. Food access away from home or school, however, still has considerable impact on diets of adolescents. The possible reasons for the modest impact of policies are poor monitoring of school store foods, a lack of awareness or understanding of the policies, profit-seeking, and a lack of interest in school stores.

In the first manuscript, we examined trends in Korean adolescents' energy intake from selected school store foods that are targets of the policies during 1998-2012. Korean adolescents had more energy intake from school store foods at home than in schools or in school stores. This was consistent with our interviewees' perception described in the third manuscript that restriction on school store foods might have a little impact on students' diets because of unhealthy foods available outside of schools. Given that students' food intake in school stores is relatively small, policies targeting a single eating place (school store) are insufficient to alter students' diets. This result highlights the need for a comprehensive policy that targeting multiple eating places, including school, at-home, and away from home to improve adolescents' overall diets. Although at-home intake contributes the highest proportion of both healthy and unhealthy foods among



adolescents (Briefel, Wilson, et al., 2009) and home food environments affect food choice (Holsten, Deatrick, Kumanyika, Pinto-Martin, & Compher, 2012), there is no policy to improve the at-home diet in South Korea. Rather, policies and efforts to improve adolescents' diets have mostly focused on the in-school consumption (S. G. Bae et al., 2012). In addition, Korean adolescents consumed more unhealthy foods, such as soft drinks away from home or school, where they have various food choices under less supervision from their parents and teachers. Future policies need to be more holistic by targeting multiple eating places and to ask both adolescents and parents for involvement in the efforts to have a healthier diet.

Our study results in the second manuscript suggest that the 2007 food-based policy aimed at decreasing availability of specific foods in school stores led to decreases in availability and consumption of the target foods in school stores, as seen in the decreased availability and intake of instant noodles and soft drinks in school stores while the intake was increased or unchanged in other eating places. The 2009 nutrient-based policy seemed to have less influence on improving school store food environments and adolescents' intake of target foods of the policy. EDNP foods that the 2009 policy targeted to remove from school stores were available in all school stores observed in 2013; in addition, there was no distinct change in adolescents' intake of school store foods after 2009.

One of the possible reasons for less impact of the 2009 policy is the nature of the nutrient-based policy. EDNP foods are determined based on serving size and amount of energy, saturated fat, protein, sugar, and sodium in the foods. In the interviews with policy actors in our third manuscript, food companies expressed that they could easily



make their previously EDNP products to non-EDNP by changing one nutrient content to avoid being EDNP. Previous studies reported that nutrient-based policies may reduce less healthy foods, but still allow unhealthy foods that technically meet the nutrient criteria of the policy to be available in schools (Samuels et al., 2009; Woodward-Lopez et al., 2010). Thus, the 2009 nutrient-based policy is not enough to remove unhealthy foods from school stores. The complex nutrient criteria also made it hard for school store owners to comply with the 2009 policy. The determination of EDNP foods requires consideration of various combinations of nutrient content that differ by food types (snack versus meal substitutes) while the 2007 food-based policy has a simpler and more specific requirement. Most school store owners did not understand the nutrient criteria to determine EDNP foods, and they had difficulties to distinguish EDNP foods. Food-based policies are easier to interpret than nutrient-based policies (Samuels et al., 2009; Woodward-Lopez et al., 2010). Because of the contract system for school store ownership in South Korea, school store owners are changed frequently (minimum of one semester). Government agencies will be able to help school store owners to overcome the complexity of the nutrient-based policy through monitoring policy implementation processes and school store foods. The monitoring is not only for detecting violations of the policies, but also for providing support to school stores that are struggling with policy implementation (Mâsse et al., 2013). Regular education, training, and technical support, especially for those new to the job, is needed to ensure the implementation of the policy (Davee et al., 2005; Durlak & DuPre, 2008).

Although soft drink availability in school stores has decreased after implementation of the 2007, half of the school stores observed still sold soft drinks in



2013. In the third manuscript, wealth was reported as one of the values that school store owners considered important. Given the school store owners' perception that students prefer soft drinks and EDNP foods, it would be difficult for school store owners to not sell the restricted foods for their profit (wealth), even though they valued students' health (wellbeing). One of the possible strategies to reduce the restricted food sales could be to fill the gaps between school store owners' two values, wealth and wellbeing (Green & Aarons, 2011). The current policies are only restrictive and may cause a decrease in school store owners' profit, but there is no benefit for them in complying with the policies. As school store owners pointed out, the high-bidding price is one of the difficulties in running school stores; therefore, providing advantages during the bidding process to the school store owners who do not sell restricted foods may encourage them to not to sell the restricted foods. Replacement of unhealthy foods to healthier foods does not necessarily generate revenue loss (French, Story, Fulkerson, & Hannan, 2004). Public officials and food companies should work with school store owners to find healthier food items that meet policies' criteria, but will not affect revenue of school stores.

Lack of interest and accountability in school stores and poor monitoring of school store foods also allowed the restricted foods being sold in school stores. Work related to school stores had a low priority for school staff and public officials; some of them thought that it is not their responsibility. Several government agencies monitored school stores concurrently, but the duplicated monitoring was not effective in capturing restricted food sales. Rather, these uncoordinated efforts added burdens to public officials as well as to school store owners, who felt that their businesses were controlled by the government. Poor coordination among multi-agencies can lead to poor implementation



(Pelletier et al., 2012). The policies will need to have a formal mechanism to enhance coordination of multi-agencies' tasks regarding policy implementation and school store monitoring. It will enable government agencies to manage school store foods in a more systematic way and to set clear responsibilities of each agency and policy actors for adequate allocation of human and financial resources (Holmes, 1992). Also, having a council at the school and/or school district level may increase awareness, interests, and responsibilities of in-school and school district actors' and family of students' on school store foods through engagement of a variety of actors in policy implementation processes (Kubik, Farbakhsh, & Lytle, 2011; Kubik, Lytle, & Farbakhsh, 2011). Designated responsibilities through the council will enable school store foods be a higher priority among actors' work. This could lead to more frequent and well-managed monitoring of school store foods and improvement on the policy implementation.

This study showed room for improvement in students' diets and school food environments. Since in-school consumption of school store foods is lower than consumption in other places, the two school store policies might not be enough to improve students' overall diets. Nevertheless, efforts to improve school food environments are still important. Schools can help students to eat healthier foods that they would not eat away from home or school. The higher energy intake from milk in school or school store compared to away from home or school shown in the first manuscript could be an example of the schools' positive influence on students' food choices. Previous studies also showed that school nutrition policies requiring healthy foods in schools are more effective to improve students' diets for those from low-income households, thus contributing to reduced disparities in healthy food consumption (Taber,



Chriqui, & Chaloupka, 2013; Vericker, 2013). Thus, in addition to the policies restricting unhealthy foods in schools, efforts to increase provision of healthy foods to students would help to improve students' diets and to reduce disparities in dietary intake. Better implementation of the policies and appropriate monitoring will also contribute to improve students' diets and food environments in school stores.

The changes reported in this study in the intake of selected foods that frequently offered in school stores of Korean adolescents are different than reported in previous studies. Previous studies reported that Korean adolescents' consumption of unhealthy foods, including soft drinks, instant noodles, confectionary, and fast foods, had decreased after implementation of the 2007 and 2009 policies (S. G. Bae et al., 2012; K. Kim et al., 2013). These studies, however, focused only on overall frequency of food intake regardless of the amount of consumption and the eating place. It is hard to know what eating episodes contribute to the decrease of the overall intake of unhealthy foods without consideration of eating places. Our study adds a deeper layer to our understanding of the impact of nutrition policies on changes in food intake by identifying where shifts in food intake occur after policy implementation.

This study has several limitations. First, we considered two policies in South Korea related to school store food environments. Other nutrition policies that have been implemented during the study period (e.g., nutrition labeling and reductions in sodium in processed foods) may affect adolescents' diets. In addition, the Special Act, the umbrella policy of the 2009 policy, originally intended to improve out-of-school food environments by prohibiting EDNP food sales in stores located near schools and restricting advertisements of EDNP foods on television during 5-7 pm beginning



September 2010. The non-sale of EDNP foods in stores outside of schools were not mandatory, however, and only a few stores voluntarily participated in stopping the sale of EDNP foods (less than 4.2% of stores as of 2012) (Korea Ministry of Food and Drug Safety, 2013). Since there was not a notable change in food intake after restriction of EDNP food advertisements on television in our analysis, we considered that other components of the 2009 policy did not affect adolescents' diets during the study period. Second, we were not able to examine changes in intake of EDNP foods and its availability in school stores. Since the KNHANES food database does not provide sugar content of foods, it was impossible to distinguish EDNP foods from non-EDNP foods. When we observed school store foods in 2006, the nutrition labeling policy in South Korea did not require labeling of saturated fat and sugar content of food products. Thus, we could not determine EDNP foods using the observation data obtained in 2006. Lastly, in our qualitative study, we had a small number of interviewees from some actor groups and were unable to conduct interviews with some of potential key actors because of their unavailability and/or refusal.

Schools could be venues to provide healthy foods to their students to promote healthy diets (Briefel, Crepinsek, et al., 2009). Many policies to improve school food environments have been enacted worldwide, but enactment of policies do not always guarantee its intended outcomes (Kingdon, 2002). The present study showed changes in Korean adolescents' consumption of selected school store foods and foods offered in school stores after the enactment of two policies and implementation processes of a nutrient-based policy. This study emphasizes the need of comprehensive policies aiming at improvement of food environments in diverse eating places. Further research is needed



to examine closely adolescents' main food sources in each eating place to find what kind of action is needed in which place. Understanding various policy actors' perspectives is important in policy implementation processes (Monterrosa et al., 2015). It can explain the underlying reasons for policy successes or failures and can provide opportunities to improve the policy processes. In the future policy development process, various policy actors should be given opportunities to participate in the policy process.



REFERENCES

- Agron, P., Berends, V., Ellis, K., & Gonzalez, M. (2010). School wellness policies: perceptions, barriers, and needs among school leaders and wellness advocates. *Journal of School Health*, 80(11), 527-535.
- Ahn, K. (2009, November 17). School stores in high schools in Gwangju still sell soft drink and instant noodles. *The Hankyoreh*. Retrieved from http://www.hani.co.kr/arti/society/area/388289.html
- Bae, S. G., Kim, J. Y., Kim, K. Y., Park, S. W., Bae, J., & Lee, W. K. (2012). Changes in dietary behavior among adolescents and their association with government nutrition policies in Korea, 2005-2009. *Journal of Preventive Medicine & Public Health*, 45(1), 47-59. doi:10.3961/jpmph.2012.45.1.47
- Bae, Y.-J., & Yeon, J.-Y. (2013). Evaluation of nutrient intake and diet quality according to beverage consumption status of elementary school, middle school, and high school students: from the Korean National Health and Nutrition Examination Surveys, 2007-2008. *Korean Journal of Nutrition*, 46(1), 34-49.
- Birkland, T. A. (2005). An introduction to the policy process: Theories, concepts, and models of public policy making (2nd ed.). Armonk, NY: ME Sharpe Inc.
- Briefel, R. R., Crepinsek, M. K., Cabili, C., Wilson, A., & Gleason, P. M. (2009). School food environments and practices affect dietary behaviors of US public school children. *Journal of the American Dietetic Association*, 109(2), S91-S107. doi:10.1016/j.jada.2008.10.059
- Briefel, R. R., Wilson, A., & Gleason, P. M. (2009). Consumption of low-nutrient, energy-dense foods and beverages at school, home, and other locations among school lunch participants and nonparticipants. *Journal of the American Dietetic Association*, 109(2), S79-S90. doi:10.1016/j.jada.2008.10.064
- Buse, K. (2008). Addressing the theoretical, practical and ethical challenges inherent in prospective health policy analysis. *Health Policy and Planning*, 23(5), 351-360. doi:10.1093/heapol/czn026
- Chang, N. S., Choi, Y. S., Oh, S. Y., Yoon, J. H., Joung, H., & Choi, S. K. (2007). *Guidance daily amounts for sugar and sodium in foods for children*. Seoul, Korea: Korea Food and Drug Administration.



- Choi, S. K., Choi, H. J., Chang, N. S., Cho, S. H., Choi, Y. S., Park, H. K., & Joung, H. (2008). Snacking behaviors of middle and high school students in Seoul. *Korean Journal of Community Nutrition*, 13(2), 199-206.
- Choi, S. K., Frongillo, E. A., Blake, C. E., & Thrasher, J. F. (2016). Comparison of food availability in school stores in Seoul, South Korea before and after implementation of two national food- and nutrient-based policies. *Faseb Journal*, 30(1), 129.127.
- Choi, S. K., Frongillo, E. A., Blake, C. E., Thrasher, J. F., & Tompkins, M. E. (2015). Why do school stores in South Korea still sell restricted foods four years after the school store policy? *Faseb Journal*, 29(1), 741.749.
- Chriqui, J. F., Pickel, M., & Story, M. (2014). Influence of school competitive food and beverage policies on obesity, consumption, and availability: a systematic review. *JAMA Pediatrics*, 168(3), 279-286. doi:10.1001/jamapediatrics.2013.4457
- Chriqui, J. F., Turner, L., Taber, D. R., & Chaloupka, F. J. (2013). Association between district and state policies and US public elementary school competitive food and beverage environments. *JAMA Pediatrics*, 167(8), 714-722.
- Clark, S. G. (2002). *The policy process: a practical guide for natural resources professionals*. New Heaven: Yale University Press.
- Clarke, J., Fletcher, B., Lancashire, E., Pallan, M., & Adab, P. (2013). The views of stakeholders on the role of the primary school in preventing childhood obesity: a qualitative systematic review. *Obesity Reviews*, 14(12), 975-988. doi:10.1111/obr.12058
- Cleveland, W. S. (1979). Robust locally weighted regression and smoothing scatterplots. *Journal of the American Statistical Association*, 74(368), 829-836.
- Cradock, A. L., McHugh, A., Mont-Ferguson, H., Grant, L., Barrett, J. L., Wang, Y. C., & Gortmaker, S. L. (2011). Effect of school district policy change on consumption of sugar-sweetened beverages among high school students, Boston, Massachusetts, 2004-2006. *Preventing Chronic Disease*, 8(4), A74.
- Cram, L. (1997). *Policy-making in the European Union: conceptual lenses and the integration process*. London: Routledge.
- Cullen, K. W., Watson, K., & Zakeri, I. (2008). Improvements in middle school student dietary intake after implementation of the Texas public school nutrition policy. *American Journal of Public Health*, 98(1), 111-117.
- Cullen, K. W., & Zakeri, I. (2004). Fruits, vegetables, milk, and sweetened beverages consumption and access to à la carte/snack bar meals at school. *American Journal of Public Health*, *94*(3), 463-467.



- Davee, A., Blum, J., Devore, R., Beaudoin, C., Kaley, L., Leiter, J., & Wigand, D. (2005). The vending and a la carte policy intervention in Maine public high schools. *Preventing Chronic Disease*, 2, A14.
- Dietz, W. H. (2013). New strategies to improve food marketing to children. *Health Affairs*, *32*(9), 1652-1658.
- Do, H.-S. (2012). Breakfast skipping, intake of school meals and snacks among junior high school students in Daegu region depending on the presence/absence of inschool canteen. (Master's thesis), Retreived from http://www.riss.kr/link?id=T13090928
- Dodson, E. A., Fleming, C., Boehmer, T. K., Haire-Joshu, D., Luke, D. A., & Brownson, R. C. (2009). Preventing childhood obesity through state policy: qualitative assessment of enablers and barriers. *Journal of Public Health Policy*, S161-S176.
- Durlak, J. A., & DuPre, E. P. (2008). Implementation matters: A review of research on the influence of implementation on program outcomes and the factors affecting implementation. *American Journal of Community Psychology*, 41(3-4), 327-350. doi:10.1007/s10464-008-9165-0
- Fernandes, M. M. (2013). A national evaluation of the impact of state policies on competitive foods in schools. *Journal of School Health*, 83(4), 249-255. doi:10.1111/josh.12024
- Finkelstein, D. M., Hill, E. L., & Whitaker, R. C. (2008). School food environments and policies in US public schools. *Pediatrics*, 122(1), e251-e259.
- Finkelstein, E., French, S., Variyam, J. N., & Haines, P. S. (2004). Pros and cons of proposed interventions to promote healthy eating. *American Journal of Preventive Medicine*, 27(3), 163-171.
- Fox, S., Meinen, A., Pesik, M., Landis, M., & Remington, P. L. (2005). Competitive food initiatives in schools and overweight in children: a review of the evidence. *Wisconsin Medical Journal*, 104(5), 38-43.
- French, S. A., Story, M., Fulkerson, J. A., & Gerlach, A. F. (2003). Food environment in secondary schools: a la carte, vending machines, and food policies and practices. *American Journal of Public Health*, 93(7), 1161-1168.
- French, S. A., Story, M., Fulkerson, J. A., & Hannan, P. (2004). An environmental intervention to promote lower-fat food choices in secondary schools: outcomes of the TACOS study. *American Journal of Public Health*, 94(9), 1507-1512.
- Gittelsohn, J. (2012). Interventions in small food stores to change the food environment, improve diet, and reduce risk of chronic disease. *Preventing Chronic Disease*, 9, E59.



- Green, A. E., & Aarons, G. A. (2011). A comparison of policy and direct practice stakeholder perceptions of factors affecting evidence-based practice implementation using concept mapping. *Implementation Science*, *6*, 104.
- Greves, H. M., & Rivara, F. P. (2006). Report card on school snack food policies among the United States' largest school districts in 2004–2005: Room for improvement. *International Journal of Behavioral Nutrition and Physical Activity*, 3, 1. doi:10.1186/1479-5868-3-1
- Hanson, N. I., Neumark-Sztainer, D., Eisenberg, M. E., Story, M., & Wall, M. (2005). Associations between parental report of the home food environment and adolescent intakes of fruits, vegetables and dairy foods. *Public Health Nutrition*, 8(1), 77-85.
- Haroun, D., Harper, C., Wood, L., & Nelson, M. (2011). The impact of the food-based and nutrient-based standards on lunchtime food and drink provision and consumption in primary schools in England. *Public Health Nutrition*, 14(2), 209-218.
- Hebert, J. R., Hurley, T. G., Steck, S. E., Miller, D. R., Tabung, F. K., Peterson, K. E., . . . Frongillo, E. A. (2014). Considering the Value of Dietary Assessment Data in Informing Nutrition-Related Health Policy. *Advanced Nutrition*, 5(4), 447-455. doi:10.3945/an.114.006189
- Henriksen, L. (2012). Comprehensive tobacco marketing restrictions: promotion, packaging, price and place. *Tobacco Control, 21*(2), 147-153.
- Hill, M. J., & Hupe, P. L. (2002). *Implementing public policy: governance in theory and practice* (1st ed.). London: SAGE Publications Ltd.
- Holmes, M. (1992). Public sector management reform: convergence or divergence? *Governance*, *5*(4), 472-483.
- Holsten, J. E., Deatrick, J. A., Kumanyika, S., Pinto-Martin, J., & Compher, C. W. (2012). Children's food choice process in the home environment. A qualitative descriptive study. *Appetite*, 58(1), 64-73.
- Jaime, P. C., & Lock, K. (2009). Do school based food and nutrition policies improve diet and reduce obesity? *Preventive Medicine, 48*, 45-53.
- Jones, S. J., Gonzalez, W., & Frongillo, E. A. (2009). Policies that restrict sweetened beverage availability may reduce consumption in elementary-school children. *Public Health Nutrition*, 13(4), 589-595. doi:10.1017/S1368980009991819
- Kant, A. K., & Graubard, B. I. (2003). Predictors of reported consumption of lownutrient-density foods in a 24-h recall by 8–16 year old US children and adolescents. *Appetite*, 41(2), 175-180.



- Kim, D. W., Park, K., Kim, J., Oh, K., Joung, H., Lee, J. E., & Paik, H. Y. (2013). Sources of variation in nutrient intake and the number of days to assess usual intake among men and women in the Seoul metropolitan area, Korea. *The British Journal of Nutrition*, *110*(10), 2098-2107.
- Kim, H. (2014). Changes in the nutrient contents of children's favorite foods after policy implementation on Energy-Dense and Nutrient-Poor foods in Korea (Masters's thesis), Retreived from http://www.riss.kr/link?id=T13438993.
- Kim, H. R. (2008). A strategy for child and adolescent health promotion: nutrition, overweight and obesity. *Health and Welfare Policy Forum*, 141(4), 20-31.
- Kim, K., Park, S. M., & Oh, K. W. (2013). The impact of nutritional policy on socioeconomic disparity in the unhealthy food intake among Korean adolescents. *Appetite*, 71, 388-395. doi:10.1016/j.appet.2013.09.010
- Kim, M. (2013, May 27). Do my children eat this hamburger? Weekly Chosun. Retrieved from http://weekly.chosun.com/client/news/viw.asp?nNewsNumb=002258100007&ctc d=C04
- Kim, S. (2010). The evaluation of food labeling effect using matching method. *Korean Journal of Agricultural Economics*, 51(3), 47-71.
- Kim, S., Lee, Y., Yoon, J., Chung, S.-J., Lee, S.-K., & Kim, H. (2013). Restriction of television food advertising in South Korea: impact on advertising of food companies. *Health Promotion International*, 28(1), 17-25. doi:10.1093/heapro/das023
- Kim, S. C., Lee, D. H., Choi, Y. J., Seo, I. W., Yoo, S., Jang, K. A., . . . Kim., H. (2010). A study on effectiveness improvement for the children's dietary life safety management (National Institute of Food and Drug Safety Evaluation publication No. 10061Eorini232). Seoul: National Institute of Food and Drug Safety Evaluation.
- Kim, Y.-J. (2012). Usage of school tuck shop and analysis of some variable related to purchase attitude of healthy snack for induction of healthy food in high school student in Daejeon Metropolitan City. (Master's thesis), Retrieved from http://www.riss.kr/link?id=T12927935
- Kim, Y. J. (2010). Comparison on snack purchasing behaviors and snack intake between male and female middle school students in Seoul (Master's thesis), Retrived from: http://www.riss.kr/link?id=T12194989
- Kingdon, J. W. (2002). Agendas, alternatives, and public policies (longman classics edition) (2 ed.). New York: Longman Publishing Group.



- Kocken, P. L., van Kesteren, N., Buijs, G., Snel, J., & Dusseldorp, E. (2015). Students' beliefs and behaviour regarding low-calorie beverages, sweets or snacks: are they affected by lessons on healthy food and by changes to school vending machines? *Public Health Nutrition*, 18(9), 1545-1553.
- Korea Food and Drug Administration. (2012, January 11). Check your smartphone when you worry about obesity -Development and dissemination of a smartphone application to find energy-dense nutrient-poor foods. Retrieved from http://www.mfds.go.kr/index.do?mid=664&pageNo=106&seq=16829&cmd=v
- Korea Food Industry Association, & Chung-Ang University. (2011). Processed food market report: Soft drinks (Korea Agro-Fisheries & Food Trade Corp. Publication). Naju: Korea Agro-Fisheries & Food Trade Corp.
- Korea Ministry of Agrigulture, Food and Rural Affairs, & Korea Food Industry Association. (2013a). *Processed food market report: Chocolate*. (Korea Agro-Fisheries & Food Trade Corp. Publication). Naju: Korea Agro-Fisheries & Food Trade Corp.
- Korea Ministry of Agrigulture, Food and Rural Affairs, & Korea Food Industry Association. (2013b). *Processed food market report: Milk* (Korea Agro-Fisheries & Food Trade Corp. Publication). Naju: Korea Agro-Fisheries & Food Trade Corp.
- Korea Ministry of Education. (2015). *The status of school meal service in 2014*. Retrieved from http://www.moe.go.kr/web/110501/ko/board/view.do?bbsId=348&boardSeq=609 67.
- Korea Ministry of Education & Human Resources Development. (2007, December 26). Status of banning soft drink sales in schools (Press Release).
- Korea Ministry of Food and Drug Safety. (2013, March 21). 2013 work report for the president: Become a power of the world in food safety by food safety management. Retrieved from http://www.mfds.go.kr/index.do?mid=664&pageNo=24&seq=19971&cmd=v
- Korea Ministry of Food and Drug Safety. (2014). A list of energy-dense nutrient-poor foods that are restricted sales in schools and exemplary stores ('14.7). Retrieved from http://www.mfds.go.kr/index.do?mid=1043&seq=8228&cmd=v
- Korea Ministry of Food and Drug Safety. (2014, July 18). Ongoing efforts to reduce sodium contents of foods (Press Release). Retrieved from http://www.mfds.go.kr/index.do?mid=675&seq=24560&cmd=v
- Korea Ministry of Health and Welfare, & Korea Centers for Disease Control and Prevention. (2013a). *Guideline for Korea National Health and Nutrition*



Examination Survey (KNAHENS V, 2010-2012) data use. Osong: Korea Centers for Disease Control and Prevention.

- Korea Ministry of Health and Welfare, & Korea Centers for Disease Control and Prevention. (2013b). Korean Health Statistics 2012: Korea National Health and Nutrition Examination Survey (KNHANES V-3). Osong: Korea Centers for Disease Control and Prevention.
- Korea National Youth Commission. (2006, March 28). 255 youth training centers nationwide will not sell soft drinks [press release].
- Kubik, M. Y., Davey, C., Nanney, M. S., MacLehose, R. F., Nelson, T. F., & Coombes, B. (2013). Vending and School Store Snack and Beverage Trends: Minnesota Secondary Schools, 2002–2010. *American Journal of Preventive Medicine*, 44(6), 583-588.
- Kubik, M. Y., Farbakhsh, K., & Lytle, L. A. (2011). Two years later: wellness councils and healthier vending in a cohort of middle and high schools. *Journal of Adolescent Health*, 49(5), 550-552.
- Kubik, M. Y., Lytle, L. A., & Farbakhsh, K. (2011). School and district wellness councils and availability of low-nutrient, energy-dense vending fare in Minnesota middle and high schools. *Journal of the American Dietetic Association*, 111(1), 150-155.
- Kubik, M. Y., Lytle, L. A., Hannan, P. J., Perry, C. L., & Story, M. (2003). The association of the school food environment with dietary behaviors of young adolescents. *American Journal of Public Health*, 93(7), 1168-1173. doi:10.2105/AJPH.93.7.1168
- Kweon, S., Kim, Y., Jang, M. J., Kim, K., Choi, S., Chun, C., . . . Oh, K. (2014). Data resource profile: the Korea National Health and Nutrition Examination Survey (KNHANES). *International Journal of Epidemiology*, 43(1), 69-77. doi:10.1093/ije/dyt228
- Larson, N., & Story, M. (2010). Are 'competitive foods' sold at school making our children fat? *Health Affairs*, 29(3), 430-435.

Lasswell, H. D. (1971). A pre-view of policy sciences. New York: American Elsevier.

- Lee, H., Jang, S., & Kim, J. (2011, September 21). Leasing school stores needs to go through open bidding. *Naeil Newspaper*. Retrieved from http://www.naeil.com/News/economy/ViewNews.asp?nnum=625384&sid=E&tid =4
- Lee, S. K. (2012). The effect of nutrition standards of high-calorie foods with lownutritional value on the domestic distribution of children's preferred food. (Doctoral dissertation), Retrieved from http://www.riss.kr/link?id=T12659231.



- Lee, S. Y., Lee, S. S., Kim, G. M., & Kim, S. C. (2012). A Study on the General and Nutritional Information of Children's Preference Foods Sold in the Middle, High School Stores. *Korean Journal of Community Nutrition*, 17(3), 302-311.
- Lee, Y., Shim, J. E., & Yoon, J. (2012). Change of children's meal structure in terms of temporal and spatial dimensions: analysis of the data from the Korea National Health and Nutrition Examination Surveys of 1998 and 2009. *Korean Journal of Community Nutrition, 17*(1), 109-118.
- Lucarelli, J. F., Alaimo, K., Mang, E., Martin, C., Miles, R., Bailey, D., . . . Liu, H. (2014). Facilitators to promoting health in schools: is school health climate the key? *Journal of School Health*, 84(2), 133-140.
- Martin, K. S., Havens, E., Boyle, K. E., Matthews, G., Schilling, E. A., Harel, O., & Ferris, A. M. (2012). If you stock it, will they buy it? Healthy food availability and customer purchasing behaviour within corner stores in Hartford, CT, USA. *Public Health Nutrition*, 15(10), 1973-1978.
- Mâsse, L. C., Naiman, D., & Naylor, P.-J. (2013). From policy to practice: implementation of physical activity and food policies in schools. *International Journal of Behavioral Nutrition and Physical Activity*, 10, 71. doi:10.1186/1479-5868-10-71
- McDonnell, E., Probart, C., Weirich, J. E., Hartman, T., & Bailey-Davis, L. (2006). School competitive food policies: perceptions of Pennsylvania public high school foodservice directors and principals. *Journal of the American Dietetic Association*, 106(2), 271-276.
- McKenna, M. L. (2003). Issues in implementing school nutrition policies. *Canadian Journal of Dietetic Practice and Research*, 64(4), 208-213.
- Mensink, F., Schwinghammer, S. A., & Smeets, A. (2012). The healthy school canteen programme: a promising intervention to make the school food environment healthier. *Journal of Environmental and Public Health*, 2012, 415746. doi:10.1155/2012/415746
- Miles, M. B., & Huberman, A. M. (1994). *Qualitative data analysis: An expanded sourcebook*. Beverly Hilles, CA: Sage Publications.
- Monterrosa, E. C., Campirano, F., Mayo, L. T., Frongillo, E. A., Cordero, S. H., Kaufer-Horwitz, M., & Rivera, J. A. (2015). Stakeholder perspectives on national policy for regulating the school food environment in Mexico. *Health Policy and Planning*, 30(1), 28-38. doi:10.1093/heapol/czt094
- National Center for Health Statistics. (1992). *Dietary methodology workshop for the third National Health and Nutrition Examination Survey* (Vol. 4). Hyattsville, MD: National Center for Health Statistics.



- Nixon, R. (2013, Feb 2). U.S. Releases New Rules for School Snack Foods. *The New York Times*, p. A14.
- O'Toole, T. P., Anderson, S., Miller, C., & Guthrie, J. (2007). Nutrition services and foods and beverages available at school: results from the School Health Policies and Programs Study 2006. *Journal of School Health*, 77(8), 500-521.
- Organization for Economic Co-operation and Development. (2009). Determination and application of administrative fines for environmental offences: Guidance for environmental enforcement authorities in EECCA countries. Retrieved from http://www.oecd.org/env/outreach/42356640.pdf
- Pan Canadian Joint Consortium for School Health. (2010). Stakeholder engagement for improved school policy: development and implementation. *Canadian Journal of Public Health*, 101(8), S20-S23.
- Panisset, U., Koehlmoos, T. P., Alkhatib, A. H., Pantoja, T., Singh, P., Kengey-Kayondo, J., & McCutchen, B. (2012). Implementation research evidence uptake and use for policy-making. *Health Research Policy and Systems*, 10, 20.
- Paudel, N. R. (2009). A critical account of policy implementation theories: status and reconsideration. *Nepalese Journal of Public Policy and Governance*, *8*, 36-54.
- Peart, T., Kao, J., Crawford, P. B., Samuels, S. E., Craypo, L., & Woodward-Lopez, G. (2012). Does competitive food and beverage legislation hurt meal participation or revenues in high schools? *Childhood Obesity*, 8(4), 339-346.
- Pelletier, D. L., Frongillo, E. A., Gervais, S., Hoey, L., Menon, P., Ngo, T., . . . Ahmed, T. (2012). Nutrition agenda setting, policy formulation and implementation: lessons from the Mainstreaming Nutrition Initiative. *Health Policy and Planning*, 27, 19-31.
- Petticrew, M., Cummins, S., Ferrell, C., Findlay, A., Higgins, C., Hoy, C., . . . Sparks, L. (2005). Natural experiments: an underused tool for public health? *Public health*, *119*(9), 751-757. doi:10.1016/j.puhe.2004.11.008
- Ripley, R. B., & Franklin, G. A. (1986). *Policy implementation and bureaucracy* (2nd ed.). Chicago, IL: Dorsey Press.
- Roberts, S. M., Pobocik, R. S., Deek, R., Besgrove, A., & Prostine, B. A. (2009). A qualitative study of junior high school principals' and school food service directors' experiences with the Texas school nutrition policy. *Journal of Nutrition Education and Behavior*, 41(4), 293-299. doi:10.1016/j.jneb.2008.05.010
- Samuels, S. E., Lawrence Bullock, S., Woodward-Lopez, G., Clark, S. E., Kao, J., Craypo, L., . . . Crawford, P. B. (2009). To what extent have high schools in California been able to implement state-mandated nutrition standards? *Journal of Adolescent Health*, 45(3), S38-S44.



- Schwartz, M. B., Henderson, K. E., Falbe, J., Novak, S. A., Wharton, C. M., Long, M. W., . . . Fiore, S. S. (2012). Strength and comprehensiveness of district school wellness policies predict policy implementation at the school level. *Journal of School Health*, 82(6), 262-267.
- Schwartz, M. B., Novak, S. A., & Fiore, S. S. (2009). The impact of removing snacks of low nutritional value from middle schools. *Health Education & Behavior*, 36(6), 999-1011.
- Shroff, M. R., Jones, S. J., Frongillo, E. A., & Howlett, M. (2012). Policy instruments used by states seeking to improve school food environments. *American Journal of Public Health*, 102(2), 222-229.
- Song, B. (2010, October 7). Large F&B companies sell unhealthy foods in schools. Maeil Economics. Retrieved from http://news.mk.co.kr/newsRead.php?year=2010&no=542995
- Song, D. Y., Park, J. E., Shim, J. E., & Lee, J. E. (2013). Trends in the major dish groups and food groups contributing to sodium intake in the Korea National Health and Nutrition Examination Survey 1998-2010. Korean Journal of Nutrition, 46(1), 72-85.
- Stallings, V. A., & Yaktine, A. L. (2007). Nutrition standards for foods in schools: Leading the way toward healthier youth. Washington DC: Institute of Medicine of the National Academy Press.
- Story, M., Kaphingst, K. M., & French, S. (2006). The role of schools in obesity prevention. *Future Child*, *16*(1), 109-142.
- Strauss, A., & Corbin, J. (1990). *Basics of qualitative research: Grounded theory* procedures and techniques. Newbury Park, CA: Sage Publications, Inc.
- Taber, D. R., Chriqui, J. F., & Chaloupka, F. J. (2013). State laws governing school meals and disparities in fruit/vegetable intake. *American Journal of Preventive Medicine*, 44(4), 365-372.
- Taber, D. R., Chriqui, J. F., Powell, L. M., & Chaloupka, F. J. (2012). Banning all sugarsweetened beverages in middle schools: reduction of in-school access and purchasing but not overall consumption. *Archives of Pediatrics and Adolescent Medicine*, 166(3), 256-262.
- Taber, D. R., Chriqui, J. F., Powell, L. M., Perna, F. M., Robinson, W. R., & Chaloupka, F. J. (2015). Socioeconomic differences in the association between competitive food laws and the school food environment. *Journal of School Health*, 85(9), 578-586.
- Taber, D. R., Chriqui, J. F., Vuillaume, R., & Chaloupka, F. J. (2014). How state taxes and policies targeting soda consumption modify the association between school



vending machines and student dietary behaviors: a cross-sectional analysis. *PloS One*, *9*(8), e98249. doi:10.1371/journal.pone.0098249

- Taber, D. R., Stevens, J., Evenson, K. R., Ward, D. S., Poole, C., Maciejewski, M. L., ...
 Brownson, R. C. (2011). State policies targeting junk food in schools:
 Racial/ethnic differences in the effect of policy change on soda consumption. *American Journal of Public Health*, 101(9), 1769-1775.
 doi:10.2105/AJPH.2011.300221
- Taillie, L. S., Afeiche, M. C., Eldridge, A. L., & Popkin, B. M. (2015). Increased snacking and eating occasions are associated with higher energy intake among Mexican children aged 2–13 years. *Journal of Nutrition*, *145*(11), 2570-2577.
- Tansey, O. (2007). Process tracing and elite interviewing: a case for non-probability sampling. *PS: Political Science & Politics*, 40(4), 765-772.
- Templeton, S. B., Marlette, M. A., & Panemangalore, M. (2005). Competitive foods increase the intake of energy and decrease the intake of certain nutrients by adolescents consuming school lunch. *Journal of the American Dietetic Association*, 105(2), 215-220.
- Thacher, A. K. (2005). From concept to practice: using the School Health Index to create healthy school environments in Rhode Island elementary schools. *Preventing Chronic Disease*, *2*, A09.
- Thrasher, J. F., Huang, L., Pérez-Hernández, R., Nuederdeooe, J., Arillo-Santillán, E., & Alday, J. (2011). Evaluation of a social marketing campaign to support Mexico City's comprehensive smoke-free law. *American Journal of Public Health*, 101(2), 328-335.
- U.S. Department of Agriculture, Food and Nutrition Service. Interim final rule. 7 CFR Parts 210 and 220. National school lunch program and school breakfast program: Nutrition standards for all foods sold in school as required by the Healthy, Hunger-Free Kids Act of 2010. Federal Register. 2013; 78 (125).
- Ulin, P. R., Robinson, E. T., & Tolley, E. E. (2004). *Qualitative methods in public health: a field guide for applied research*. San Francisco, CA: Jossey-Bass.
- Varvasovszky, Z., & Brugha, R. (2000). A stakeholder analysis. *Health Policy and Planning*, 15(3), 338-345.
- Vatanparast, H., Lo, E., Henry, C. J., & Whiting, S. J. (2006). A negative trend in calcium intake was accompanied by a substitution of milk by noncarbonated soft drinks in Canadian female students. *Nutrition Research*, *26*(7), 325-329.
- Vecchiarelli, S., Takayanagi, S., & Neumann, C. (2006). Students' perceptions of the impact of nutrition policies on dietary behaviors. *Journal of School Health*, 76(10), 525-531. doi:10.1111/j.1746-1561.2006.00153.x



- Vericker, T. C. (2013). Limited evidence that competitive food and beverage practices affect adolescent consumption behaviors. *Health Education & Behavior, 40*(1), 19-23.
- Vine, M. M., & Elliott, S. J. (2014). Examining local-level factors shaping school nutrition policy implementation in Ontario, Canada. *Public Health Nutrition*, 17(6), 1290-1298. doi:10.1017/S136898001300251
- Walt, G., Shiffman, J., Schneider, H., Murray, S. F., Brugha, R., & Gilson, L. (2008).
 'Doing 'health policy analysis: methodological and conceptual reflections and challenges. *Health Policy and Planning*, 23(5), 308-317.
- Wansink, B. (2004). Environmental factors that increase the food intake and consumption volume of unknowing consumers. *Annual Review of Nutrition, 24*, 455-479.
- Wechsler, H., Devereaux, R. S., Davis, M., & Collins, J. (2000). Using the school environment to promote physical activity and healthy eating. *Preventive Medicine*, 31(2), S121-S137.
- Winter, S. (2003). Implementation perspectives: status and reconsideration *Handbook of public administration* (pp. 212-222). Thousand Oaks, CA: SAGE.
- Wojcicki, J. M., & Heyman, M. B. (2006). Healthier choices and increased participation in a middle school lunch program: effects of nutrition policy changes in San Francisco. *American Journal of Public Health*, 96(9), 1542-1547. doi:10.2105/AJPH.2005.070946
- Woodward-Lopez, G., Gosliner, W., Samuels, S. E., Craypo, L., Kao, J., & Crawford, P.
 B. (2010). Lessons learned from evaluations of California's statewide school nutrition standards. *American Journal of Public Health*, 100(11), 2137-2145.
- World Health Organization. (2010). Set of recommendations on the marketing of foods and non-alcoholic beverages to children. Geneva: World Health Organization.
- Yoon, J., Kwon, S., & Shim, J. E. (2012). Present status and issues of school nutrition programs in Korea. Asia Pacific Journal of Clinical Nutrition, 21(1), 128-133.
- Yun, S.-y. (2014). *The utilization and demand of some high school students for school cafeteria in Gyeongnam area* (Master's thesis), Retrived from http://www.riss.kt/link?id=T13391333



APPENDIX A

Interview Guide

Interview questions: School store owners

- 1. Could you tell about your school store?
 - Please tell me your work experience.
 - How long have you run the current school store?
 - Before running this school store, have you run other school stores? If so, how long and where?
 - What do you do? [perspectives: identity]
 - What you are doing for the school store?
 - Why do you think your role is important?
 - What do you find satisfying or fulfilling about your work? [base values]
 - What are the challenges that you face in doing your work?
- 2. How do you decide food and beverage items that you sell in your school store?
 - Could you tell me about your priority when you decide the food items what you are going to sell? [base values]
 - Does anyone influence your decision about the food items? If so, who influence your decision and in what ways? [participants]



- Have anyone monitored foods that sold in your school stores?
 - (If yes) Who are responsible to monitor foods that sold in your school store?
 - How often do(es) he/she/they monitor school stores?
 - What elements do they focus on when they monitor school stores?
 - If they find faults (e.g. some foods that are not allowed to be sold in school stores), then what happens?

3. Please tell me about policies or regulations that make change of foods and beverages in your school store.

- Have you heard that you need to change foods and beverages in your stores?
 - (If yes) Could you explain more what do you know about it?
 - Do you know why you need to change foods and beverages in your school store?
 - What do you think about it?
 - How does it affect your choice of food and beverage items that you sell?
- Does your school have own policy related to the school store?
 - (If yes) Is the school's own policy different from another policy/regulation that you told before? Could you tell me about the differences?

[Question 4: ask only to the interviewees who are aware of the school store policy]

4. How is food and beverage items changed in your school store?

- How do you change the foods and beverages in the school store? [strategies]
 - How did you know that you need to change food and beverages in your store?
 - Have you been informed about the change? If yes, by whom?
 - How did you prepare to implement the changes?


- Did you need other's help or advice? Has anyone helped you? [perspectives: demands]
- What do you think of your role in the changes of foods and beverages? [perspectives: identity]
 - Do you think your role in the change is consistent with the role that you are supposed to do?
 - Do you satisfy with your role? Why do you think so?
- Do you feel your opinion is respected in the changes of foods and beverages?
 Why?
- Let's talk about other people who are involved in the changes.
 - Who do they represent? [participants]
 - What are their roles and influences?
 - How do they interact with each other to change foods and beverages in the school store?
 - Are there any other people or organizations that are not directly involved, but who may have impact on the process?
 - (If yes) Who are these people/organizations?
 - How do they impact the process?
- Please tell me what has been changed.
 - Could you tell me the change in foods and beverages in your school store?
 - Has the number of [chips, cookies, soft drinks, etc.] changed?
 - What items are increased/decreased?



- How has students' behavior changed after foods and beverages in school stores changed? [outcome]
 - (If yes) Please describe students' behavior change that you notice.
 - Is the change desirable for the students and for you? [outcome, effects]
- What did you expect regarding the influence of the change of foods and beverages in the school store? [perspectives: expectation]
 - Influence to the school store/students/other people
 - Was your expectation positive or negative? Why?
 - Was the expectation the same with the current situation?
- Are there any conflicts of interest in change of foods and beverages in your school store? For example, does the change affect management of your school store?
- 5. I would like to ask about your school store more.
 - Did you make contract for running the school store through a competitive bidding or a private contract?
 - How long is the contract duration?
 - If you want to renew the contract, how can you renew?
 - What factors are considered for renewing the contract?

Interview questions: Policy actors in schools (principal, teachers, school staff)

- 1. Could you tell about you regarding the school store in your school?
 - o Please tell me your work experience.
 - How long have you worked in this school?
 - How long have you worked related to school store in this school?



- Before working in this school, have your work related to school stores in other schools? If so, how long and where?
- What do you do? [perspectives: identity]
 - What you are doing in the school?
 - Why do you think it is important?
 - What do you find satisfying or fulfilling about your work? [base values]
 - What are the challenges that you face in doing your work?
 - What is your roles regarding the school store?
- 2. In general, how are food and beverage items selected in the school store?
 - Is there anyone who can influence food availability in school stores?
 - (If yes) Who and in what ways?
 - What is the priority when the food items are selected?
 - Have anyone monitored foods that sold in your school stores?
 - (If yes) Who are responsible to monitor foods that sold in your school store?
 - How often do(es) he/she/they monitor school stores?
 - What elements do they focus on when they monitor school stores?
 - If they find faults (e.g. some foods that are not allowed to be sold in school stores), then what happens?
- 3. How would you describe the current school store policy?
 - Have you heard that school stores need to change foods and beverages that they sell?
 - (If yes) Could you explain more what do you know about it?
 - Do you know why school stores need to change foods and beverages that they



sold?

- What do you think about it?
- How does it affect choice of food and beverage items in the school store?
- Does your school have own policy related to the school store?
 - (If yes) Is the school's own policy different from another policy/regulation that you told before? Could you tell me about the differences?

[Question 4: ask only to the interviewees who are aware of the school store policy]

- 4. How is food and beverage items changed in the school store?
 - How do you change the foods and beverages in the school store? [strategies]
 - How did you know that the school store in your school need to change food and beverages that they sell?
 - Have you been informed about the change? If yes, by whom?
 - How did you prepare to implement the changes?
 - Did you need other's help or advice? Has anyone helped you? [perspectives: demands]
 - What do you think of your role in the changes of foods and beverages?

[perspectives: identity]

- Do you think your role in the change is consistent with the role that you are supposed to do?
- Do you satisfy with your role? Why do you think so?
- Do you feel your opinion is respected in the changes of foods and beverages?
 Why?
- Let's talk about other people who are involved in the changes.



- Who do they represent? [participants]
- What are their roles and influences?
- How do they interact with each other to change foods and beverages in the school store?
- Are there any other people or organizations that are not directly involved, but who may have impact on the process?
 - (If yes) Who are these people/organizations?
 - How do they impact the process?
- Please tell me what has been changed.
 - Could you tell me the change in foods and beverages in the school store in your school?
 - Has the number of [chips, cookies, soft drinks, etc.] changed?
 - What items are increased/decreased?
 - How has students' behavior changed after foods and beverages in the school store changed? [outcome]
 - (If yes) Please describe students' behavior change that you notice.
 - Is the change desirable for the students and for your school? [outcome, effects]
- What did you expect regarding the influence of the change of foods and beverages in the school store? [perspectives: expectation]
 - Influence to the school store/students/other people
 - Was your expectation positive or negative? Why?
 - Was the expectation the same with the current situation?



- Are there any conflicts of interest in change of foods and beverages in the school store? For example, does the change affect the school store or your school?
- 5. I would like to ask about your school store more.
 - Did you make contract with the school store through a competitive bidding or a private contract?
 - How long is the contract duration?
 - If the school store owner wants to renew the contract, how can he/she renew?
 - What factors are considered for renewing the contract?

Interview questions: Public officials, politicians

- 1. Could you tell about your school district?
 - Please tell me your work experience.
 - How long have you worked in this school district?
 - Before working in this school district, have you worked in other school districts or in schools? If so, how long and where?
 - What do you do? [perspectives: identity]
 - What you are doing in the government agency?
 - Why do you think your role is important?
 - What do you find satisfying or fulfilling about your work? [base values]
 - What are the challenges that you face in doing your work?
 - What is your duty regarding school food policy? [perspectives: identity]
- 2. Could you tell me about food availability in school stores in your school district?



- In general, how do school stores decide food and beverage items that they sell in school stores?
 - In your opinion, what should be considered when school stores decide the food items what they are going to sell? [base values]
 - Is there anyone who can influence food availability in school stores? If so, who and in what ways?
- Have you monitored foods that sold in school stores (in your school district)?
 - (If no) Is there anyone who is responsible to monitor foods that sold in school store?
 - How often do(es) he/she/they monitor school stores?
 - What elements do they focus on when they monitor school stores?
 - If they find faults (e.g. some foods that are not allowed to be sold in school stores), then what happens?
- 3. How would you describe the current school store policy?
 - Have you heard about 'the Special Act on safety management of children's dietary life'?
 - o If so, what do you know about it and what do you think about this policy?
 - Is there any other regulation or policy that affects food and beverage items in school stores in your school district?
 - (If yes) Could you explain what the policy or regulation is and how it affects food availability in school stores?
 - Does your school district have own policy related to school stores?
 - (If yes) Is the school district's own policy different from the national policy?



Could you tell me about the differences?

- Has the national policy (and school district's own policy) been met with acceptance or opposition?
 - (If yes) By whom and why?
- 4. How is the current policy implemented in your school district?
 - How have you been involved with the implementation of the policy in your school district?
 - When the school store policy was enacted, how did you know there is a policy?
 - Have you been informed about the school store policy? If yes, by whom?
 - What do you think of your role in the implementation of the policy?

[perspectives: identity]

- Do you think your role in the implementation of the policy is consistent with the role that you are supposed to do?
- Do you satisfy with your role in the policy implementation? Why do you think so?
- Do you feel your opinion is respected in the policy implementation? Why?
- How do you implement the policy? [strategies]
 - How did you prepare to implement the policy?
 - Did you need other's help or advice? Has anyone helped you for the policy implementation? [perspectives: demands]
 - (If so) Who were they? What were needed? Did they help you?
- o Let's talk about other people who are involved in the process of policy



implementation.

- Who do they represent? [participants]
- What are their roles and influences?
- How do they interact with each other to implement the policy?
 - How is information shared between/among the people involved?
 - How are the decisions made?
- Are there any other people or organizations that are not directly involved, but who may have impact on the process?
 - (If yes) Who are these people/organizations?
 - How do they impact the process?
- Please tell me what has changed after the policy implementation.
 - Could you tell me the change in food items in school stores in your school district after the policy implementation?
 - Has the number of [chips, cookies, soft drinks, etc.] changed after the policy implementation?
 - What food items are increased/decreased?
 - Have you seen students' behavior change after the policy implementation?
 [outcome]
 - (If yes) Please describe students' behavior change that you notice.
 - Is the change desirable for the students? [outcome, effects]
- What did you expect regarding the influence of the policy? [perspectives: expectation]
 - Influence to school stores/students/other people



- Was your expectation positive or negative? Why?
- Was the expectation the same with the current situation?
- Are there any conflicts of interest in the policy implementation? For example, does the policy affect management of school stores in your school district?



APPENDIX B

List of Children's Favorite Foods

Classification		Food types
Snack	Processed Food	- Confectionaries: chips, cookies, biscuits, snacks,
		and others (excluding Hangwa - Korean traditional
		confectionaries), candies, ice candies
		- Bakery foods and chocolates
		- Processed dairy products: processed milk,
		fermented milk (excluding fermented butter milk and
		fermented milk powder), ice creams
		- Processed fish meat products: fish meat sausages
		- Beverages: fruit/vegetable beverages, carbonated
		beverages, yogurt drink, and mixed drinks
	Prepared Food	Confectionaries, bakery foods, and ice creams
Meal	Processed Food	- Noodles (only applied to noodles in containers):
substitute		deep-fried noodles and noodles
		- Ready-to-eat foods: Kimbab (rolled rice in sea
		weed), hamburgers, and sandwiches
	Prepared Food	Hamburgers, pizzas

